# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM S-8 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933



(Exact name of registrant as specified in its charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization) 33-0174996 (I.R.S. Employer Identification No.)

6155 El Camino Real Carlsbad, California 92009 (760) 476-2200 (Address of Principal Executive Offices)

1996 Equity Participation Plan of Viasat, Inc. RigNet, Inc. 2010 Omnibus Incentive Plan RigNet, Inc. 2019 Omnibus Incentive Plan

Nonqualified Stock Option Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 Restricted Stock Unit Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 (Full Titles of the Plans)

Robert J. Blair, Esq.
Vice President, General Counsel and Secretary
Viasat, Inc.
6155 El Camino Real
Carlsbad, California 92009
(760) 476-2200

(Name, Address and Telephone Number, Including Area Code, of Agent for Service)

Copies to:
Craig M. Garner, Esq.
Latham & Watkins LLP
12670 High Bluff Drive
San Diego, California 92130
(858) 523-5400

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	$\boxtimes$	Accelerated filer	
Non-accelerated filer		Smaller reporting company	
		Emerging growth company	

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act.  $\Box$ 

**CALCULATION OF REGISTRATION FEE** 

Title of Each Class of Securities to be Registered	Amount to be Registered (1)	Proposed Maximum Offering Price Per Share	Proposed Maximum Aggregate Offering Price	Amount of Registration Fee
Common stock, \$0.0001 par value (2)	415,340 shares	\$48.415(3)	\$20,108,686.10	\$2,193.86
Common stock, \$0.0001 par value (4)	53,654 shares	\$108.54(5)	\$5,823,605.16	\$635.36
Common stock, \$0.0001 par value (6)	18,019 shares	\$48.415(3)	\$872,389.89	\$95.18
Common stock, \$0.0001 par value (7)	13,668 shares	\$31.28(5)	\$427,535.04	\$46.65
Common stock, \$0.0001 par value (8)	153,865 shares	\$48.415(3)	\$7,449,373.98	\$812.73
Common stock, \$0.0001 par value (9)	1,900 shares	\$33.61(5)	\$63,859.00	\$6.97
Common stock, \$0.0001 par value (10)	2,310 shares	\$48.415(3)	\$111,838.65	\$12.21
Total	658,756 shares	_	\$34,857,287.82	\$3,802.96

- Pursuant to Rule 416 under the Securities Act of 1933, as amended (the "Securities Act"), this registration statement shall also cover any additional shares of common stock of Viasat, Inc., a Delaware corporation ("Viasat" or the "registrant"), which become issuable under the 1996 Equity Participation Plan of Viasat, Inc. (the "1996 Plan"), the RigNet, Inc. 2010 Omnibus Incentive Plan (the "RigNet 2010 Plan"), the RigNet, Inc. 2019 Omnibus Incentive Plan (the "RigNet 2019 Plan"), that certain Nonqualified Stock Option Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 (the "Olivier Option Agreement"), and that certain Restricted Stock Unit Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 (the "Olivier RSU Agreement," and together with the RigNet 2010 Plan, the RigNet 2019 Plan, and the Olivier Option Agreement, the "RigNet Plans") by reason of any stock split, stock dividend, recapitalization or any other similar transaction effected without the receipt of consideration which results in an increase in the number of outstanding shares of common stock. The RigNet Plans and the RigNet Assumed Awards thereunder were assumed by Viasat pursuant to the Merger Agreement at the Effective Time (such terms as defined below in the Explanatory Note).
- Represents shares of common stock available for future issuance under the 1996 Plan in accordance with, and subject to the terms and conditions of, an exception under Nasdaq Stock Market Rule 5635(c)(3), which shares consist of the shares of RigNet common stock that were available for issuance under the RigNet 2019 Plan as of the Effective Time and assumed by Viasat for issuance under the 1996 Plan and adjusted by the exchange ratio pursuant to the Merger Agreement.
- The Proposed Maximum Offering Price Per Share has been estimated in accordance with Rule 457(c) under the Securities Act solely for the purpose of calculating the registration fee. The computation is based upon the average of the high and low prices of Viasat common stock as reported on the Nasdaq Global Select Market on April 26, 2021, because the offering price of the securities to be granted in the future is not currently determinable.
- Represents shares of Viasat common stock issuable pursuant to outstanding stock options granted under the RigNet 2010 Plan that were assumed by Viasat and converted into options to purchase Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement).
- The Proposed Maximum Offering Price Per Share has been estimated in accordance with Rule 457(h) under the Securities Act solely for the purpose of calculating the registration fee. The computation is based upon the weighted average exercise price of the options (as adjusted by the exchange ratio pursuant to the Merger Agreement).
- Represents shares of Viasat common stock issuable pursuant to outstanding restricted stock units granted under the RigNet 2010 Plan that were assumed by Viasat and converted into restricted stock units in respect of Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement).
- Represents shares of Viasat common stock issuable pursuant to outstanding stock options granted under the RigNet 2019 Plan that were assumed by Viasat and converted into stock options to purchase Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement). Shares of Viasat common stock subject to awards granted under the RigNet 2019 Plan may become issuable under the 1996 Plan as a result of the expiration, cancellation or forfeiture of such awards.
- Represents shares of Viasat common stock issuable pursuant to outstanding restricted stock units granted under the RigNet 2019 Plan that were assumed by Viasat and converted into restricted stock units in respect of Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement). Shares of Viasat common stock subject to awards granted under the RigNet 2019 Plan may become issuable under the 1996 Plan as a result of the expiration, cancellation or forfeiture of such awards.
- Represents shares of Viasat common stock issuable pursuant to outstanding stock options granted under the Olivier Option Agreement that were assumed by Viasat and converted into options to purchase Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement).
- Represents shares of Viasat common stock issuable pursuant to outstanding restricted stock units granted under the Olivier RSU Agreement that were assumed by Viasat and converted into restricted stock units in respect of Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement).

Proposed sales to take place as soon after the effective date of the registration statement as awards granted under the above-named plans are granted, exercised and/or distributed.

#### **EXPLANATORY NOTE**

This registration statement on Form S-8 is filed by Viasat in relation to the Merger (as defined below) and the assumption by Viasat of the RigNet Plans.

On April 30, 2021, in connection with the Agreement and Plan of Merger, dated as of December 20, 2020 (the "Merger Agreement"), by and among Viasat, RigNet, Inc., a Delaware corporation ("RigNet"), and Royal Acquisition Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Viasat ("Acquisition Sub"), Acquisition Sub merged with and into RigNet, with RigNet becoming a wholly owned subsidiary of Viasat (the "Merger").

In connection with the consummation of the Merger, the RigNet Plans, and the awards outstanding thereunder (the "RigNet Assumed Awards"), were assumed by Viasat at the effective time of the Merger (the "Effective Time"), including any outstanding shares of RigNet common stock available for issuance under the RigNet 2019 Plan at such time. The RigNet Assumed Awards remain subject to the same terms and conditions as were applicable to such RigNet Assumed Awards immediately prior to the Effective Time, but each such RigNet Assumed Award was automatically converted into an award with the right to shares of Viasat common stock (as adjusted by the exchange ratio pursuant to the Merger Agreement).

In connection with the assumption of the RigNet Plans, shares of Viasat common stock available for issuance under the RigNet 2019 Plan (as adjusted by the exchange ratio pursuant to the Merger Agreement) became available for future awards under the 1996 Plan, and will not reduce the number of shares of common stock authorized for grant under the 1996 Plan, provided that such awards may only be made to individuals who were not employees or members of the board of directors of Viasat prior to the Effective Time.

This registration statement is being filed for the purpose of registering (1) up to 243,416 shares of Viasat common stock issuable by Viasat upon the exercise or settlement of RigNet Assumed Awards that were granted under the RigNet Plans and assumed by Viasat in connection with the Merger and (2) 415,340 shares of Viasat common stock issuable under the 1996 Plan in accordance with, and subject to the terms and conditions of, an exception under Nasdaq Stock Market Rule 5635(c)(3) ("Rule 5635(c)(3)"), which additional shares of Viasat common stock represent the number of shares of RigNet common stock that were available for issuance under the RigNet 2019 Plan immediately prior to the Effective Time, as appropriately adjusted to reflect such transaction (including the application of the exchange ratio under the Merger Agreement) and assumed by Viasat, and not otherwise reserved for issuance under the RigNet 2019 Plan in satisfaction of the RigNet Assumed Awards outstanding thereunder immediately prior to the Effective Time. Pursuant to Rule 5635(c)(3), shares available under certain shareholder-approved plans acquired in acquisitions and mergers may, without further shareholder approval, be used for post-transaction grants of options and other equity awards by the listed company (after appropriate adjustment of the number of shares to reflect the transaction), either under the pre-existing plan or arrangement or another plan or arrangement, without further shareholder approval, provided: (i) the time during which those shares are available for grants is not extended beyond the period when they would have been available under the pre-existing plan, absent the transaction, and (ii) such options and other awards are not granted to individuals who were employed by the granting company or its subsidiaries at the time the merger or acquisition was consummated. The shares of Viasat common stock issuable pursuant to the RigNet Assumed Awards outstanding under the RigNet 2019 Plan may become available for issuance under the 1996 Plan to the extent such awards are terminated, expire or are otherwise cancelled following the Effective Time, which shares are also registered.

In accordance with Instruction E to Form S-8, the contents of the prior Form S-8 registration statements (Commission File Nos. 333-21113, 333-68757, 333-67010, 333-109959, 333-153828, 333-169593, 333-184029, 333-207064, 333-220556, 333-228221, 333-234634 and 333-249941) are incorporated herein by reference.

## PART I INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The documents containing the information specified in Part I of Form S-8 will be sent or given to participants as specified by Rule 428(b)(1) under the Securities Act. These documents and the documents incorporated by reference into this registration statement pursuant to Item 3 of Part II of this registration statement, taken together, constitute a prospectus that meets the requirements of Section 10(a) of the Securities Act.

## PART II INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

#### Item 3. Incorporation of Documents by Reference.

Viasat hereby incorporates the following documents in this registration statement by reference:

- (a) Viasat's Annual Report on Form 10-K for the fiscal year ended March 31, 2020 filed with the Securities and Exchange Commission ("SEC") on May 29, 2020;
- (b) Viasat's Quarterly Reports on Form 10-Q for the fiscal quarters ended <u>June 30, 2020</u> (filed with the SEC on August 7, 2020), <u>September 30, 2020</u> (filed with the SEC on November 6, 2020), and <u>December 31, 2020</u> (filed with the SEC on February 8, 2021):
- (c) Viasat's Current Reports on Form 8-K filed with the SEC on <u>June 17, 2020</u>, <u>June 18, 2020</u>, <u>June 24, 2020</u>, <u>July 24, 2020</u>, <u>September 9, 2020</u>, <u>November 5, 2020</u>, <u>December 21, 2020</u>, and <u>April 1, 2021</u>;
- (d) All other reports filed pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), since the end of the fiscal year covered by Viasat's Annual Report on Form 10-K referred to in clause (a) above; and
- (e) The description of Viasat common stock set forth in Viasat's registration statement on Form 8-A filed with the SEC on November 20, 1996, including any amendment or report filed for the purpose of updating such description.

In addition, all documents filed by Viasat pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act subsequent to the filing of this registration statement and prior to the filing of a post-effective amendment which indicates that all securities offered hereby have been sold or which de-registers all securities then remaining unsold shall be deemed to be incorporated by reference into this registration statement and to be a part hereof from the date of filing such documents, except as to specific sections of such statements as set forth therein. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this registration statement to the extent that a statement contained herein or in any other subsequently filed document which also is incorporated to be modified or superseded for purposes of this registration statement to the extent that a statement contained herein shall be deemed to be modified or superseded for purposes of this registration statement to the extent that a statement contained in any subsequently filed document which also is incorporated or deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this registration statement.

Under no circumstances shall any information furnished under Item 2.02 or 7.01 of Form 8-K be deemed incorporated herein by reference unless such Form 8-K expressly provides to the contrary.

#### Item 4. Description of Securities.

Not applicable.

#### Item 5. Interests of Named Experts and Counsel.

Not applicable.

#### Item 6. Indemnification of Directors and Officers.

Subsection (a) of Section 145 of the General Corporation Law of the State of Delaware, or the DGCL, empowers a corporation to indemnify any person who was or is a party or who is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe the person's conduct was unlawful.

Subsection (b) of Section 145 empowers a corporation to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that the person acted in any of the capacities set forth above, against expenses (including attorneys' fees) actually and reasonably incurred by the person in connection with the defense or settlement of such action or suit if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Section 145 further provides that to the extent a director or officer of a corporation has been successful on the merits or otherwise in the defense of any action, suit or proceeding referred to in subsections (a) and (b) of Section 145, or in defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith; that indemnification provided for by Section 145 shall not be deemed exclusive of any other rights to which the indemnified party may be entitled; and the indemnification provided for by Section 145 shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of such person's heirs, executors and administrators. Section 145 also empowers the corporation to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against such person and incurred by such person in any such capacity, or arising out of his status as such, whether or not the corporation would have the power to indemnify such person against such liabilities under Section 145.

Section 102(b)(7) of the DGCL provides that a corporation's certificate of incorporation may contain a provision eliminating or limiting the personal liability of a director to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, provided that such provision shall not eliminate or limit the liability of a director (i) for any breach of the director's duty of loyalty to the corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the DGCL, or (iv) for any transaction from which the director derived an improper personal benefit. Viasat's certificate of incorporation limits the liability of its directors to

Viasat or its stockholders (in their capacity as directors but not in their capacity as officers) to the fullest extent permitted by the DGCL. Specifically, Viasat's directors will not be personally liable for monetary damages for breach of a director's fiduciary duty as director, notwithstanding any provision of law imposing such liability, except to the extent that DGCL prohibits the elimination or limitation of liability of directors for breaches of fiduciary duty.

Viasat's bylaws generally provide for mandatory indemnification of its directors and officers to the full extent provided by the DGCL. In addition, Viasat has entered into indemnification agreements with its directors and officers that generally provide for mandatory indemnification under circumstances for which indemnification would otherwise be discretionary under Delaware law.

Viasat maintains insurance on behalf of any person who is or was a director or officer of Viasat, or is or was a director or officer of Viasat serving at the request of Viasat as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not Viasat would have the power or obligation to indemnify him against such liability under the provisions of the bylaws.

#### Item 7. Exemption from Registration Claimed.

Not applicable.

#### Item 8. Exhibits.

Exhibit			Incorporate	d by Refer	ence	Filed or Furnished
Number	Exhibit Description	Form	File No.	<u>Exhibit</u>	Filing Date	Herewith
5.1	Opinion of Latham & Watkins LLP					Χ
10.1	1996 Equity Participation Plan of Viasat, Inc. (As Amended and Restated Effective September 3, 2020)	8-K	000-21767	10.1	09/09/2020	
10.2	Amendment to 1996 Equity Participation Plan of Viasat, Inc. (As Amended and Restated Effective September 3, 2020)					Χ
10.3	RigNet, Inc. 2010 Omnibus Incentive Plan					Χ
10.4	Amendment to RigNet, Inc. 2010 Omnibus Incentive Plan					X
10.5	RigNet, Inc. 2019 Omnibus Incentive Plan					X
10.6	Amendment to RigNet, Inc. 2019 Omnibus Incentive Plan					X
10.7	Nonqualified Stock Option Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020					Χ
10.8	Restricted Stock Unit Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020					Χ
23.1	Consent of Independent Registered Public Accounting Firm					X

Exhibit			Incorporat	ed by Referen	ce	Filed or Furnished
Number	Exhibit Description	<u>Form</u>	File No.	<u>Exhibit</u>	Filing Date	Herewith
23.2	Consent of Latham & Watkins LLP (included in Exhibit 5.1 hereto)					Χ
24.1	Power of Attorney (see signature page)					Χ

#### Item 9. Undertakings.

- (a) The undersigned registrant hereby undertakes:
- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
  - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
- (ii) To reflect in the prospectus any facts or events arising after the effective date of this Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in this Registration Statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective Registration Statement; and
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in this Registration Statement or any material change to such information in this Registration Statement;

provided, however, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the Commission by the registrant pursuant to Section 13 or Section 15(d) of the Exchange Act that are incorporated by reference in this Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such posteffective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof; and
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in this Registration Statement shall be deemed to be a new registration statement relating to the securities offered herein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid

by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that the registrant meets all of the requirements for filing on Form S-8 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Carlsbad, State of California, on April 30, 2021.

#### Viasat, Inc.

By: /s/ Richard Baldridge

Richard Baldridge

President, Chief Executive Officer and Director

Each person whose signature appears below hereby constitutes and appoints Richard Baldridge and Robert Blair, jointly and severally, his or her attorneys-in-fact, each with the full power of substitution, for him or her in any and all capacities, to sign this registration statement, and any amendments thereto (including post-effective amendments), and to file the same, with all exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, hereby ratifying and confirming all that each of said attorneys-in-fact, or his or her substitute or substitutes, may do or cause to be done by virtue hereof. Pursuant to the requirements of the Securities Act, this registration statement has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

<u>Signature</u>	<u>Title</u>	<u>Date</u>
/s/ Richard Baldridge Richard Baldridge	President, Chief Executive Officer and Director (Principal Executive Officer)	April 30, 2021
/s/ Shawn Duffy Shawn Duffy	Senior Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)	April 30, 2021
/s/ Mark Dankberg Mark Dankberg	Chairman of the Board of Directors	April 30, 2021
/s/ James Bridenstine James Bridenstine	Director	April 30, 2021
/s/ Robert Johnson Robert Johnson	Director	April 30, 2021
/s/ Sean Pak Sean Pak	Director	April 30, 2021
/s/ Varsha Rao Varsha Rao	Director	April 30, 2021
/s/ John Stenbit John Stenbit	Director	April 30, 2021
/s/ Theresa Wise Theresa Wise	Director	April 30, 2021

#### LATHAM & WATKINS LLP

April 30, 2021

Viasat, Inc. 6155 El Camino Real

Registration Statement on Form S-8; 658,756 Shares of Common Stock, par value \$0.0001 per share

Carlsbad, California 92009

#### Ladies and Gentlemen:

We have acted as special counsel to Viasat, Inc., a Delaware corporation (the "Company"), in connection with the proposed issuance of 658,756 shares (the "Shares") of common stock, par value \$0.0001 per share, of the Company (the "Common Stock"), pursuant to the Company's 1996 Equity Participation Plan (as amended and restated, the "1996 Plan"), the RigNet, Inc. 2010 Omnibus Incentive Plan (the "RigNet 2010 Plan"), the RigNet, Inc. 2019 Omnibus Incentive Plan (the "RigNet 2019 Plan"), that certain Nonqualified Stock Option Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 (the "Olivier Option Agreement"), and that certain Restricted Stock Unit Award Agreement, by and between RigNet, Inc. and Errol Olivier, effective as of January 8, 2020 (the "Olivier RSU Agreement", and collectively, the "Plans"). The Shares are included in a Registration Statement on Form S-8 under the Securities Act of 1933, as amended (the "Act"), filed with the Securities and Exchange Commission (the "Commission") on April 30, 2021 (the "Registration Statement"). This opinion is being furnished in connection with the requirements of Item 601(b)(5) of Regulation S-K under the Act, and no opinion is expressed herein as to any matter pertaining to the contents of the Registration Statement, other than as expressly stated herein with respect to the issuance of the Shares.

As such counsel, we have examined such matters of fact and questions of law as we have considered appropriate for purposes of this letter. With your consent, we have relied upon certificates and other assurances of officers of the Company and others as to factual matters without having independently verified such factual matters. We are opining herein as to the General Corporation Law of the State of Delaware, and we express no opinion with respect to any other laws.

12670 High Bluff Drive San Diego, California 92130 Tel: +1.858.523.5400 Fax: +1.858.523.5450 www.lw.com

FIRM / AFFILIATE OFFICES

Beijing Moscow Boston Munich Brussels New York Century City Orange County Chicago Dubai Riyadh Düsseldorf San Diego Frankfurt San Francisco Hamburg Seoul Hong Kong Shanghai

London Singapore Los Angeles Tokyo

Silicon Valley

Madrid Washington, D.C.

Milan

Houston

#### LATHAM&WATKINSUP

In our examination, we have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals and the conformity to authentic original documents of all documents submitted to us as copies.

Subject to the foregoing and the other matters set forth herein, it is our opinion that, as of the date hereof, when the Shares shall have been duly registered on the books of the transfer agent and registrar therefor in the name or on behalf of the recipients, or certificates representing the Shares (in the form of the specimen certificate incorporated by reference as an exhibit to the Company's most recent Annual Report on Form 10-K) have been manually signed by an authorized officer of the transfer agent and registrar therefor, and subject to the Company completing all actions and proceedings required on its part to be taken prior to the issuance of the Shares, and when the Shares have been issued by the Company in the circumstances contemplated by the Plans for legal consideration in excess of par value, the issuance of the Shares will have been duly authorized by all necessary corporate action of the Company, and the Shares will be validly issued, fully paid and nonassessable. In rendering the foregoing opinion, we have assumed that the Company will comply with all applicable notice requirements regarding uncertificated shares provided in the General Corporation Law of the State of Delaware.

This opinion is for your benefit in connection with the Registration Statement and may be relied upon by you and by persons entitled to rely upon it pursuant to the applicable provisions of the Act. We consent to your filing this opinion as an exhibit to the Registration Statement. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Act or the rules and regulations of the Commission thereunder.

Very truly yours,

/s/ Latham & Watkins LLP

#### AMENDMENT NO. 1 TO THE 1996 EQUITY PARTICIPATION PLAN OF VIASAT, INC. (AS AMENDED AND RESTATED EFFECTIVE SEPTEMBER 3, 2020)

This Amendment No. 1 (this "Amendment") to the 1996 Equity Participation Plan of Viasat, Inc., as amended and restated effective September 3, 2020 (the "Plan") is adopted by the Compensation Committee (the "Committee") of the Board of Directors of Viasat, Inc., a Delaware corporation (the "Company"), as of April 29, 2021 (the "Amendment Effective Date").

**WHEREAS**, the Company maintains the Plan and pursuant to Section 10.2 of the Plan, the Committee has the authority to amend the Plan;

**WHEREAS**, in connection with the transactions contemplated by that certain Agreement and Plan of Merger dated as of December 20, 2020 by and among the Company, Royal Acquisition Sub, Inc., and RigNet, Inc. (the "*Merger Agreement*"), the Company assumed the RigNet, Inc. Omnibus Incentive Plan (as adopted May 8, 2019) (the "*RigNet Plan*") and certain outstanding awards thereunder (as provided in the Merger Agreement), including the assumption of the shares of RigNet common stock available for issuance under the RigNet Plan, in accordance with, and subject to the terms and conditions of, an exception under Nasdaq Stock Market Rule 5635(c)(3) (the "*RigNet Plan Assumption*");

WHEREAS, Nasdaq Stock Market Rule 5635(c)(3) provides that shares available under certain shareholder-approved plans acquired in acquisitions and mergers may, without further shareholder approval, be used for post-transaction grants of options and other equity awards by the listed company (after appropriate adjustment of the number of shares to reflect the transaction), either under the pre-existing plan or arrangement or another plan or arrangement, without further shareholder approval, provided: (i) the time during which those shares are available for grants is not extended beyond the period when they would have been available under the pre-existing plan, absent the transaction, and (ii) such options and other awards are not granted to individuals who were employed by the granting company or its subsidiaries at the time the merger or acquisition was consummated;

**WHEREAS**, the RigNet Plan was approved by RigNet's shareholders and was not adopted in contemplation of the transactions contemplated by the Merger Agreement; and

WHEREAS, the Committee desires to amend the Plan as set forth herein to (i) reflect the RigNet Plan Assumption and (ii) provide that, among other things, certain substitute awards and shares available under certain pre-existing plans of a company that merges with the Company or a subsidiary of the Company will not be counted against the aggregate number of shares of Common Stock (as defined in the Plan) that may be issued pursuant to awards under the Plan.

#### NOW, THEREFORE, BE IT:

**RESOLVED**, that the Plan be and hereby is amended as follows, effective as of the Amendment Effective Date:

1. Immediately after Section 2.2 of the Plan, the following new Section 2.3 shall be inserted:

"2.3 Substitute Awards. In connection with an entity's merger or consolidation with the Company or any Subsidiary or the Company's acquisition, or any Subsidiary's acquisition, of an entity's property or stock, the Committee may grant awards in substitution for any options or other stock or stock-based awards granted before such merger or consolidation by such entity or its affiliate. Substitute Awards may be granted on such terms as the Committee deems appropriate, notwithstanding limitations on awards in the Plan. Substitute Awards will not count against the aggregate number of shares of Common Stock that may be issued pursuant to awards under the Plan (nor shall shares of Common Stock subject to a Substitute Award be added to the shares of Common Stock available for awards under the Plan as provided above), except that shares of Common Stock acquired by exercise of substitute Incentive Stock Options will count against the maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options under the Plan. Additionally, in the event that a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines has shares available under a pre-existing plan approved by stockholders and not adopted in contemplation of such acquisition or combination, the shares available for grant pursuant to the terms of such pre-existing plan (as adjusted, to the extent appropriate, using the exchange ratio or other adjustment or valuation ratio or formula used in such acquisition or combination to determine the consideration payable to the holders of common stock of the entities party to such acquisition or combination) may be used for awards under the Plan and shall not reduce the shares of Common Stock authorized for grant under the Plan (and shares of Common Stock subject to such awards shall not be added to the shares of Common Stock available for awards under the Plan as provided above); provided that awards using such available shares shall not be made after the date awards or grants could have been made under the terms of the pre-existing plan, absent the acquisition or combination, and shall only be made to individuals who were not Employees, consultants or directors of the Company or any Subsidiary immediately prior to such acquisition or combination and to the extent that grants of awards using such available shares are permitted without stockholder approval under the rules of the principal securities exchange on which the Common Stock is then listed.

In furtherance of the foregoing, the shares assumed by the Company under the RigNet, Inc. Omnibus Incentive Plan (as adopted May 8, 2019) (the "RigNet Shares," and such plan, the "RigNet Plan"), including all shares available for issuance under the RigNet Plan as of immediately following the closing of the Company's acquisition of RigNet, Inc. (the "RigNet Closing") as well as any shares subject to outstanding awards under the RigNet Plan as of the RigNet Closing that would otherwise become available for issuance under the RigNet Plan after the RigNet Closing in accordance with the terms of the RigNet Plan as a result of the expiration, cancellation or forfeiture of such awards (in each case after appropriate adjustment of the number of shares to reflect the transaction), may be used for awards under the Plan in accordance with this Section 2.3 and shall not reduce the shares of Common Stock authorized for grant under the Plan. Notwithstanding anything to the contrary contained herein, awards under the Plan with respect to the

RigNet Shares may not be granted (i) to individuals who were employed by or providing services to the Company or any of its Subsidiaries immediately prior to the closing of the Company's acquisition of RigNet, Inc., (ii) following the tenth (10<sup>th</sup>) anniversary of the Effective Date (as defined in the RigNet Plan) of the RigNet Plan, or (iii) in any other manner that would violate the exception under Nasdaq Stock Market Rule 5635(c)(3) relied upon by the Company in connection with the assumption of the RigNet Shares and the reservation of such RigNet Shares for issuance under this Plan. Shares subject to awards granted with respect to the RigNet Shares (whether or not subject to Full Value Awards) shall be counted against the total number of RigNet Shares issuable pursuant to awards under the Plan pursuant to this Section 2.3 as one share for every share subject thereto, and no Incentive Stock Options may be granted from the RigNet Shares."

2. Immediately after the definition of "Subsidiary" in Section 1.33 of the Plan, the following definition shall be inserted as a new Section 1.34 (with corresponding numbering changes to the remainder of Section 1):

"Substitute Awards." Substitute Awards" shall mean awards granted or shares of Common Stock issued by the Company in assumption of, or in substitution or exchange for, awards previously granted, or the right or obligation to make future awards, in each case by a company acquired by the Company or any of its Subsidiaries or with which the Company or any of its Subsidiaries combines."

**IN WITNESS WHEREOF**, I hereby certify that the foregoing Amendment was duly adopted by the Compensation Committee of the Board of Directors of Viasat, Inc. effective as of the Amendment Effective Date.

#### VIASAT, INC.

By: /s/ Brett Church
Name: Brett Church

Title: Associate General Counsel

#### RIGNET, INC. 2010 OMNIBUS INCENTIVE PLAN (As Adopted May 26, 2010)

#### RIGNET, INC. 2010 OMNIBUS INCENTIVE PLAN

ARTICLE I ESTABLISHMENT, PURPOSE AND DURATION	1
1.1 Establishment	1
1.2 Purpose of the Plan	1
1.3 Duration of Plan	1
ARTICLE II DEFINITIONS	1
2.1 "Affiliate"	1
2.2 "Annual Cash Incentive Award"	1
2.3 "Authorized Shares"	2
2.4 "Award"	2
2.5 "Award Agreement"	2
2.6 "Beneficial Owner"	2
2.7 "Board"	2
2.8 "Cash-Based Award"	2
2.9 "Code"	2
2.10 "Committee"	2
2.11 "Company"	2
2.12 "Corporate Change"	2
2.13 "Covered Employee"	2
2.14 "Director"	2
2.15 "Disability"	2
2.16 "Dividend Equivalent"	3
2.17 "Effective Date"	3
2.18 "Employee"	3
2.19 "Exchange Act"	3
2.20 "Fair Market Value"	3
2.21 "Fiscal Year"	3
2.22 "Freestanding SAR"	3
2.23 "Holder"	4
2.24 "Incentive Stock Option" or "ISO"	4
2.25 "Insider"	4
2.26 "Mature Shares"	4
2.27 "Minimum Statutory Tax Withholding Obligation"	4
2.28 "Nonqualified Stock Option" or "NQSO"	4
2.29 "Option"	4
2.30 "Option Price"	4
2.31 "Other Stock-Based Award"	4
2.32 "Parent Corporation"	4
2.33 "Performance-Based Compensation" 2.34 "Performance Goals"	4
	4
2.35 "Performance Stock Award" 2.36 "Performance Unit Award"	4
2.37 "Period of Restriction"	4
2.38 "Period of Restriction" 2.38 "Permissible under Section 409A"	4
2.39 "Plan"	5
2.40 "Restricted Stock"	5
2.41 "Restricted Stock Award"	5 5
2.71 RESUITERED STOCK AWARD	J

2.42 "RSU"	5
2.43 "RSU Award"	
2.44 "SAR"	
2.45 "Section 409A"	
2.46 "Separation from Service"	
2.47 "Stock"	
2.48 "Subsidiary Corporation"	
2.49 "Substantial Risk of Forfeiture"	5
2.50 "Tandem SAR"	5
2.51 "Ten Percent Stockholder"	5
2.52 "Third Party Service Provider"	$\epsilon$
ARTICLE III ELIGIBILITY	•
ARTICLE IV GENERAL PROVISIONS RELATING TO AWARDS	(
4.1 Authority to Grant Awards	$\epsilon$
4.2 Shares That Count Against Limit	7
4.3 Non-Transferability	7
4.4 Requirements of Law	7
4.5 Changes in the Company's Capital Structure	3
4.6 Election Under Section 83(b) of the Code	10
4.7 Forfeiture for Cause	10
4.8 Forfeiture Events	11
4.9 Award Agreements	11
4.10 Amendments of Award Agreements	11
4.11 Rights as Stockholder	11
4.12 Issuance of Shares of Stock	11
4.13 Restrictions on Stock Received	11
4.14 Compliance With Section 409A	11
4.15 Date of Grant	12
4.16 Source of Shares Deliverable Under Awards	12
ARTICLE V OPTIONS	12
5.1 Authority to Grant Options	12
5.2 Type of Options Available	12
5.3 Option Agreement	12
5.4 Option Price	12
5.5 Duration of Option	12
5.6 Amount Exercisable	13
5.7 Exercise of Option	13
5.8 Transferability—Incentive Stock Options	14
5.9 Notification of Disqualifying Disposition	14
5.10 No Rights as Stockholder	15
5.11 \$100,000 Limitation on ISOs	15
5.12 Separation from Service	15
ARTICLE VI STOCK APPRECIATION RIGHTS	15
6.1 Authority to Grant SAR Awards	15
6.2 Type of Stock Appreciation Rights Available	15
6.3 General Terms	15

6.4 SAR Agreement	15
6.5 Term of SAR	16
6.6 Exercise of Freestanding SARs	16
6.7 Exercise of Tandem SARs	16
6.8 Payment of SAR Amount	16 16
6.9 Separation from Service 6.10 Nontransferability of SARs	16
6.11 No Rights as Stockholder	16
6.12 Restrictions on Stock Received	17
ARTICLE VII RESTRICTED STOCK AWARDS	17
7.1 Restricted Stock Awards	17
7.2 Restricted Stock Award Agreement	17
7.3 Holder's Rights as Stockholder	17
ARTICLE VIII RESTRICTED STOCK UNIT AWARDS	17
8.1 Authority to Grant RSU Awards	17
8.2 RSU Award	18
8.3 RSU Award Agreement	18
8.4 Dividend Equivalents	18
8.5 Form of Payment Under RSU Award	18
8.6 Time of Payment Under RSU Award	18
8.7 Holder's Rights as Stockholder	18
ARTICLE IX PERFORMANCE STOCK AWARDS AND PERFORMANCE UNIT AWARDS	18
9.1 Authority to Grant Performance Stock Awards and Performance Unit Awards	18
9.2 Performance Goals	18
9.3 Time of Establishment of Performance Goals	19
9.4 Written Agreement	19
9.5 Form of Payment Under Performance Unit Award	19
9.6 Time of Payment Under Performance Unit Award 9.7 Holder's Rights as Stockholder With Respect to a Performance Stock Award	19 19
9.8 Increases Prohibited	19
9.9 Stockholder Approval	20
9.10 Dividend Equivalents	20
ARTICLE X DIRECTOR AWARDS	20
ARTICLE XI ANNUAL CASH INCENTIVE AWARDS	20
11.1 Authority to Grant Annual Cash Incentive Awards	20
11.2 Covered Employees	20
11.3 Written Agreement	20
11.4 Form of Payment Under Annual Cash Incentive Award	20
11.5 Time of Payment Under Annual Cash Incentive Award	20
ARTICLE XII OTHER STOCK-BASED AWARDS	20
12.1 Authority to Grant Other Stock-Based Awards	20
12.2 Value of Other Stock-Based Award	21

12.3 Payment of Other Stock-Based Award

21

12.4 Separation from Service 12.5 Time of Payment of Other Stock-Based Award	21 21
ARTICLE XIII CASH-BASED AWARDS	21
13.1 Authority to Grant Cash-Based Awards 13.2 Value of Cash-Based Award 13.3 Payment of Cash-Based Award 13.4 Time of Payment of Cash-Based Award 13.5 Separation from Service	21 21 21 21 21
ARTICLE XIV SUBSTITUTION AWARDS	22
ARTICLE XV ADMINISTRATION	22
15.1 Awards 15.2 Authority of the Committee 15.3 Decisions Binding 15.4 No Liability	22 22 23 23
ARTICLE XVI AMENDMENT OR TERMINATION OF PLAN	23
16.1 Amendment, Modification, Suspension, and Termination 16.2 Awards Previously Granted	23 23
	2.4
ARTICLE XVII MISCELLANEOUS	24
17.1 Unfunded Plan/No Establishment of a Trust Fund 17.2 No Employment Obligation 17.3 Tax Withholding 17.4 Indemnification of the Committee 17.5 Gender and Number 17.6 Severability 17.7 Headings 17.8 Other Compensation Plans 17.9 Retirement and Welfare Plans	24 24 24 25 25 25 25 25 25
17.1 Unfunded Plan/No Establishment of a Trust Fund 17.2 No Employment Obligation 17.3 Tax Withholding 17.4 Indemnification of the Committee 17.5 Gender and Number 17.6 Severability 17.7 Headings 17.8 Other Compensation Plans	24 24 24 25 25 25 25 25
17.1 Unfunded Plan/No Establishment of a Trust Fund 17.2 No Employment Obligation 17.3 Tax Withholding 17.4 Indemnification of the Committee 17.5 Gender and Number 17.6 Severability 17.7 Headings 17.8 Other Compensation Plans 17.9 Retirement and Welfare Plans 17.10 Other Awards	24 24 24 25 25 25 25 25 25 25

# RIGNET, INC. 2010 OMNIBUS INCENTIVE PLAN (As Adopted May 26, 2010) ARTICLE I ESTABLISHMENT. PURPOSE AND DURATION

- 1.1 **Establishment.** The Company hereby establishes an incentive compensation plan, to be known as the "*RigNet, Inc. 2010 Omnibus Incentive Plan*", as set forth in this document. The Plan permits the grant of Incentive Stock Options, Nonqualified Stock Options, SARs, Restricted Stock, RSUs, Performance Stock Awards, Performance Unit Awards, Annual Cash Incentive Awards, Other Stock-Based Awards and Cash-Based Awards. The Plan is effective as of May 26, 2010 (the "*Effective Date*"), provided that the Company's stockholders approve the adoption of the Plan within 12 months after the date of adoption of the Plan by the Board.
- 1.2 **Purpose of the Plan.** The Plan is intended to advance the best interests of the Company, its Affiliates and its stockholders by providing those persons who have substantial responsibility for the management and growth of the Company and its Affiliates with additional performance incentives and an opportunity to obtain or increase their proprietary interest in the Company, thereby encouraging them to continue in their employment or affiliation with the Company or its Affiliates.
- 1.3 **Duration of Plan.** The Plan shall continue indefinitely until it is terminated pursuant to Section 16.1. No Award may be granted under the Plan on or after the tenth anniversary of the Effective Date. The applicable provisions of the Plan will continue in effect with respect to an Award granted under the Plan for as long as such Award remains outstanding. Notwithstanding the foregoing, no Incentive Stock Option may be granted under the Plan on or after the date that is ten years from the earlier of (a) adoption of the Plan by the Board and (b) the Effective Date.

## ARTICLE II DEFINITIONS

Each word and phrase defined in this Article shall have the meaning set out below throughout the Plan, unless the context in which any such word or phrase appears reasonably requires a broader, narrower or different meaning.

- 2.1 "Affiliate" means any corporation, partnership, limited liability company or association, trust or other entity or organization which, directly or indirectly, controls, is controlled by, or is under common control with, the Company. For purposes of the preceding sentence, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any entity or organization, shall mean the possession, directly or indirectly, of the power (a) to vote more than fifty percent (50%) of the securities having ordinary voting power for the election of directors or comparable individuals of the controlled entity or organization, or (b) to direct or cause the direction of the management and policies of the controlled entity or organization, whether through the ownership of voting securities or by contract or otherwise.
  - 2.2 "Annual Cash Incentive Award" means an Award granted pursuant to Article XI to an individual who is then a key executive Employee.

- 2.3 "Authorized Shares" shall have the meaning ascribed to that term in Section 4.1(a).
- 2.4 "Award" means, individually or collectively, a grant under the Plan of an Incentive Stock Option, a Nonqualified Stock Option, a SAR, Restricted Stock, a RSU, a Performance Stock Award, a Performance Unit Award, an Annual Cash Incentive Award, an Other Stock-Based Award or a Cash-Based Award, in each case subject to the terms and provisions of the Plan.
  - 2.5 "Award Agreement" means an agreement that sets forth the terms and conditions applicable to an Award granted under the Plan.
- 2.6 "Beneficial Owner" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.
  - 2.7 "Board" means the board of directors of the Company.
  - 2.8 "Cash-Based Award" means an Award granted pursuant to Article XIII.
  - 2.9 "Code" means the United States Internal Revenue Code of 1986, as amended from time to time.
- 2.10 "Committee" means (a) in the case of an Award granted to a Director, the Board, and (b) in the case of any other Award granted under the Plan, the Compensation Committee of the Board or, if the Compensation Committee of the Board chooses to delegate it duties, a committee of at least two persons who are members of the Compensation Committee of the Board and are appointed by the Compensation Committee of the Board to administer the Plan. Each member of the Committee in respect of his or her participation in any decision with respect to an Award that is intended to satisfy the requirements of section 162(m) of the Code must satisfy the requirements of "outside director" status within the meaning of section 162(m) of the Code; provided, however, that the failure to satisfy such requirement shall not affect the validity of the action of any committee otherwise duly authorized and acting in the matter. As to Awards, grants or other transactions that are authorized by the Committee and that are intended to be exempt under Rule 16b-3 of the General Rules and Regulations under the Exchange Act, the requirements of Rule 16b-3(d)(1) of the General Rules and Regulations under the Exchange Act with respect to committee action must also be satisfied.
  - 2.11 "Company" means RigNet, Inc., a Delaware corporation, or any successor (by reincorporation, merger or otherwise).
  - 2.12 "Corporate Change" shall have the meaning ascribed to that term in Section 4.5(c).
- 2.13 "Covered Employee" means an Employee who is a "covered employee," as defined in section 162(m) of the Code and the regulations and other guidance promulgated by the United States Department of Treasury and/or the Internal Revenue Service under section 162(m) of the Code, or any successor statute.
  - 2.14 "*Director*" means a director of the Company who is not an Employee.
- 2.15 "Disability" means as determined by the Committee in its discretion exercised in good faith, (a) in the case of an Award that is exempt from the application of the requirements of Section 409A, a physical or mental condition of the Holder that would entitle him or her to payment of disability income payments under the Company's long-term disability insurance policy or plan for employees as then in effect; or in the event that the Holder is a Director or is not covered, for whatever reason, under the Company's long-term disability insurance policy or plan for employees or in the event the Company does not maintain such a long-term disability insurance policy or for purposes of an ISO granted under the Plan, "Disability" means a permanent and total disability as defined in section 22(e)(3) of the Code and (b) in the case of an Award that is not exempt from the application of

the requirements of Section 409A, (i) the Holder is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, or (ii) the Holder is, by reason of any medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, receiving income replacement benefits for a period of not less than three months under an accident and health plan covering employees of the Company. A determination of Disability may be made by a physician selected or approved by the Committee and, in this respect, the Holder shall submit to an examination by such physician upon request by the Committee.

- 2.16 "Dividend Equivalent" means a payment equivalent in amount to dividends paid to the Company's stockholders.
- 2.17 "Effective Date" shall have the meaning ascribed to that term in Section 1.1.
- 2.18 "*Employee*" means (a) a person employed by the Company or any Affiliate as a common law employee or (b) a person who has agreed to become a common law employee of the Company or any Affiliate and is expected to become such within six (6) months from the date of a determination made for purposes of the Plan.
  - 2.19 "Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor act.
  - 2.20 "Fair Market Value" of the Stock as of any particular date means,
  - (a) if the Stock is traded on a stock exchange,
  - (i) and if the Stock is traded on that date, the closing sale price of the Stock on that date; or
- (ii) and if the Stock is not traded on that date, the closing sale price of the Stock on the last trading date immediately preceding that date; as reported on the principal securities exchange on which the Stock is traded; or
  - (b) if the Stock is traded in the over-the-counter market,
  - (i) and if the Stock is traded on that date, the average between the high bid and low asked price on that date; or
- (ii) and if the Stock is not traded on that date, the average between the high bid and low asked price on the last trading date immediately preceding that date;

as reported in such over-the-counter market; provided, however, that (x) if the Stock is not so traded, or (y) if, in the discretion of the Committee, another means of determining the fair market value of a share of Stock at such date shall be necessary or advisable, the Committee may provide for another method or means for determining such fair market value, which method or means shall comply with the requirements of a reasonable valuation method as described under Section 409A.

- 2.21 "Fiscal Year" means the calendar year.
- 2.22 "Freestanding SAR" means a SAR that is granted independently of any Options, as described in Article VI.

- 2.23 "Holder" means a person who has been granted an Award or any person who is entitled to receive shares of Stock or cash under an Award.
- 2.24 "*Incentive Stock Option*" or "*ISO*" means an option to purchase Stock granted pursuant to Article V that is designated as an incentive stock option and that is intended to satisfy the requirements of section 422 of the Code.
- 2.25 "*Insider*" shall mean an individual who is, on the relevant date, an officer, a Director, or more than ten percent (10%) Beneficial Owner of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, as determined by the Board in accordance with Section 16 of the Exchange Act.
  - 2.26 "Mature Shares" means shares of Stock that the Holder has held for at least six months.
- 2.27 "*Minimum Statutory Tax Withholding Obligation*" means, with respect to an Award, the amount the Company, an Affiliate or other subsidiary is required to withhold for federal, state, local and foreign taxes based upon the applicable minimum statutory withholding rates required by the relevant tax authorities.
- 2.28 "Nonqualified Stock Option" or "NQSO" means a "nonqualified stock option" to purchase Stock granted pursuant to Article V that does not satisfy the requirements of section 422 of the Code.
  - 2.29 "Option" means an Incentive Stock Option or a Nonqualified Stock Option.
  - 2.30 "Option Price" shall have the meaning ascribed to that term in Section 5.4.
- 2.31 "Other Stock-Based Award" means an equity-based or equity-related Award not otherwise described by the terms and provisions of the Plan that is granted pursuant to Article XII.
- 2.32 "Parent Corporation" means any corporation (other than the Company) in an unbroken chain of corporations ending with the Company if, at the time of the action or transaction, each of the corporations other than the Company owns stock possessing 50 percent or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
- 2.33 "*Performance-Based Compensation*" means compensation under an Award that satisfies the requirements of section 162(m) of the Code for deductibility of remuneration paid to Covered Employees.
- 2.34 "*Performance Goals*" means one or more of the criteria described in Section 9.2 on which the performance goals applicable to an Award are based.
  - 2.35 "Performance Stock Award" means an Award designated as a performance stock award granted to a Holder pursuant to Article IX.
  - 2.36 "Performance Unit Award" means an Award designated as a performance unit award granted to a Holder pursuant to Article IX.
- 2.37 "*Period of Restriction*" means the period during which Restricted Stock is subject to a substantial risk of forfeiture (based on the passage of time, the achievement of Performance Goals, or upon the occurrence of other events as determined by the Committee, in its discretion), as provided in Article VII.

- 2.38 "*Permissible under Section 409A*" means with respect to a particular action (such as, the grant, payment, vesting, settlement or deferral of an amount or award under the Plan) that such action shall not subject the compensation at issue to the additional tax or interest applicable under Section 409A.
  - 2.39 "Plan" means the RigNet, Inc. 2010 Omnibus Incentive Plan, as set forth in this document as it may be amended from time to time.
  - 2.40 "Restricted Stock" means shares of restricted Stock issued or granted under the Plan pursuant to Article VII.
  - 2.41 "Restricted Stock Award" means an authorization by the Committee to issue or transfer Restricted Stock to a Holder.
  - 2.42 "RSU" means a restricted stock unit credited to a Holder's ledger account maintained by the Company pursuant to Article VIII.
  - 2.43 "RSU Award" means an Award granted pursuant to Article VIII.
  - 2.44 "SAR" means a stock appreciation right granted under the Plan pursuant to Article VI.
- 2.45 "Section 409A" means section 409A of the Code and the regulations and other guidance promulgated by the United States Department of Treasury and/or the United States Internal Revenue Service under section 409A of the Code, or any successor statute.
- 2.46 "Separation from Service" means, except as otherwise provided in the case of an ISO in the following sentence of this Section 2.45, the termination of the Award recipient's employment or service relationship with the Company and all Affiliates as determined under Section 409A. "Separation from Service" means, in the case of an ISO, the termination of the Employee's employment relationship with all of the Company, any Parent Corporation, any Subsidiary Corporation and any parent or subsidiary corporation (within the meaning of section 422(a)(2) of the Code) of any such corporation that issues or assumes an ISO in a transaction to which section 424(a) of the Code applies.
- 2.47 "Stock" means the common stock of the Company, \$0.001 par value per share (or such other par value as may be designated by act of the Company's stockholders).
- 2.48 "Subsidiary Corporation" means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if, at the time of the action or transaction, each of the corporations other than the last corporation in an unbroken chain owns stock possessing 50 percent or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
  - 2.49 "Substantial Risk of Forfeiture" shall have the meaning ascribed to that term in Section 409A.
- 2.50 "*Tandem SAR*" means a SAR that is granted in connection with a related Option pursuant to Article VI herein, the exercise of which shall require forfeiture of the right to purchase a share of Stock under the related Option (and when a share of Stock is purchased under the Option, the Tandem SAR shall similarly be canceled).
- 2.51 "*Ten Percent Stockholder*" means an individual, who, at the time the applicable Option is granted, owns stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Parent Corporation or Subsidiary Corporation. An individual shall be considered as owning the stock owned, directly or indirectly, by or for his or her brothers and sisters (whether by the whole or

half blood), spouse, ancestors, and lineal descendants; and stock owned, directly or indirectly, by or for a corporation, partnership, estate, or trust, shall be considered as being owned proportionately by or for its stockholders, partners, or beneficiaries.

2.52 "*Third Party Service Provider*" means any consultant, agent, representative, advisor, or independent contractor who renders services to the Company or an Affiliate that (a) are not in connection with the offer and sale of the Company's securities in a capital raising transaction, and (b) do not directly or indirectly promote or maintain a market for the Company's securities, or any other person as determined by the Committee.

### ARTICLE III ELIGIBILITY

Except as otherwise specified in this Article III, the persons who are eligible to receive Awards under the Plan are Employees, Directors and Third Party Service Providers, provided, however, that (a) only those persons who are, on the dates of grant, key employees of the Company or any Parent Corporation or Subsidiary Corporation are eligible for grants of Incentive Stock Options under the Plan, (b) the only persons who are eligible to receive Annual Cash Incentive Awards under the Plan are key executive Employees who, by the nature and scope of their positions, regularly directly make or influence policy decisions which significantly impact the overall results or success of the Company and (c) Directors and Third Party Service Providers are only eligible to receive NQSOs, SARs, Restricted Stock, RSUs, Performance Stock Awards and Performance Unit Awards. Awards other than ISOs, Performance Stock Awards, Performance Units Awards or Annual Cash Incentive Awards may also be granted to a person who is expected to become a key Employee within six months.

## ARTICLE IV GENERAL PROVISIONS RELATING TO AWARDS

- 4.1 **Authority to Grant Awards**. The Committee may grant Awards to those Employees, Directors and Third Party Service Providers as the Committee shall from time to time determine, under the terms and conditions of the Plan. Subject only to any applicable limitations set out in the Plan, the number of shares of Stock or other value to be covered by any Award to be granted under the Plan shall be as determined by the Committee in its sole discretion.
  - (a) The aggregate number of shares of Stock with respect to which Awards may be granted under the Plan is 3,000,000 (the "Authorized Shares").
  - (b) The aggregate number of shares of Stock with respect to which ISOs may be granted under the Plan is equal to the Authorized Shares.
- (c) The maximum number of shares of Stock with respect to which ISOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which NQSOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which SARs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Stock Awards may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Unit Awards payable in shares of Stock may be granted to an Employee during a Fiscal Year, determined as of the dates of grants of the Performance Unit Awards, is \$3,000,000. The maximum

amount that may be paid to a key executive Employee under Annual Cash Incentive Award(s) granted to an Employee during a Fiscal Year is \$3,000,000.

(d) Each of the foregoing numerical limits stated in this Section 4.2 shall be subject to adjustment in accordance with the provisions of Section 4.5.

#### 4.2 Shares That Count Against Limit.

- (a) If shares of Stock are withheld from payment of an Award to satisfy tax obligations with respect to the Award, such shares of Stock will not count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.
- (b) If shares of Stock are tendered in payment of an Option Price of an Option, such shares of Stock will not count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.
- (c) To the extent that any outstanding Award is forfeited or cancelled for any reason or is settled in cash in lieu of shares of Stock, the shares of Stock allocable to such portion of the Award may again be subject to an Award granted under the Plan.
- (d) When a SAR is settled in shares of Stock, the number of shares of Stock subject to the SAR under the SAR Award Agreement will be counted against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan as one share for every share subject to the SAR, regardless of the number of shares used to settle the SAR upon exercise.
- (e) The maximum number of shares of Stock available for issuance under the Plan shall not be reduced to reflect any dividends or Dividend Equivalents that are reinvested into additional shares of Stock or credited as additional Restricted Stock, Restricted Stock Units, Performance Shares, or other Stock-Based Awards.
- 4.3 **Non-Transferability.** Except as specified in the applicable Award Agreements or in domestic relations court orders, an Award shall not be transferable by the Holder other than by will or under the laws of descent and distribution, and shall be exercisable, during the Holder's lifetime, only by him or her. Any attempted assignment of an Award in violation of this Section shall be null and void. In the discretion of the Committee, any attempt to transfer an Award other than under the terms of the Plan and the applicable Award Agreement may terminate the Award.
- 4.4 **Requirements of Law.** The Company shall not be required to sell or issue any shares of Stock under any Award if issuing those shares of Stock would constitute or result in a violation by the Holder or the Company of any provision of any law, statute or regulation of any governmental authority. Specifically, in connection with any applicable statute or regulation relating to the registration of securities, upon exercise of any Option or pursuant to any other Award, the Company shall not be required to issue any shares of Stock unless the Committee has received evidence satisfactory to it to the effect that the Holder will not transfer the shares of Stock except in accordance with applicable law, including receipt of an opinion of counsel satisfactory to the Company to the effect that any proposed transfer complies with applicable law. The determination by the Committee on this matter shall be final, binding and conclusive. The Company may, but shall in no event be obligated to, register any shares of Stock covered by the Plan pursuant to applicable securities laws of any country or any political subdivision. In the event the shares of Stock issuable on exercise of an Option or pursuant to any other Award are not registered, the Company may imprint on the certificate evidencing the shares of Stock any legend that counsel for the Company considers necessary or advisable to comply with applicable law, or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take

such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law. The Company shall not be obligated to take any other affirmative action in order to cause or enable the exercise of an Option or any other Award, or the issuance of shares of Stock pursuant thereto, to comply with any law or regulation of any governmental authority.

#### 4.5 Changes in the Company's Capital Structure.

- (a) The existence of outstanding Awards shall not affect in any way the right or power of the Company or its stockholders to make or authorize any or all adjustments, recapitalizations, reorganizations or other changes in the Company's capital structure or its business, any merger or consolidation of the Company, any issue of bonds, debentures, preferred or prior preference shares ahead of or affecting the Stock or Stock rights, the dissolution or liquidation of the Company, any sale or transfer of all or any part of its assets or business or any other corporate act or proceeding, whether of a similar character or otherwise.
- (b) If the Company shall effect a subdivision or consolidation of Stock or other capital readjustment, the payment of a Stock dividend, or other increase or reduction of the number of shares of Stock outstanding, without receiving compensation therefor in money, services or property, then (i) the number, class or series and per share price of Stock subject to outstanding Awards under the Plan shall be appropriately adjusted in such a manner as to entitle a Holder to receive upon exercise of an Award, for the same aggregate cash consideration, the equivalent total number and class or series of Stock the Holder would have received had the Holder exercised his or her Award in full immediately prior to the event requiring the adjustment, and (ii) the number and class or series of Stock then reserved to be issued under the Plan shall be adjusted by substituting for the total number and class or series of Stock then reserved, that number and class or series of Stock that would have been received by the owner of an equal number of outstanding shares of Stock of each class or series of Stock as the result of the event requiring the adjustment.
- (c) If while unexercised Awards remain outstanding under the Plan (i) the Company shall not be the surviving entity in any merger, consolidation or other reorganization (or survives only as a subsidiary of an entity other than an entity that was wholly-owned by the Company immediately prior to such merger, consolidation or other reorganization), (ii) the Company sells, leases or exchanges or agrees to sell, lease or exchange all or substantially all of its assets to any other person or entity (other than an entity wholly-owned by the Company), (iii) the Company is to be dissolved or (iv) the Company is a party to any other corporate transaction (as defined under section 424(a) of the Code and applicable Department of Treasury regulations) that is not described in clauses (i), (ii) or (iii) of this sentence (each such event is referred to herein as a "Corporate Change"), then, except as otherwise provided in an Award Agreement or another agreement between the Holder and the Company (provided that such exceptions shall not apply in the case of a reincorporation merger), or as a result of the Committee's effectuation of one or more of the alternatives described below, there shall be no acceleration of the time at which any Award then outstanding may be exercised, and no later than ten days after the approval by the stockholders of the Company of such Corporate Change, the Committee, acting in its sole and absolute discretion without the consent or approval of any Holder, shall act to effect one or more of the following alternatives, which may vary among individual Holders and which may vary among Awards held by any individual Holders of the Company's ordinary shares will receive one ordinary share of the successor corporation for each ordinary share of the Company, none of such alternatives shall apply and, without Committee action, each Award shall automatically convert into a similar award of the successor corporation exercisable for the same number of ordinary shares of the successor as the Award was exercisabl

- (1) accelerate the time at which some or all of the Awards then outstanding may be exercised so that such Awards may be exercised in full for a limited period of time on or before a specified date (before or after such Corporate Change) fixed by the Committee, after which specified date all such Awards that remain unexercised and all rights of Holders thereunder shall terminate;
- (2) require the mandatory surrender to the Company by all or selected Holders of some or all of the then outstanding Awards held by such Holders (irrespective of whether such Awards are then exercisable under the provisions of the Plan or the applicable Award Agreement evidencing such Award) as of a date, before or after such Corporate Change, specified by the Committee, in which event the Committee shall thereupon cancel such Award and the Company shall pay to each such Holder an amount of cash per share equal to the excess, if any, of the per share price offered to stockholders of the Company in connection with such Corporate Change over the exercise prices under such Award for such shares;
- (3) with respect to all or selected Holders, have some or all of their then outstanding Awards (whether vested or unvested) assumed or have a new award of a similar nature substituted for some or all of their then outstanding Awards under the Plan (whether vested or unvested) by an entity which is a party to the transaction resulting in such Corporate Change and which is then employing such Holder or which is affiliated or associated with such Holder in the same or a substantially similar manner as the Company prior to the Corporate Change, or a parent or subsidiary of such entity, provided that (A) such assumption or substitution is on a basis where the excess of the aggregate fair market value of the Stock subject to the Award immediately after the assumption or substitution over the aggregate exercise price of such Stock is equal to the excess of the aggregate fair market value of all Stock subject to the Award immediately before such assumption or substitution over the aggregate exercise price of such Stock, and (B) the assumed rights under such existing Award or the substituted rights under such new Award, as the case may be, will have the same terms and conditions as the rights under the existing Award assumed or substituted for, as the case may be;
- (4) provide that the number and class or series of Stock covered by an Award (whether vested or unvested) theretofore granted shall be adjusted so that such Award when exercised shall thereafter cover the number and class or series of Stock or other securities or property (including, without limitation, cash) to which the Holder would have been entitled pursuant to the terms of the agreement or plan relating to such Corporate Change if, immediately prior to such Corporate Change, the Holder had been the holder of record of the number of shares of Stock then covered by such Award; or
- (5) make such adjustments to Awards then outstanding as the Committee deems appropriate to reflect such Corporate Change (provided, however, that the Committee may determine in its sole and absolute discretion that no such adjustment is necessary to reflect such Corporate Change).

Any adjustment effected by the Committee under Section 4.5 shall be designed to provide the Holder with the intrinsic value of his or her Award, as determined prior to the Corporate Change, or, if applicable, equalize the Fair Market Value of the Award before and after the Corporate Change.

In effecting one or more of the alternatives set out in paragraphs (3), (4) or (5) immediately above, and except as otherwise may be provided in an Award Agreement, the Committee, in its sole and

absolute discretion and without the consent or approval of any Holder, may accelerate the time at which some or all Awards then outstanding may be exercised.

- (d) In the event of changes in the outstanding Stock by reason of recapitalizations, reorganizations, mergers, consolidations, combinations, exchanges or other relevant changes in capitalization occurring after the date of the grant of any Award and not otherwise provided for by this Section 4.5, any outstanding Award and any Award Agreement evidencing such Award shall be subject to adjustment by the Committee in its sole and absolute discretion as to the number and price of Stock or other consideration subject to such Award. In the event of any such change in the outstanding Stock, the aggregate number of shares of Stock available under the Plan may be appropriately adjusted by the Committee, whose determination shall be conclusive.
- (e) After a merger of one or more corporations into the Company or after a consolidation of the Company and one or more corporations in which the Company shall be the surviving corporation, each Holder shall be entitled to have his or her Restricted Stock appropriately adjusted based on the manner in which the shares of Stock were adjusted under the terms of the agreement of merger or consolidation.
- (f) The issuance by the Company of stock of any class or series, or securities convertible into, or exchangeable for, stock of any class or series, for cash or property, or for labor or services either upon direct sale or upon the exercise of rights or warrants to subscribe for them, or upon conversion or exchange of stock or obligations of the Company convertible into, or exchangeable for, stock or other securities, shall not affect, and no adjustment by reason of such issuance shall be made with respect to, the number, class or series, or price of shares of Stock then subject to outstanding Awards.
- 4.6 **Election Under Section 83(b) of the Code.** No Holder shall exercise the election permitted under section 83(b) of the Code with respect to any Award without the prior written approval of the General Counsel or the Chief Financial Officer of the Company. Any Holder who makes an election under section 83(b) of the Code with respect to any Award without the prior written approval of the General Counsel or the Chief Financial Officer of the Company may, in the discretion of the Committee, forfeit any or all Awards granted to him or her under the Plan.
- 4.7 Forfeiture for Cause. Notwithstanding any other provision of the Plan or an Award Agreement, if the Committee finds by a majority vote that a Holder, before or after his or her Separation from Service (a) committed a felony or a crime involving moral turpitude or committed any other act or omission involving fraud, embezzlement or any other act of dishonesty in the course of his or her employment by the Company or an Affiliate which conduct damaged the Company or an Affiliate; (b) substantially and repeatedly failed to perform duties of the office held by the Holder as reasonably directed by the Company or an Affiliate, (c) committed gross negligence or willful misconduct with respect to the Company or an Affiliate; (d) committed a material breach of any employment agreement between the Holder and the Company or an Affiliate that is not cured within ten (10) days after receipt of written notice thereof from the Company or the Affiliate, as applicable; (e) failed, within ten (10) days after receipt by the Holder of written notice thereof from the Company or an Affiliate, to correct, cease or otherwise alter any failure to comply with instructions or other action or omission which the Board reasonably believes does or may materially or adversely affect the Company's or an Affiliate's business or operations, (f) committed misconduct which is of such a serious or substantial nature that a reasonable likelihood exists that such misconduct will materially injure the reputation of the Company or an Affiliate, (g) harassed or discriminated against the Company's or an Affiliate's employees, customers or vendors in violation of the Company's policies with respect to such matters, (h) misappropriated funds or assets of the Company or an Affiliate for personal use or willfully violated the Company policies or standards of business conduct as determined in good faith by the Board, (i) failed, due to some action or inaction on the part of the Holder, to have immigration status that permits the Holder to maintain full-tim

the United States in compliance with all applicable immigration law, (j) disclosed trade secrets of the Company or an Affiliate, then as of the date the Committee makes its finding, any Awards awarded to the Holder that have not been exercised by the Holder (including all Awards that have not yet vested) will be forfeited to the Company. The findings and decision of the Committee or the Board, if applicable, with respect to such matter, including those regarding the acts of the Holder and the damage done to the Company, will be final for all purposes. No decision of the Committee, however, will affect the finality of the discharge of the individual by the Company or an Affiliate.

- 4.8 **Forfeiture Events.** The Committee may specify in an Award Agreement that the Holder's rights, payments, and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture, or recoupment upon the occurrence of certain specified events, in addition to any otherwise applicable vesting or performance conditions of an Award. Such events may include, but shall not be limited to, Separation from Service for cause, Separation from Service for any other reason, violation of material policies of the Company and its Affiliates, breach of noncompetition, confidentiality, or other restrictive covenants that may apply to the Holder, or other conduct by the Holder that is detrimental to the business or reputation of the Company and its Affiliates.
- 4.9 **Award Agreements**. Each Award shall be embodied in a written Award Agreement that shall be subject to the terms and conditions of the Plan. The Award Agreement shall be signed by an executive officer of the Company, other than the Holder, on behalf of the Company, and may be signed by the Holder to the extent required by the Committee. The Award Agreement may specify the effect of a change in control of the Company on the Award. The Award Agreement may contain any other provisions that the Committee in its discretion shall deem advisable which are not inconsistent with the terms and provisions of the Plan.
- 4.10 **Amendments of Award Agreements.** The terms of any outstanding Award under the Plan may be amended from time to time by the Committee in its discretion in any manner that it deems appropriate and that is consistent with the terms of the Plan or necessary to implement the requirements of the Plan. However, no such amendment shall adversely affect in a material manner any right of a Holder without his or her written consent. Except as specified in Section 4.5(b), the Committee may not directly or indirectly lower the exercise price of a previously granted Option or the grant price of a previously granted SAR.
- 4.11 **Rights as Stockholder.** A Holder shall not have any rights as a stockholder with respect to Stock covered by an Option, a SAR, an RSU, a Performance Unit, or an Other Stock-Based Award payable in Stock until the date, if any, such Stock is issued by the Company; and, except as otherwise provided in Section 4.5, no adjustment for dividends, or otherwise, shall be made if the record date therefor is prior to the date of issuance of such Stock.
  - 4.12 **Issuance of Shares of Stock.** Shares of Stock, when issued, may be represented by a certificate or by book or electronic entry.
- 4.13 **Restrictions on Stock Received.** The Committee may impose such conditions and/or restrictions on any shares of Stock issued pursuant to an Award as it may deem advisable or desirable. These restrictions may include, but shall not be limited to, a requirement that the Holder hold the shares of Stock for a specified period of time.
- 4.14 **Compliance With Section 409A.** Awards shall be designed, granted and administered in such a manner that they are either exempt from the application of, or comply with, the requirements of Section 409A. The Plan and each Award Agreement under the Plan that is intended to comply the requirements of Section 409A shall be construed and interpreted in accordance with such intent. If the Committee determines that an Award, Award Agreement, payment, distribution, deferral election, transaction, or any other action or arrangement contemplated by the provisions of the Plan would, if undertaken, cause a Holder to become subject to additional

taxes under Section 409A, then unless the Committee specifically provides otherwise, such Award, Award Agreement, payment, distribution, deferral election, transaction or other action or arrangement shall not be given effect to the extent it causes such result and the related provisions of the Plan and/or Award Agreement will be deemed modified, or, if necessary, suspended in order to comply with the requirements of Section 409A to the extent determined appropriate by the Committee, in each case without the consent of or notice to the Holder. The exercisability of an Option or a SAR shall not be extended to the extent that such extension would subject the Holder to additional taxes under Section 409A.

- 4.15 **Date of Grant**. The date on which an option or SAR is granted shall be the date the Company completes the corporate action constituting an offer of stock for sale to a Holder under the terms and conditions of the Option or SAR; *provided* that such corporate action shall not be considered complete until the date on which the *maximum* number of shares that can be purchased under the Option and the minimum Option price are fixed or determinable. If the corporate action contemplates an immediate offer of stock for sale to a class of individuals, then the date of the granting of an Option is the time or date of that corporate action, if the offer is to be made immediately. If the corporate action contemplates a particular date on which the offer is to be made, then the date of grant is the contemplated date of the offer.
- 4.16 **Source of Shares Deliverable Under Awards.** Any shares of Stock delivered pursuant to an Award may consist, in whole or in part, of authorized and unissued shares of Stock or of treasury shares of Stock.

#### ARTICLE V OPTIONS

- 5.1 **Authority to Grant Options.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Options under the Plan to eligible persons in such number and upon such terms as the Committee shall determine; *provided* that ISOs may be granted only to eligible Employees of the Company or of any Parent Corporation or Subsidiary Corporation (as permitted by section 422 of the Code and the regulations thereunder).
  - 5.2 **Type of Options Available.** Options granted under the Plan may be NQSOs or ISOs.
- 5.3 **Option Agreement.** Each Option grant under the Plan shall be evidenced by an Award Agreement that shall specify (a) whether the Option is intended to be an ISO or an NQSO, (b) the Option Price, (c) the duration of the Option, (d) the number of shares of Stock to which the Option pertains, (e) the exercise restrictions, if any, applicable to the Option and (f) such other provisions as the Committee shall determine that are not inconsistent with the terms and provisions of the Plan. Notwithstanding the designation of an Option as an ISO in the applicable Award Agreement for such Option, to the extent the limitations of Section 5.11 of the Plan are exceeded with respect to the Option, the portion of the Option in excess of the limitation shall be treated as a NQSO. An Option granted under the Plan may not be granted with any Dividend Equivalents rights.
- 5.4 **Option Price.** The price at which shares of Stock may be purchased under an Option (the "*Option Price*") shall not be less than one hundred percent (100%) of the Fair Market Value of the shares of Stock on the date the Option is granted; *provided*, *however*, if the Option is an ISO granted to a Ten Percent Stockholder, the Option Price must not be less than one hundred ten percent (110%) of the Fair Market Value of the shares of Stock on the date the ISO is granted. Subject to the limitations set forth in the preceding sentences of this Section 5.4, the Committee shall determine the Option Price for each grant of an Option under the Plan.
- 5.5 **Duration of Option.** An Option shall not be exercisable after the earlier of (a) the general term of the Option specified in the applicable Award Agreement (which shall not exceed ten years, or, in the case of a Ten Percent Stockholder, no ISO shall be exercisable later than the fifth (5th) anniversary of the date of its grant)

or (b) the period of time specified in the applicable Award Agreement that follows the Holder's Separation from Service.

5.6 **Amount Exercisable.** Each Option may be exercised at the time, in the manner and subject to the conditions the Committee specifies in the Award Agreement in its sole discretion.

#### 5.7 Exercise of Option.

(a) *General Method of Exercise*. Subject to the terms and provisions of the Plan and the applicable Award Agreement, Options may be exercised in whole or in part from time to time by the delivery of written notice in the manner designated by the Committee stating (i) that the Holder wishes to exercise such Option on the date such notice is so delivered, (ii) the number of shares of Stock with respect to which the Option is to be exercised and (iii) the address to which a stock certificate, if any, representing such shares of Stock should be mailed or delivered, or the account to which the shares of Stock represented by book or electronic entry should be delivered. Except in the case of exercise by a third party broker as provided below, in order for the notice to be effective the notice must be accompanied by payment of the Option Price (and all applicable federal, state, local and foreign withholding taxes described in Section 17.3) by any combination of the following: (w) cash, certified check, bank draft or postal or express money order for an amount equal to the Option Price under the Option, (x) Mature Shares with a Fair Market Value on the date of exercise equal to the Option Price under the Option (if approved in advance by the Committee or an executive officer of the Company), (y) an election to make a cashless exercise through a registered broker-dealer (if approved in advance by the Committee or an executive officer of the Company) or (z) except as specified below, any other form of payment which is acceptable to the Committee. If Mature Shares are used for payment by the Holder, the aggregate Fair Market Value of the shares of Stock tendered must be equal to or less than the aggregate Option Price of the shares of Stock being purchased upon exercise of the Option, and any difference must be paid by cash, certified check, bank draft or postal or express money order payable to the order of the Company.

If, at the time of receipt by the Company or its delegate of such written notice, (i) the Company has unrestricted surplus in an amount not less than the Option Price of such shares of Stock, (ii) all accrued cumulative preferential dividends and other current preferential dividends on all outstanding shares of preferred stock of the Company have been fully paid, (iii) the acquisition by the Company of its own shares of Stock for the purpose of enabling such Holder to exercise such Option is otherwise permitted by applicable law, does not require any vote or consent of any stockholder of the Company and does not violate the terms of any agreement to which the Company is a party or by which it is bound, and (iv) there shall have been adopted, and there shall be in full force and effect, a resolution of the Board authorizing the acquisition by the Company of its own shares of stock for such purpose, then such Holder may deliver to the Company, in payment of the Option Price of the shares of Stock with respect to which such Option is exercised, (x) certificates registered in the name of such Holder that represent a number of shares of stock legally and beneficially owned by such Holder (free of all liens, claims and encumbrances of every kind) and having a Fair Market Value on the date of receipt by the Company or its delegate of such written notice that is not greater than the Option Price of the shares of Stock with respect to which such Option is to be exercised, such certificates to be accompanied by stock powers duly endorsed in blank by the record holder of the shares of Stock represented by such certificates, with the signature of such record holder guaranteed by a national banking association, and (y) if the Option Price of the shares of Stock with respect to which such Option is to be exercised exceeds such Fair Market Value, a cashier's check drawn on a national banking association and payable to the order of the Company, in an amount, in United States dollars, equal to the amount of such excess. Notwithstanding the provisions of the immediately preceding sentence, the Committee, in its sole discretion, may refuse to accept shares of Stock in payment of the Option Price of the shares of Stock with respect to which such Option is to be

exercised and, in that event, any certificates representing shares of Stock that were received by the Company or its delegate with such written notice shall be returned to such Holder, together with notice by the Company or its delegate to such Holder of the refusal of the Committee to accept such shares of Stock. If, at the expiration of seven business days after the delivery to such Holder of such written notice from the Company or its delegate, such Holder shall not have delivered to the Company or its delegate a cashier's check drawn on a national banking association and payable to the order of the Company in an amount, in United States dollars, equal to the Option Price of the shares of Stock with respect to which such Option is to be exercised, such written notice from the Holder to the Company or its delegate shall be ineffective to exercise such Option.

Whenever an Option is exercised by exchanging shares of Stock owned by the Holder, the Holder shall deliver to the Company or its delegate certificates registered in the name of the Holder representing a number of shares of Stock legally and beneficially owned by the Holder, free of all liens, claims, and encumbrances of every kind, accompanied by stock powers duly endorsed in blank by the record holder of the shares represented by the certificates, (with signature guaranteed by a commercial bank or trust company or by a brokerage firm having a membership on a registered national stock exchange). The delivery of certificates upon the exercise of Option is subject to the condition that the person exercising the Option provide the Company with the information the Company might reasonably request pertaining to exercise, sale or other disposition of an Option.

- (b) *Issuance of Shares*. Subject to Section 4.3 and Section 5.7(c), as promptly as practicable after receipt of written notification and payment, in the form required by Section 5.7(a), of an amount of money necessary to satisfy any withholding tax liability that *may* result from the exercise of such Option, the Company shall deliver to the Holder certificates for the number of shares with respect to which the Option has been exercised, issued in the Holder's name. Delivery of the shares shall be deemed effected for all purposes when a stock transfer agent of the Company shall have deposited the certificates in the United States mail, addressed to the Holder, at the address specified by the Holder or shall have transferred to the account designated by the Holder to which the shares of Stock represented by book or electronic entry are to be delivered.
- (c) Exercise Through Third-Party Broker. The Committee may permit a Holder to elect to pay the Option Price and any applicable tax withholding resulting from such exercise by authorizing a third-party broker to sell all or a portion of the shares of Stock acquired upon exercise of the Option and remit to the Company a sufficient portion of the sale proceeds to pay the Option Price and any applicable federal, state, local and foreign tax withholding resulting from such exercise.
- (d) *Limitations on Exercise Alternatives*. The Committee shall not permit a Holder to pay such Holder's Option Price upon the exercise of an Option by having the Company reduce the number of shares of Stock that will be delivered pursuant to the exercise of the Option. In addition, the Committee shall not permit a Holder to pay such Holder's Option Price upon the exercise of an Option by using shares of Stock other than Mature Shares. An Option may not be exercised for a fraction of a share of Stock.
- 5.8 **Transferability—Incentive Stock Options**. Notwithstanding anything in the Plan or an Award Agreement to the contrary, no ISO granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution, and all ISOs granted to an Employee under this Article V shall be exercisable during his or her lifetime only by such Employee.
- 5.9 **Notification of Disqualifying Disposition**. If any Employee shall make any disposition of shares of Stock issued pursuant to the exercise of an ISO under the circumstances described in section 421(b) of the

Code (relating to certain disqualifying dispositions), such Employee shall notify the Company of such disposition within ten (10) days thereof.

- 5.10 **No Rights as Stockholder**. A Holder of an Option shall not have any rights as a stockholder with respect to Stock covered by an Option until the date a stock certificate for such Stock is issued by the Company; and, except as otherwise provided in Section 4.5, no adjustment for dividends, or otherwise, shall be made if the record date therefor is prior to the date of issuance of such certificate.
- 5.11 **\$100,000** Limitation on ISOs. To the extent that the aggregate Fair Market Value of shares of Stock with respect to which ISOs first become exercisable by a Holder in any calendar year exceeds \$100,000, taking into account both shares of Stock subject to ISOs under the Plan and Stock subject to ISOs under all other plans of the Company, such Options shall be treated as NQSOs. For this purpose, the "Fair Market Value" of the shares of Stock subject to Options shall be determined as of the date the Options were awarded. In reducing the number of Options treated as ISOs to meet the \$100,000 limit, the most recently granted Options shall be reduced first. To the extent a reduction of simultaneously granted Options is necessary to meet the \$100,000 limit, the Committee may, in the manner and to the extent permitted by law, designate which shares of Stock are to be treated as shares acquired pursuant to the exercise of an ISO.
- 5.12 **Separation from Service.** Each Award Agreement shall set forth the extent to which the Holder of an Option shall have the right to exercise the Option following the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee, need not be uniform among all Options issued pursuant to the Award Agreement or the Plan, and may reflect distinctions based on the reasons for termination.

#### ARTICLE VI STOCK APPRECIATION RIGHTS

- 6.1 **Authority to Grant SAR Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant SARs under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. Subject to the terms and conditions of the Plan, the Committee shall have complete discretion in determining the number of SARs granted to each Holder and, consistent with the provisions of the Plan, in determining the terms and conditions pertaining to such SARs.
- 6.2 **Type of Stock Appreciation Rights Available**. The Committee may grant Freestanding SARs, Tandem SARs, or any combination of these forms of SARs.
- 6.3 **General Terms.** Subject to the terms and conditions of the Plan, a SAR granted under the Plan shall confer on the recipient a right to receive, upon exercise thereof, an amount equal to the excess of (a) the Fair Market Value of one share of the Stock on the date of exercise over (b) the grant price of the SAR, which shall not be less than one hundred percent (100%) of the Fair Market Value of one share of the Stock on the date of grant of the SAR. The grant price of Tandem SARs shall be equal to the Option Price of the related Option. A SAR granted under the Plan may not be granted with any Dividend Equivalents rights.
- 6.4 **SAR Agreement.** Each Award of SARs granted under the Plan shall be evidenced by an Award Agreement that shall specify (a) the grant price of the SAR, (b) the term of the SAR, (c) the vesting and termination provisions of the SAR and (d) such other provisions as the Committee shall determine that are not inconsistent with the terms and provisions of the Plan. The Committee may impose such additional conditions or restrictions on the exercise of any SAR as it may deem appropriate.

- 6.5 **Term of SAR.** The term of a SAR granted under the Plan shall be determined by the Committee, in its sole discretion; provided that no SAR shall be exercisable on or after the tenth anniversary date of its grant. Notwithstanding any other provision of this Plan to the contrary, with respect to a Tandem SAR granted in connection with an ISO: (a) the Tandem SAR will expire no later than the expiration of the underlying ISO; (b) the value of the payout with respect to the Tandem SAR may be for no more than one hundred percent (100%) of the excess of the Fair Market Value of the shares of Stock subject to the underlying ISO; and (c) the Tandem SAR may be exercised only when the Fair Market Value of the shares of Stock subject to the ISO exceeds the Option Price of the ISO.
- 6.6 Exercise of Freestanding SARs. Subject to the terms and provisions of the Plan and the applicable Award Agreement, Freestanding SARs may be exercised in whole or in part from time to time by the delivery of written notice in the manner designated by the Committee stating (a) that the Holder wishes to exercise such SAR on the date such notice is so delivered, (b) the number of shares of Stock with respect to which the SAR is to be exercised and (c) the address to which the payment due under such SAR should be delivered. In accordance with applicable law, a Freestanding SAR may be exercised upon whatever additional terms and conditions the Committee, in its sole discretion, imposes.
- 6.7 Exercise of Tandem SARs. Subject to the terms and provisions of the Plan and the applicable Award Agreement, Tandem SARs may be exercised for all or part of the shares of Stock subject to the related Option upon the surrender of the right to exercise the equivalent portion of the related Option and by the delivery of written notice in the manner designated by the Committee stating (a) that the Holder wishes to exercise such SAR on the date such notice is so delivered, (b) the number of shares of Stock with respect to which the SAR is to be exercised and (c) the address to which the payment due under such SAR should be delivered. A Tandem SAR may be exercised only with respect to the shares of Stock for which its related Option is then exercisable. In accordance with applicable law, a Tandem SAR may be exercised upon whatever additional terms and conditions the Committee, in its sole discretion, imposes.
- 6.8 **Payment of SAR Amount.** Upon the exercise of a SAR, a Holder shall be entitled to receive payment from the Company in an amount determined by multiplying the excess of the Fair Market Value of a share of Stock on the date of exercise over the grant price of the SAR by the number of shares of Stock with respect to which the SAR is exercised. At the discretion of the Committee, the payment upon SAR exercise may be in cash, in Stock of equivalent value, in some combination thereof or in any other manner approved by the Committee in its sole discretion. The Committee's determination regarding the form of SAR payout shall be set forth in the Award Agreement pertaining to the grant of the SAR.
- 6.9 **Separation from Service**. Each Award Agreement shall set forth the extent to which the Holder of a SAR shall have the right to exercise the SAR following the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee, may be included in the Award Agreement entered into with the Holder, need not be uniform among all SARs issued pursuant to the Plan, and may reflect distinctions based on the reasons for termination or severance.
- 6.10 **Nontransferability of SARs.** Except as otherwise provided in a Holder's Award Agreement, no SAR granted under the Plan may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, except as otherwise provided in a Holder's Award Agreement, all SARs granted to a Holder under the Plan shall be exercisable during his or her lifetime only by the Holder, and after that time, by the Holder's heirs or estate. Any attempted assignment of a SAR in violation of this Section 6.10 shall be null and void.
  - 6.11 **No Rights as Stockholder.** A grantee of a SAR award, as such, shall have no rights as a stockholder.

6.12 **Restrictions on Stock Received.** The Committee may impose such conditions and/or restrictions on any shares of Stock received upon exercise of a SAR granted pursuant to the Plan as it may deem advisable or desirable. These restrictions may include, but shall not be limited to, a requirement that the Holder hold the shares of Stock received upon exercise of a SAR for a specified period of time.

#### ARTICLE VII RESTRICTED STOCK AWARDS

- 7.1 **Restricted Stock Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may make Awards of Restricted Stock under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any Restricted Stock Award shall be determined by the Committee in its sole discretion. If the Committee imposes vesting, forfeiture or transferability restrictions on a Holder's rights with respect to Restricted Stock, the Committee may issue such instructions to the Company's share transfer agent in connection therewith as it deems appropriate. The Committee may also cause the certificate for shares of Stock issued pursuant to a Restricted Stock Award to be imprinted with any legend which counsel for the Company considers advisable with respect to the restrictions or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law.
- 7.2 **Restricted Stock Award Agreement.** Each Restricted Stock Award shall be evidenced by an Award Agreement that contains any vesting, forfeiture and transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
- 7.3 **Holder's Rights as Stockholder.** Subject to the terms and conditions of the Plan, each recipient of a Restricted Stock Award shall have all the rights of a stockholder with respect to the shares of Restricted Stock included in the Restricted Stock Award during the Period of Restriction established for the Restricted Stock Award. Dividends paid with respect to Restricted Stock in cash or property other than shares of Stock or rights to acquire shares of Stock shall be paid to the recipient of the Restricted Stock Award currently. Dividends paid in shares of Stock or rights to acquire shares of Stock shall be added to and become a part of the Restricted Stock. During the Period of Restriction, certificates representing the Restricted Stock shall be registered in the Holder's name and bear a restrictive legend to the effect that ownership of such Restricted Stock, and the enjoyment of all rights appurtenant thereto, are subject to the restrictions, terms, and conditions provided in the Plan and the applicable Award Agreement. Such certificates shall be deposited by the recipient with the Secretary of the Company or such other officer or agent of the Company as may be designated by the Committee, together with all stock powers or other instruments of assignment, each endorsed in blank, which will permit transfer to the Company of all or any portion of the Restricted Stock which shall be forfeited in accordance with the Plan and the applicable Award Agreement.

## ARTICLE VIII RESTRICTED STOCK UNIT AWARDS

8.1 **Authority to Grant RSU Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant RSU Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any RSU Award shall be determined by the Committee in its sole discretion. The Committee shall maintain a bookkeeping ledger account which reflects the number of RSUs credited under the Plan for the benefit of a Holder.

- 8.2 **RSU** Award. An RSU Award shall be similar in nature to a Restricted Stock Award except that no shares of Stock are actually transferred to the Holder until a later date specified in the applicable Award Agreement. Each RSU shall have a value equal to the Fair Market Value of a share of Stock
- 8.3 **RSU Award Agreement.** Each RSU Award shall be evidenced by an Award Agreement that contains any Substantial Risk of Forfeiture, transferability and forfeiture restrictions, form and time of payment provisions and other provisions not inconsistent with the Plan as the Committee may specify.
- 8.4 **Dividend Equivalents.** An Award Agreement for an RSU Award may specify that the Holder shall be entitled to the payment of Dividend Equivalents under the Award.
- 8.5 **Form of Payment Under RSU Award.** Payment under an RSU Award shall be made in cash, shares of Stock or any combination thereof, as specified in the applicable Award Agreement.
- 8.6 **Time of Payment Under RSU Award.** A Holder's payment under an RSU Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the Fiscal Year in which the RSU Award payment is no longer subject to a Substantial Risk of Forfeiture or (b) at a time that is Permissible under Section 409A.
- 8.7 **Holder's Rights as Stockholder.** Each recipient of an RSU Award shall have no rights of a stockholder with respect to the Holder's RSUs. A Holder shall have no voting rights with respect to any RSU Awards.

# ARTICLE IX PERFORMANCE STOCK AWARDS AND PERFORMANCE UNIT AWARDS

- 9.1 **Authority to Grant Performance Stock Awards and Performance Unit Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Performance Stock Awards and Performance Unit Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any Performance Stock Award or Performance Unit Award shall be based upon the attainment of such Performance Goals as the Committee may determine; provided, however, that the performance period for any Performance Stock Award or Performance Unit Award shall not be less than one year. If the Committee imposes vesting or transferability restrictions on a Holder's rights with respect to Performance Stock Award or Performance Unit Awards, the Committee may issue such instructions to the Company's share transfer agent in connection therewith as it deems appropriate. The Committee may also cause the certificate for shares of Stock issued pursuant to a Performance Stock Award or Performance Unit Award to be imprinted with any legend which counsel for the Company considers advisable with respect to the restrictions or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law.
- 9.2 **Performance Goals.** A Performance Goal must be objective such that a third party having knowledge of the relevant facts could determine whether the goal is met. Unless and until the Committee proposes for stockholder vote and the stockholders approve a change in the general Performance Goals set forth in this Article IX, the Performance Goals upon which the payment or vesting of an Award to a Covered Employee that is intended to qualify as Performance-Based Compensation shall be limited to one or more of the following Performance Goals, which may be based on one or more business criteria that apply to the Holder, one or more business units or subsidiaries of the Company, or the Company as a whole: earnings per share, earnings per share

growth, total stockholder return, economic value added, cash return on capitalization, increased revenue, revenue ratios (per employee or per customer), net income, stock price, market share, return on equity, return on assets, return on capital, return on capital compared to cost of capital, return on capital employed, return on invested capital, stockholder value, net cash flow, operating income, earnings before interest and taxes, cash flow, cash flow from operations, cost ratios (per employee or per customer), proceeds from dispositions, project completion time and budget goals, net cash flow before financing activities, customer growth, and total market value. Goals may also be based on performance relative to a peer group of companies. Unless otherwise stated, such a Performance Goal need not be based upon an increase or positive result under a particular business criterion and could include, for example, maintaining the status quo or limiting economic losses (measured, in each case, by reference to specific business criteria). In interpreting Plan provisions applicable to Performance Goals and Performance Stock or Performance Unit Awards, it is intended that the Plan will conform with the standards of section 162(m) of the Code and Treasury Regulations § 1.162-27(e)(2)(i), and the Committee in establishing such goals and interpreting the Plan shall be guided by such provisions. Prior to the payment of any compensation based on the achievement of Performance Goals, the Committee must certify in writing that applicable Performance Goals and any of the material terms thereof were, in fact, satisfied. Subject to the foregoing provisions, the terms, conditions and limitations applicable to any Performance Stock or Performance Unit Awards made pursuant to the Plan shall be determined by the Committee.

- 9.3 **Time of Establishment of Performance Goals.** With respect to a Covered Employee, a Performance Goal for a particular Performance Stock Award or Performance Unit Award must be established by the Committee prior to the earlier to occur of (a) 90 days after the commencement of the period of service to which the Performance Goal relates or (b) the lapse of 25 percent of the period of service, and in any event while the outcome is substantially uncertain.
- 9.4 **Written Agreement.** Each Performance Stock Award or Performance Unit Award shall be evidenced by an Award Agreement that contains any vesting, transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
- 9.5 **Form of Payment Under Performance Unit Award.** Payment under a Performance Unit Award shall be made in cash, shares of Stock or any combination thereof, as specified in the applicable Award Agreement.
- 9.6 **Time of Payment Under Performance Unit Award.** A Holder's payment under a Performance Unit Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Performance Unit Award payment is no longer subject to a Substantial Risk of Forfeiture or (b) at a time that is Permissible under Section 409A.
- 9.7 **Holder's Rights as Stockholder With Respect to a Performance Stock Award.** Subject to the terms and conditions of the Plan, each Holder of a Performance Stock Award shall have all the rights of a stockholder with respect to the shares of Stock issued to the Holder pursuant to the Award during any period in which such issued shares of Stock are subject to forfeiture and restrictions on transfer, including without limitation, the right to vote such shares of Stock.
- 9.8 **Increases Prohibited.** Neither the Committee nor the Board may increase the amount of compensation payable under a Performance Stock or Performance Unit Award. If the time at which a Performance Stock Award or Performance Unit Award will vest or be paid is accelerated for any reason, the number of shares of Stock subject to, or the amount payable under, the Performance Stock or Performance Unit Award shall be reduced pursuant to Department of Treasury Regulation § 1.162-27(e)(2)(iii) to reasonably reflect the time value of money.

- 9.9 **Stockholder Approval.** No payments of Stock or cash will be made to a Covered Employee pursuant to this Article IX unless the stockholder approval requirements of Department of Treasury Regulation § 1.162-27(e)(4) are satisfied.
- 9.10 **Dividend Equivalents**. An Award Agreement for a Performance Unit Award may specify that the Holder shall be entitled to the payment of Dividend Equivalents under the Award.

# ARTICLE X DIRECTOR AWARDS

All Awards to Directors shall be determined by the Board.

## ARTICLE XI ANNUAL CASH INCENTIVE AWARDS

- 11.1 **Authority to Grant Annual Cash Incentive Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Annual Cash Incentive Awards under the Plan to key executive Employees who, by the nature and scope of their positions, regularly directly make or influence policy decisions which significantly impact the overall results or success of the Company in such amounts and upon such terms as the Committee shall determine. Subject to the following provisions in this Article XI, the amount of any Annual Cash Incentive Awards shall be based on the attainment of such Performance Goals as the Committee may determine and the term, conditions and limitations applicable to any Annual Cash Incentive Awards made pursuant to the Plan shall be determined by the Committee.
- 11.2 **Covered Employees**. The Performance Goals upon which the payment or vesting of an Annual Cash Incentive Award to a Covered Employee that is intended to qualify as Performance-Based Compensation must meet the requirements of Sections 9.2, 9.3, 9.8 and 9.9 as applied to such Annual Cash Incentive Award.
- 11.3 **Written Agreement.** Each Annual Cash Incentive Award shall be evidenced by an Award Agreement that contains any vesting, transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
  - 11.4 Form of Payment Under Annual Cash Incentive Award. Payment under an Annual Cash Incentive Award shall be made in cash.
- 11.5 **Time of Payment Under Annual Cash Incentive Award.** A Holder's payment under an Annual Cash Incentive Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Annual Cash Incentive Award payment is no longer subject to a Substantial Risk of Forfeiture or (b) at a time that is Permissible under Section 409A.

## ARTICLE XII OTHER STOCK-BASED AWARDS

12.1 **Authority to Grant Other Stock-Based Awards**. Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant other types of equity-based or equity-related Awards not otherwise described by the terms and provisions of the Plan (including the grant or offer for sale of

unrestricted shares of Stock) under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. Such Awards may involve the transfer of actual shares of Stock to Holders, or payment in cash or otherwise of amounts based on the value of shares of Stock and may include, without limitation, Awards designed to comply with or take advantage of the applicable local laws of jurisdictions other than the United States.

- 12.2 **Value of Other Stock-Based Award**. Each Other Stock-Based Award shall be expressed in terms of shares of Stock or units based on shares of Stock, as determined by the Committee.
- 12.3 **Payment of Other Stock-Based Award**. Payment, if any, with respect to an Other Stock-Based Award shall be made in accordance with the terms of the Award, in cash, shares of Stock or any combination thereof, as the Committee determines.
- 12.4 **Separation from Service**. The Committee shall determine the extent to which a Holder's rights with respect to Other Stock-Based Awards shall be affected by the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee and need not be uniform among all Other Stock-Based Awards issued pursuant to the Plan.
- 12.5 **Time of Payment of Other Stock-Based Award.** A Holder's payment under an Other Stock-Based Award shall be made at such time as is specified in the applicable Award Agreement. If a payment under the Award Agreement is subject to Section 409A, the Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Other Stock-Based Award payment is no longer subject to a Substantial Risk of Forfeiture or (b) at a time that is Permissible under Section 409A.

## ARTICLE XIII CASH-BASED AWARDS

- 13.1 **Authority to Grant Cash-Based Awards**. Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Cash-Based Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine.
  - 13.2 Value of Cash-Based Award. Each Cash-Based Award shall specify a payment amount or payment range as determined by the Committee.
- 13.3 **Payment of Cash-Based Award**. Payment, if any, with respect to a Cash-Based Award shall be made in accordance with the terms of the Award, in cash.
- 13.4 **Time of Payment of Cash-Based Award.** Payment under a Cash-Based Award shall be made at such time as is specified in the applicable Award Agreement. If a payment under the Award Agreement is subject to Section 409A, the Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Cash-Based Award payment is no longer subject to a Substantial Risk of Forfeiture or (b) at a time that is Permissible under Section 409A.
- 13.5 **Separation from Service**. The Committee shall determine the extent to which a Holder's rights with respect to Cash-Based Awards shall be affected by the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee and need not be uniform among all Cash-Based Awards issued pursuant to the Plan.

# ARTICLE XIV SUBSTITUTION AWARDS

Awards may be granted under the Plan from time to time in substitution for stock options and other awards held by employees of other entities who are about to become Employees, or whose employer is about to become an Affiliate as the result of a merger or consolidation of the Company with another corporation, or the acquisition by the Company of substantially all the assets of another corporation, or the acquisition by the Company of at least fifty percent (50%) of the issued and outstanding stock of another corporation as the result of which such other corporation will become a subsidiary of the Company. The terms and conditions of the substitute Awards so granted may vary from the terms and conditions set forth in the Plan to such extent as the Board at the time of grant may deem appropriate to conform, in whole or in part, to the provisions of the awards in substitution for which they are granted. If shares of Stock are issued under the Plan with respect to an Award granted under this Article such shares of Stock will not count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.

## ARTICLE XV ADMINISTRATION

- 15.1 **Awards.** The Plan shall be administered by the Committee or, in the absence of the Committee or in the case of awards issued to Directors, the Plan shall be administered by the Board. The members of the Committee (that is not itself the Board) shall serve at the discretion of the Board. The Committee shall have full and exclusive power and authority to administer the Plan and to take all actions that the Plan expressly contemplates or are necessary or appropriate in connection with the administration of the Plan with respect to Awards granted under the Plan.
- 15.2 **Authority of the Committee.** The Committee shall have full and exclusive power to interpret and apply the terms and provisions of the Plan and Awards made under the Plan, and to adopt such rules, regulations and guidelines for implementing the Plan as the Committee may deem necessary or proper, all of which powers shall be exercised in the best interests of the Company and in keeping with the objectives of the Plan. A majority of the members of the Committee shall constitute a quorum for the transaction of business relating to the Plan or Awards made under the Plan, and the vote of a majority of those members present at any meeting shall decide any question brought before that meeting. Any decision or determination reduced to writing and signed by a majority of the members shall be as effective as if it had been made by a majority vote at a meeting properly called and held. All questions of interpretation and application of the Plan, or as to Awards granted under the Plan, shall be subject to the determination, which shall be final and binding, of a majority of the whole Committee. No member of the Committee shall be liable for any act or omission of any other member of the Committee or for any act or omission on his or her own part, including but not limited to the exercise of any power or discretion given to him or her under the Plan, except those resulting from his or her own willful misconduct. In carrying out its authority under the Plan, the Committee shall have full and final authority and discretion, including but not limited to the following rights, powers and authorities to (a) determine the persons to whom and the time or times at which Awards will be made; (b) determine the number and exercise price of shares of Stock covered in each Award subject to the terms and provisions of the Plan; (c) determine the terms, provisions and conditions of each Award, which need not be identical and need not match the default terms set forth in the Plan; (d) accelerate the time at which any outstanding Award will vest; (e) prescribe, amend and rescind rules and regulations relating to administration of the Plan; and (f) make all other determinations and take all other actions deemed necessary, appropriate or advisable for the proper administration of the Plan.

The Committee may make an Award to an individual who the Company expects to become an Employee of the Company or any of its Affiliates within six (6) months after the date of grant of the Award, with the Award

being subject to and conditioned on the individual actually becoming an Employee within that time period and subject to other terms and conditions as the Committee may establish.

The Committee may correct any defect or supply any omission or reconcile any inconsistency in the Plan or in any Award to a Holder in the manner and to the extent the Committee deems necessary or desirable to further the Plan's objectives. Further, the Committee shall make all other determinations that may be necessary or advisable for the administration of the Plan. As permitted by law and the terms and provisions of the Plan, the Committee may delegate to one or more of its members or to one or more officers of the Company, and/or its Affiliates or to one or more agents or advisors such administrative duties or powers as it may deem advisable, and the Committee or any person to whom it has delegated duties or powers as aforesaid may employ one or more persons to render advice with respect to any responsibility the Committee or such person may have under the Plan. The Committee may, by resolution, authorize one or more officers of the Company to do one or more of the following on the same basis as can the Committee: (a) designate Employees to be recipients of Awards; (b) designate Third Party Service Providers to be recipients of Awards; and (c) determine the size of any such Awards; *provided, however*, (i) the Committee shall not delegate such responsibilities to any such officer for Awards granted to an Employee that is considered an Insider; (ii) the resolution providing such authorization sets forth the total number of Awards such officer(s) may grant; and (iii) the officer(s) shall report periodically to the Committee regarding the nature and scope of the Awards granted pursuant to the authority delegated. The Committee may employ attorneys, consultants, accountants, agents, and other persons, any of whom may be an Employee, and the Committee, the Company, and its officers and Board shall be entitled to rely upon the advice, opinions, or valuations of any such persons.

- 15.3 **Decisions Binding.** All determinations and decisions made by the Committee or the Board, as the case may be, pursuant to the provisions of the Plan and all related orders and resolutions of the Committee or the Board, as the case may be, shall be final, conclusive and binding on all persons, including the Company, its Affiliates, its stockholders, Holders and the estates and beneficiaries of Holders.
- 15.4 **No Liability.** Under no circumstances shall the Company, its Affiliates, the Board or the Committee incur liability for any indirect, incidental, consequential or special damages (including lost profits) of any form incurred by any person, whether or not foreseeable and regardless of the form of the act in which such a claim may be brought, with respect to the Plan or the Company's, its Affiliates', the Committee's or the Board's roles in connection with the Plan.

# ARTICLE XVI AMENDMENT OR TERMINATION OF PLAN

- 16.1 **Amendment, Modification, Suspension, and Termination**. Subject to Section 16.2, the Board may, at any time and from time to time, alter, amend, modify, suspend, or terminate the Plan and the Committee may, at any time and from time to time, alter, amend, modify, suspend, or terminate any Award Agreement in whole or in part; provided, however, that, without the prior approval of the Company's stockholders and except as provided in Section 4.5, the Committee shall not directly or indirectly lower the Option Price of a previously granted Option or the grant price of a previously granted SAR, cancel a previously granted Option or previously granted SAR for a payment of cash or other property if the aggregate fair market value of such Award is less than the aggregate Option Price of such Award in the case of an Option or the aggregate grant price of such Award in the case of a SAR, and no amendment of the Plan shall be made without stockholder approval if stockholder approval is required by applicable law or stock exchange rules.
- 16.2 **Awards Previously Granted**. Notwithstanding any other provision of the Plan to the contrary, no termination, amendment, suspension, or modification of the Plan or an Award Agreement shall adversely

affect in any material way any Award previously granted under the Plan, without the written consent of the Holder holding such Award.

## ARTICLE XVII MISCELLANEOUS

- 17.1 **Unfunded Plan/No Establishment of a Trust Fund.** Holders shall have no right, title, or interest whatsoever in or to any investments that the Company or any of its Affiliates may make to aid in meeting obligations under the Plan. Nothing contained in the Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any Holder, beneficiary, legal representative, or any other person. To the extent that any person acquires a right to receive payments from the Company under the Plan, such right shall be no greater than the right of an unsecured general creditor of the Company. All payments to be made hereunder shall be paid from the general funds of the Company and no special or separate fund shall be established and no segregation of assets shall be made to assure payment of such amounts, except as expressly set forth in the Plan. No property shall be set aside nor shall a trust fund of any kind be established to secure the rights of any Holder under the Plan. The Plan is not intended to be subject to the Employee Retirement Income Security Act of 1974, as amended.
- 17.2 **No Employment Obligation.** The granting of any Award shall not constitute an employment or service contract, express or implied, and shall not impose upon the Company or any Affiliate any obligation to employ or continue to employ, or to utilize or continue to utilize the services of, any Holder. The right of the Company or any Affiliate to terminate the employment of, or the provision of services by, any person shall not be diminished or affected by reason of the fact that an Award has been granted to him, and nothing in the Plan or an Award Agreement shall interfere with or limit in any way the right of the Company or its Affiliates to terminate any Holder's employment or service relationship at any time or for any reason not prohibited by law.
- 17.3 **Tax Withholding.** The Company or any Affiliate shall be entitled to deduct from other compensation payable to each Holder any sums required by federal, state, local or foreign tax law to be withheld with respect to the vesting or exercise of an Award or lapse of restrictions on an Award. In the alternative, the Company may require the Holder (or other person validly exercising the Award) to pay such sums for taxes directly to the Company or any Affiliate in cash or by check within one day after the date of vesting, exercise or lapse of restrictions. In the discretion of the Committee, and with the consent of the Holder, the Company may reduce the number of shares of Stock issued to the Holder upon such Holder's exercise of an Award or the vesting of an Award to satisfy the tax withholding obligations of the Company or an Affiliate; *provided* that the Fair Market Value of the shares of Stock held back shall not exceed the Company's or the Affiliate's Minimum Statutory Tax Withholding Obligation.

The Committee may, in its discretion, permit a Holder to satisfy any Minimum Statutory Tax Withholding Obligation arising upon the vesting of or payment under an Award by delivering to the Holder a reduced number of shares of Stock in the manner specified herein. If permitted by the Committee and acceptable to the Holder, at the time of vesting of shares under the Award, the Company shall (a) calculate the amount of the Company's or an Affiliate's Minimum Statutory Tax Withholding Obligation on the assumption that all such shares of Stock vested under the Award are made available for delivery, (b) reduce the number of such shares of Stock made available for delivery so that the Fair Market Value of the shares of Stock withheld on the vesting date approximates the Company's or an Affiliate's Minimum Statutory Tax Withholding Obligation and (c) in lieu of the withheld shares of Stock, remit cash to the United States Treasury and/or other applicable governmental authorities, on behalf of the Holder, in the amount of the Minimum Statutory Tax Withholding Obligation. The Company shall withhold only whole shares of Stock to satisfy its Minimum Statutory Tax Withholding Obligation, the Company shall withhold shares of Stock does not equal the amount of the Minimum Statutory Tax Withholding Obligation, the Company shall withhold shares of Stock with a Fair

Market Value slightly less than the amount of the Minimum Statutory Tax Withholding Obligation and the Holder must satisfy the remaining minimum withholding obligation in some other manner permitted under this Section 17.3. The withheld shares of Stock not made available for delivery by the Company shall be retained as treasury shares or will be cancelled and the Holder's right, title and interest in such shares of Stock shall terminate.

The Company shall have no obligation upon vesting or exercise of any Award or lapse of restrictions on an Award until the Company or an Affiliate has received payment sufficient to cover the Minimum Statutory Tax Withholding Obligation with respect to that vesting, exercise or lapse of restrictions. Neither the Company nor any Affiliate shall be obligated to advise a Holder of the existence of the tax or the amount which it will be required to withhold.

- 17.4 **Indemnification of the Committee.** The Company shall indemnify each past, present and future member of the Committee against, and each member of the Committee shall be entitled without further action on his or her part to indemnity from the Company for, all expenses (including attorney's fees, the amount of judgments and the amount of approved settlements made with a view to the curtailment of costs of litigation, other than amounts paid to the Company itself) reasonably incurred by such member in connection with or arising out of any action, suit or proceeding in which such member may be involved by reason of such member being or having been a member of the Committee, whether or not he or she continues to be a member of the Committee at the time of incurring the expenses, including, without limitation, matters as to which such member shall be finally adjudged in any action, suit or proceeding to have been negligent in the performance of such member's duty as a member of the Committee. However, this indemnity shall not include any expenses incurred by any member of the Committee in respect of matters as to which such member shall be finally adjudged in any action, suit or proceeding to have been guilty of willful misconduct in the performance of his or her duty as a member of the Committee. In addition, no right of indemnification under the Plan shall be available to or enforceable by any member of the Committee unless, within 60 days after institution of any action, suit or proceeding, such member shall have offered the Company, in writing, the opportunity to handle and defend same at its own expense. This right of indemnification shall inure to the benefit of the heirs, executors or administrators of each member of the Committee and shall be in addition to all other rights to which a member of the Committee may be entitled as a matter of law, contract or otherwise.
- 17.5 **Gender and Number.** If the context requires, words of one gender when used in the Plan shall include the other and words used in the singular or plural shall include the other.
- 17.6 **Severability**. In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- 17.7 **Headings.** Headings of Articles and Sections are included for convenience of reference only and do not constitute part of the Plan and shall not be used in construing the terms and provisions of the Plan.
- 17.8 **Other Compensation Plans.** The adoption of the Plan shall not affect any other option, incentive or other compensation or benefit plans in effect for the Company or any Affiliate, nor shall the Plan preclude the Company from establishing any other forms of incentive compensation arrangements for Employees, Directors or Third Party Service Providers.
- 17.9 **Retirement and Welfare Plans**. Neither Awards made under the Plan nor shares of Stock or cash paid pursuant to such Awards, may be included as "compensation" for purposes of computing the benefits payable to any person under the Company's or any Affiliate's retirement plans (both qualified and non-qualified) or welfare benefit plans unless such other plan expressly provides that such compensation shall be taken into account in computing a participant's benefit.

- 17.10 **Other Awards.** The grant of an Award shall not confer upon the Holder the right to receive any future or other Awards under the Plan, whether or not Awards may be granted to similarly situated Holders, or the right to receive future Awards upon the same terms or conditions as previously granted.
- 17.11 **Successors.** All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business and/or assets of the Company.
- 17.12 **Law Limitations/Governmental Approvals**. The granting of Awards and the issuance of shares of Stock under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
- 17.13 **Delivery of Title**. The Company shall have no obligation to issue or deliver evidence of title for shares of Stock issued under the Plan prior to (a) obtaining any approvals from governmental agencies that the Company determines are necessary or advisable; and (b) completion of any registration or other qualification of the Stock under any applicable national or foreign law or ruling of any governmental body that the Company determines to be necessary or advisable.
- 17.14 **Inability to Obtain Authority**. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any shares of Stock hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such shares of Stock as to which such requisite authority shall not have been obtained.
- 17.15 **Investment Representations**. The Committee may require any person receiving Stock pursuant to an Award under the Plan to represent and warrant in writing that the person is acquiring the shares of Stock for investment and without any present intention to sell or distribute such Stock.
- 17.16 **Persons Residing Outside of the United States.** Notwithstanding any provision of the Plan to the contrary, in order to comply with the laws in other countries in which the Company or any of its Affiliates operates or has employees, the Committee, in its sole discretion, shall have the power and authority to (a) determine which Affiliates shall be covered by the Plan; (b) determine which persons employed outside the United States are eligible to participate in the Plan; (c) amend or vary the terms and provisions of the Plan and the terms and conditions of any Award granted to persons who reside outside the United States; (d) establish subplans and modify exercise procedures and other terms and procedures to the extent such actions may be necessary or advisable any subplans and modifications to Plan terms and procedures established under this Section 17.16 by the Committee shall be attached to the Plan document as Appendices; and (e) take any action, before or after an Award is made, that it deems advisable to obtain or comply with any necessary local government regulatory exemptions or approvals. Notwithstanding the above, the Committee may not take any actions hereunder, and no Awards shall be granted, that would violate the Exchange Act, the Code, any securities law or governing statute or any other applicable law.
- 17.17 **Arbitration of Disputes.** Any controversy arising out of or relating to the Plan or an Award Agreement shall be resolved by arbitration conducted in Houston, Texas pursuant to the arbitration rules of the American Arbitration Association. The arbitration shall be final and binding on the parties.
- 17.18 **No Fractional Shares**. No fractional shares of Stock shall be issued or delivered pursuant to the Plan or any Award. The Committee shall determine whether cash, additional Awards, or other property shall be issued or paid in lieu of fractional shares of Stock or whether such fractional shares or any rights thereto shall be forfeited or otherwise eliminated.

17.19 **Governing Law.** The provisions of the Plan and the rights of all persons claiming thereunder shall be construed, administered and governed under the laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of the Plan to the substantive law of another jurisdiction. Unless otherwise provided in the Award Agreement, recipients of an Award under the Plan are deemed to submit to the sole and exclusive jurisdiction and venue of the federal or state courts of the State of Texas to resolve any and all issues that may arise out of or relate to the Plan or any related Award Agreement.

# FIRST AMENDMENT TO THE RIGNET, INC. 2010 OMNIBUS INCENTIVE PLAN

THIS FIRST AMENDMENT is made by RigNet, Inc. (the "Sponsor"),

#### WITNESSETH:

**WHEREAS**, the Sponsor adopted on May 26, 2010, and continues to sponsor and maintain the plan known as the "RigNet, Inc. 2010 Omnibus Incentive Plan"; and

WHEREAS, the Board of Directors of the Sponsor retained the right in Section 16.1 of the Plan to amend the Plan from time to time; and

WHEREAS, the Board of Directors of the Sponsor approved resolutions on March 10, 2016, to amend the Plan as set forth below;

NOW, THEREFORE, the Sponsor agrees that, effective as set forth below, the Plan is amended as set forth below:

- 1. Effective upon the approval of a majority of the shareholders of the Sponsor represented in person or by proxy at the next Annual Meeting of Shareholders, Section 4.1(a) of the Plan is hereby amended and restated in its entirety to read as follows:
  - (a) The aggregate number of shares of Stock with respect to which Awards may be granted under the Plan is 4,000,000 (the "Authorized Shares").
  - 2. Effective March 10, 2016, Section 4.1(d) of the Plan is hereby amended and restated to be renamed as Section 4.1(e).
  - 3. Effective March 10, 2016, a new Section 4.1(d) is hereby added to the Plan to read as follows:
- (d) The aggregate dollar value of shares of Stock that may be granted under the Plan to any director in any Fiscal Year shall be no more than \$300,000; provided, however, that with respect to any Chairman or Vice Chairman of the board, that dollar limit shall be \$400,000.
  - 3. Effective March 10, 2016, Section 4.2(a) and (b) of the Plan are hereby amended and restated in their entirety to read as follows:

## 4.2 Shares That Count Against Limit.

- (a) Effective on and after March 10, 2016, if shares of Stock are withheld from payment of an Award to satisfy tax obligations with respect to the Award, such shares of Stock will count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.
- (b) Effective on and after March 10, 2016, if shares of Stock are tendered in payment of the Option Price of an Option, such shares of Stock will count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.

Page 1 of 2

4. Effective March 10, 2016, Section 9.2 of the Plan is hereby amended and restated in its entirety to read as follows:

9.2 Performance Goals. A Performance Goal must be objective such that a third party having knowledge of the relevant facts could determine whether the goal is met. Unless and until the Committee proposes for stockholder vote and the stockholders approve a change in the general Performance Goals set forth in this Article IX, the Performance Goals upon which the payment or vesting of an Award to a Covered Employee that is intended to qualify as Performance-Based Compensation shall be limited to one or more of the following Performance Goals, which may be based on one or more business criteria that apply to the Holder, one or more business units or subsidiaries of the Company, or the Company as a whole: earnings, earnings before interest and taxes, earnings before interest, taxes, depreciation, and amortization, earnings before any one of, or combination of two or more of, interest, taxes, depreciation, amortization and/or any other financial adjustment to earnings set forth in the Company's audited financial statements that is allowed under generally accepted accounting principles, adjusted earnings before interest, taxes, depreciation and amortization, net earnings, earnings per share, earnings per share growth, economic value added, economic value, operating profits, net operating profit, net profits, profit return, gross margin, profit margins, profit before tax, operating margin, cash return on capitalization, operating expense, operating expense as a percentage of revenue, revenue, increase in revenue, revenue ratios (including per employee or per customer), net revenue, billings, net income, stock price, market share, return on equity, return on assets, return on capital, return on capital compared to cost of capital, return on capital employed, return on invested capital, debt to capital ratio, total stockholder return, stockholder return, stockholder value, growth in stockholder value relative to a pre-determined index, financial return ratio, operating income, cash flow, net cash flow, cash flow from operations, free cash flow, cash value added performance, cost reductions, cost ratios (per employee or per customer), proceeds from dispositions, project completion time and budget goals, net cash flow before financing activities, customer growth, total market value, or people value added. The Committee may select one criterion or multiple criteria for measuring performance. Goals may also be based on performance relative to a peer group of companies. Unless otherwise stated, such a Performance Goal need not be based upon an increase or positive result under a particular business criterion and could include, for example, maintaining the status quo or limiting economic losses (measured, in each case, by reference to specific business criteria). In interpreting Plan provisions applicable to Performance Goals and Performance Stock or Performance Unit Awards, it is intended that the Plan will conform with the standards of section 162(m) of the Code and Treasury Regulations § 1.162-27(e)(2)(i), and the Committee in establishing such goals and interpreting the Plan shall be guided by such provisions. Prior to the payment of any compensation based on the achievement of Performance Goals, the Committee must certify in writing that applicable Performance Goals and any of the material terms thereof were, in fact, satisfied. Subject to the foregoing provisions, the terms, conditions and limitations applicable to any Performance Stock or Performance Unit Awards made pursuant to the Plan shall be determined by the Committee.

Approved by the Board of Directors on March 10, 2016.

# RIGNET, INC. OMNIBUS INCENTIVE PLAN

(As Adopted May 8. 2019)

#### ARTICLE I

### ESTABLISHMENT, PURPOSE AND DURATION

- 1.1 **Establishment.** The Company hereby establishes an incentive compensation plan, to be known as the "*RigNet, Inc. Omnibus Incentive Plan*", as set forth in this document. The Plan permits the grant of Incentive Stock Options, Nonqualified Stock Options, SARs, Restricted Stock, RSUs, Performance Stock Awards, Performance Unit Awards, Annual Cash Incentive Awards, Other Stock-Based Awards and Cash-Based Awards. The Plan is effective as of [DATE] (the "*Effective Date*"); provided that the Company's stockholders approve the adoption of the Plan within twelve (12) months after the date of adoption of the Plan by the Board.
- 1.2 **Purpose of the Plan.** The Plan is intended to advance the best interests of the Company, its Affiliates and its stockholders by providing those persons who have substantial responsibility for the management and growth of the Company and its Affiliates with additional performance incentives and an opportunity to obtain or increase their proprietary interest in the Company, thereby encouraging them to continue in their employment or affiliation with the Company or its Affiliates.
- 1.3 **Duration of Plan.** The Plan shall continue indefinitely until it is terminated pursuant to Section 16.1. No Award may be granted under the Plan on or after the tenth (10<sup>th</sup>) anniversary of the Effective Date. The applicable provisions of the Plan will continue in effect with respect to an Award granted under the Plan for as long as such Award remains outstanding. Notwithstanding the foregoing, no Incentive Stock Option may be granted under the Plan on or after the date that is ten (10) years from the earlier of (a) adoption of the Plan by the Board and (b) the Effective Date.

### **ARTICLE II**

#### **DEFINITIONS**

Each word and phrase defined in this Article shall have the meaning set out below throughout the Plan, unless the context in which any such word or phrase appears reasonably requires a broader, narrower or different meaning.

- 2.1 "Affiliate" means any corporation, partnership, limited liability company or association, trust or other entity or organization, which, directly or indirectly, controls, is controlled by, or is under common control with, the Company. For purposes of the preceding sentence, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any entity or organization, shall mean the possession, directly or indirectly, of the power (a) to vote more than fifty percent (50%) of the securities having ordinary voting power for the election of directors or comparable individuals of the controlled entity or organization, or (b) to direct or cause the direction of the management and policies of the controlled entity or organization, whether through the ownership of voting securities or by contract or otherwise.
  - 2.2 "Annual Cash Incentive Award" means an Award granted pursuant to Article XI.
  - 2.3 "Authorized Shares" shall have the meaning ascribed to that term in Section 4.1(a).
- 2.4 "Award" means, individually or collectively, a grant under the Plan of an Incentive Stock Option, a Nonqualified Stock Option, a SAR, Restricted Stock, a RSU, a Performance Stock Award, a Performance Unit Award, an Annual Cash Incentive Award, an Other Stock-Based Award or a Cash-Based Award, in each case subject to the terms and provisions of the Plan.
  - 2.5 "Award Agreement" means an agreement that sets forth the terms and conditions applicable to an Award granted under the Plan.

- 2.6 "Beneficial Owner" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.
  - 2.7 "Board" means the board of directors of the Company.
  - 2.8 "Cash-Based Award" means an Award granted pursuant to Article XIII.
- 2.9 "Change in Control" means (i) a change in ownership occurring as the result of a person or group acquiring Stock of the Company, which, when combined with the Stock held by such person or group, constitutes more than eighty percent (80%) of the total fair market value or total voting power of the Company; provided the person or group was not considered as owning more than eighty percent (80%) of the value or voting power prior to the acquisition; (ii) a change in effective control of the Company occurring as the result of the replacement of a majority of the members of the Board by directors whose appointment or election is not endorsed by a majority of the members of the Board before the date of the appointment or election; or (iii) a change in the ownership of a substantial portion of the assets of the Company occurring as the result of a person or group acquiring assets from the Company that have a total gross fair market value equal to or more than eighty percent (80%) of the total gross fair market value of all the assets of the Company immediately prior to such acquisition. The determination of whether a Change of Control has occurred will be made in accordance with Code Section 409A and the regulations thereunder.
  - 2.10 "Code" means the United States Internal Revenue Code of 1986, as amended from time to time.
- 2.11 "Committee" means (a) in the case of an Award granted to a Director, the Board, and (b) in the case of any other Award granted under the Plan, the Compensation Committee of the Board or, if the Compensation Committee of the Board chooses to delegate it duties, a committee of at least two (2) persons who are members of the Compensation Committee of the Board and are appointed by the Compensation Committee of the Board to administer the Plan. The Board may appoint a special committee consisting of one or more Directors for the purpose of granting certain specified Awards under the Plan. As to Awards, grants or other transactions that are authorized by the Committee and that are intended to be exempt under Rule 16b-3 of the General Rules and Regulations under the Exchange Act, the requirements of Rule 16b-3(d)(1) of the General Rules and Regulations under the Exchange Act with respect to committee action must also be satisfied.
  - 2.12 "Company" means RigNet, Inc., a Delaware corporation, or any successor (by reincorporation, merger or otherwise).
  - 2.13 "*Director*" means a director of the Company who is not an Employee.
- 2.14 "*Disability*" or "*Disabled*" means a determination by the Company's long-term disability carrier that a Holder is disabled in accordance with the Company's long-term disability insurance plan, provided the definition of disability applied under such plan complies with the requirements of Treas. Reg. Section 1.409A-3(i)(4), or, in the case of a Holder who is not covered under such plan, a determination made by the Social Security Administration that the Holder is totally disabled.
  - 2.15 "Dividend Equivalent" means a payment equivalent in amount to dividends paid to the Company's stockholders.
  - 2.16 "*Effective Date*" shall have the meaning ascribed to that term in Section 1.1.
- 2.17 "*Employee*" means (a) a person employed by the Company or any Affiliate as a common law employee, or (b) a person who has agreed to become a common law employee of the Company or any Affiliate and is expected to become such within six (6) months from the date of a determination made for purposes of the Plan.
  - 2.18 "Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor act.
  - 2.19 "Fair Market Value" of the Stock as of any particular date means,
    - (a) if the Stock is traded on a stock exchange,
      - (i) and if the Stock is traded on that date, the closing sale price of the Stock on that date; or

(ii) and if the Stock is not traded on that date, the closing sale price of the Stock on the last trading date immediately preceding that date;

as reported on the principal securities exchange on which the Stock is traded; or

- (b) if the Stock is traded in the over-the-counter market,
  - (i) and if the Stock is traded on that date, the average between the high bid and low asked price on that date; or
  - (ii) and if the Stock is not traded on that date, the average between the high bid and low asked price on the last trading date immediately preceding that date;

as reported in such over-the-counter market; provided, however, that (x) if the Stock is not so traded, or (y) if, in the discretion of the Committee, another means of determining the fair market value of a share of Stock at such date shall be necessary or advisable, the Committee may provide for another method or means for determining such fair market value, which method or means shall comply with the requirements of a reasonable valuation method as described under Section 409A.

- 2.20 "Fiscal Year" means the calendar year.
- 2.21 "Freestanding SAR" means a SAR that is granted independently of any Options, as described in Article VI.
- 2.22 "Holder" means a person who has been granted an Award or any person who is entitled to receive shares of Stock or cash under an Award.
- 2.23 "*Incentive Stock Option*" or "*ISO*" means an option to purchase Stock granted pursuant to Article V that is designated as an incentive stock option and that is intended to satisfy the requirements of section 422 of the Code.
- 2.24 "*Insider*" shall mean an individual who is, on the relevant date, an officer, a Director, or more than ten percent (10%) Beneficial Owner of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, as determined by the Board in accordance with Section 16 of the Exchange Act.
  - 2.25 "Mature Shares" means shares of Stock that the Holder has held for at least six (6) months.
- 2.26 "Nonqualified Stock Option" or "NQSO" means a "nonqualified stock option" to purchase Stock granted pursuant to Article V that does not satisfy the requirements of section 422 of the Code.
  - 2.27 "Option" means an Incentive Stock Option or a Nonqualified Stock Option.
  - 2.28 "Option Price" shall have the meaning ascribed to that term in Section 5.4.
- 2.29 "Other Stock-Based Award" means an equity-based or equity-related Award not otherwise described by the terms and provisions of the Plan that is granted pursuant to Article XII.
- 2.30 "Parent Corporation" means any corporation (other than the Company) in an unbroken chain of corporations ending with the Company if, at the time of the action or transaction, each of the corporations other than the Company owns stock possessing fifty percent (50%) or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
- 2.31 "*Performance Goals*" means one or more of the criteria described in Section 9.2 on which the performance goals applicable to an Award are based.

- 2.32 "Performance Stock Award" means an Award designated as a performance stock award granted to a Holder pursuant to Article IX.
- 2.33 "Performance Unit Award" means an Award designated as a performance unit award granted to a Holder pursuant to Article IX.
- 2.34 "*Period of Restriction*" means the period during which Restricted Stock is subject to a substantial risk of forfeiture (based on the passage of time, the achievement of Performance Goals, or upon the occurrence of other events as determined by the Committee, in its discretion), as provided in Article VII.
  - 2.35 "Plan" means the RigNet, Inc. 2019 Omnibus Incentive Plan, as set forth in this document as it may be amended from time to time.
  - 2.36 "Restricted Stock" means shares of restricted Stock issued or granted under the Plan pursuant to Article VII.
  - 2.37 "Restricted Stock Award" means an authorization by the Committee to issue or transfer Restricted Stock to a Holder.
  - 2.38 "RSU" means a restricted stock unit credited to a Holder's ledger account maintained by the Company pursuant to Article VIII.
  - 2.39 "RSU Award" means an Award granted pursuant to Article VIII.
  - 2.40 "SAR" means a stock appreciation right granted under the Plan pursuant to Article VI.
- 2.41 "Section 409A" means section 409A of the Code and the regulations and other guidance promulgated by the United States Department of Treasury and/or the United States Internal Revenue Service under section 409A of the Code, or any successor statute.
- 2.42 "Separation from Service" means, except as otherwise provided in the case of an ISO in the following sentence of this Section 2.42, the termination of the Award recipient's employment or service relationship with the Company and all Affiliates as determined under Section 409A. "Separation from Service" means, in the case of an ISO, the termination of the Employee's employment relationship with all of the Company, any Parent Corporation, any Subsidiary Corporation and any parent or subsidiary corporation (within the meaning of section 422(a)(2) of the Code) of any such corporation that issues or assumes an ISO in a transaction to which section 424(a) of the Code applies.
- 2.43 "*Stock*" means the common stock of the Company, \$0.001 par value per share (or such other par value as may be designated by act of the Company's stockholders).
- 2.44 "*Subsidiary Corporation*" means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if, at the time of the action or transaction, each of the corporations other than the last corporation in an unbroken chain owns stock possessing fifty percent (50%) or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
  - 2.45 "Substantial Risk of Forfeiture" shall have the meaning ascribed to that term in Section 409A.
- 2.46 "*Tandem SAR*" means a SAR that is granted in connection with a related Option pursuant to Article VI herein, the exercise of which shall require forfeiture of the right to purchase a share of Stock under the related Option (and when a share of Stock is purchased under the Option, the Tandem SAR shall similarly be canceled).
- 2.47 "**Ten Percent Stockholder**" means an individual, who, at the time the applicable Option is granted, owns stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Parent Corporation or Subsidiary Corporation. An individual shall be considered as owning the stock owned, directly or indirectly, by or for his or her brothers and sisters (whether by the whole or half-blood), spouse, ancestors, and lineal descendants; and stock owned, directly or indirectly, by or for a corporation, partnership, estate, or trust, shall be considered as being owned proportionately by or for its stockholders, partners, or beneficiaries.

2.48 "*Third Party Service Provider*" means any consultant, agent, representative, advisor, or independent contractor who renders services to the Company or an Affiliate that (a) are not in connection with the offer and sale of the Company's securities in a capital raising transaction; and (b) do not directly or indirectly promote or maintain a market for the Company's securities, or any other person as determined by the Committee.

### ARTICLE III

#### **ELIGIBILITY**

Except as otherwise specified in this Article III, the persons who are eligible to receive Awards under the Plan are Employees, Directors and Third Party Service Providers; provided, however, that only those persons who are, on the dates of grant, employees of the Company or any Parent Corporation or Subsidiary Corporation are eligible for grants of Incentive Stock Options under the Plan.

### ARTICLE IV

## GENERAL PROVISIONS RELATING TO AWARDS

- 4.1 **Authority to Grant Awards**. The Committee may grant Awards to those Employees, Directors and Third Party Service Providers as the Committee shall from time to time determine, under the terms and conditions of the Plan. Subject only to any applicable limitations set out in the Plan, the number of shares of Stock or other value to be covered by any Award to be granted under the Plan shall be as determined by the Committee in its sole discretion.
  - (a) The aggregate number of shares of Stock with respect to which Awards may be granted under the Plan is 2,175,00 less one share of Stock for every one share of Stock granted under the RigNet. Inc. 2010 Omnibus Incentive Plan after December 31, 2018. Upon effectiveness of the Plan, no further awards will be made under the 2010 Omnibus Incentive Plan. (the "*Authorized Shares*").
    - (b) The aggregate number of shares of Stock with respect to which ISOs may be granted under the Plan is equal to the Authorized Shares.
  - (c) The maximum number of shares of Stock with respect to which ISOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which NQSOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which SARs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Stock Awards may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Unit Awards payable in cash may be granted to an Employee during a Fiscal Year, determined as of the dates of grants of the Performance Unit Awards, is \$3,000,000. The maximum amount that may be paid under Annual Cash Incentive Award(s) granted to an Employee during a Fiscal Year is \$3,000,000.
  - (d) The aggregate dollar value of shares of Stock that may be granted under the Plan to any director in any Fiscal Year shall be no more than \$300,000; provided, however, that with respect to any Chairman or Vice Chairman of the board, that dollar limit shall be \$400,000.
  - (e) Each of the foregoing numerical limits stated in this Section 4.2 shall be subject to adjustment in accordance with the provisions of Section 4.5.

### 4.2 Shares That Count Against Limit.

- (a) If shares of Stock are withheld from payment of an Award to satisfy tax obligations with respect to the Award, such shares of Stock will count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.
- (b) If shares of Stock are tendered in payment of the Option Price of an Option, such shares of Stock will count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.
- (c) To the extent that any outstanding Award is forfeited or cancelled for any reason or is settled in cash in lieu of shares of Stock, the shares of Stock allocable to such portion of the Award may again be subject to an Award granted under the Plan.
- (d) When a SAR is settled in shares of Stock, the number of shares of Stock subject to the SAR under the SAR Award Agreement will be counted against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan as one share for every share subject to the SAR, regardless of the number of shares used to settle the SAR upon exercise.
- (e) The maximum number of shares of Stock available for issuance under the Plan shall not be reduced to reflect any dividends or Dividend Equivalents that are reinvested into additional shares of Stock or credited as additional Restricted Stock, Restricted Stock Units, Performance Shares, or other Stock-Based Awards.
- 4.3 **Non-Transferability.** Except as specified in the applicable Award Agreements or in domestic relations court orders, an Award shall not be transferable by the Holder other than by will or under the laws of descent and distribution, and shall be exercisable, during the Holder's lifetime, only by him or her. Any attempted assignment of an Award in violation of this Section shall be null and void. In the discretion of the Committee, any attempt to transfer an Award other than under the terms of the Plan and the applicable Award Agreement may terminate the Award.
- 4.4 **Requirements of Law.** The Company shall not be required to sell or issue any shares of Stock under any Award if issuing those shares of Stock would constitute or result in a violation by the Holder or the Company of any provision of any law, statute or regulation of any governmental authority. Specifically, in connection with any applicable statute or regulation relating to the registration of securities, upon exercise of any Option or pursuant to any other Award, the Company shall not be required to issue any shares of Stock unless the Committee has received evidence satisfactory to it to the effect that the Holder will not transfer the shares of Stock except in accordance with applicable law, including receipt of an opinion of counsel satisfactory to the Company to the effect that any proposed transfer complies with applicable law. The determination by the Committee on this matter shall be final, binding and conclusive. The Company may, but shall in no event be obligated to, register any shares of Stock covered by the Plan pursuant to applicable securities laws of any country or any political subdivision. In the event the shares of Stock issuable on exercise of an Option or pursuant to any other Award are not registered, the Company may imprint on the certificate evidencing the shares of Stock any legend that counsel for the Company considers necessary or advisable to comply with applicable law, or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law. The Company shall not be obligated to take any other affirmative action in order to cause or enable the exercise of an Option or any other Award, or the issuance of shares of Stock pursuant thereto, to comply with any law or regulation of any governmental authority.

## 4.5 Changes in the Company's Capital Structure.

(a) The existence of outstanding Awards shall not affect in any way the right or power of the Company or its stockholders to make or authorize any or all adjustments, recapitalizations, reorganizations or other changes in the Company's capital structure or its business, any merger or consolidation of the Company, any issue of bonds, debentures, preferred or prior preference shares ahead of or affecting the Stock or Stock rights, the dissolution or liquidation of the Company, any sale or transfer of all or any part of its assets or business or any other corporate act or proceeding, whether of a similar character or otherwise.

- (b) If the Company shall effect a subdivision or consolidation of Stock or other capital readjustment, the payment of a Stock dividend, or other increase or reduction of the number of shares of Stock outstanding, without receiving compensation therefor in money, services or property, then (i) the number, class or series and per share price of Stock subject to outstanding Awards under the Plan shall be appropriately adjusted in such a manner as to entitle a Holder to receive upon exercise of an Award, for the same aggregate cash consideration, the equivalent total number and class or series of Stock the Holder would have received had the Holder exercised his or her Award in full immediately prior to the event requiring the adjustment; and (ii) the number and class or series of Stock then reserved to be issued under the Plan shall be adjusted by substituting for the total number and class or series of Stock then reserved, that number and class or series of Stock that would have been received by the owner of an equal number of outstanding shares of Stock of each class or series of Stock as the result of the event requiring the adjustment.
- (c) If while unexercised Awards remain outstanding under the Plan a Change in Control occurs, then, except as otherwise provided in an Award Agreement or another agreement between the Holder and the Company (provided that such exceptions shall not apply in the case of a reincorporation merger), or as a result of the Committee's effectuation of one or more of the alternatives described below, there shall be no acceleration of the time at which any Award then outstanding may be exercised, and no later than ten (10) days after the approval by the stockholders of the Company of such Change in Control, the Committee, acting in its sole and absolute discretion without the consent or approval of any Holder, shall act to effect one or more of the following alternatives, which may vary among individual Holders and which may vary among Awards held by any individual Holder (provided that, with respect to a reincorporation merger in which Holders of the Company's ordinary shares will receive one ordinary share of the successor corporation for each ordinary share of the Company, none of such alternatives shall apply and, without Committee action, each Award shall automatically convert into a similar award of the successor corporation exercisable for the same number of ordinary shares of the Successor as the Award was exercisable for ordinary shares of Stock of the Company):
  - (1) accelerate the time at which some or all of the Awards then outstanding may be exercised so that such Awards may be exercised in full for a limited period of time on or before a specified date (before or after such Change in Control) fixed by the Committee, after which specified date all such Awards that remain unexercised and all rights of Holders thereunder shall terminate;
  - (2) require the mandatory surrender to the Company by all or selected Holders of some or all of the then outstanding Awards held by such Holders (irrespective of whether such Awards are then exercisable under the provisions of the Plan or the applicable Award Agreement evidencing such Award) as of a date, before or after such Change in Control, specified by the Committee, in which event the Committee shall thereupon cancel such Award and the Company shall pay to each such Holder an amount of cash per share equal to the excess, if any, of the per share price offered to stockholders of the Company in connection with such Change in Control over the exercise prices under such Award for such shares;
  - (3) with respect to all or selected Holders, have some or all of their then outstanding Awards (whether vested or unvested) assumed or have a new award of a similar nature substituted for some or all of their then outstanding Awards under the Plan (whether vested or unvested) by an entity which is a party to the transaction resulting in such Change in Control and which is then employing such Holder or which is affiliated or associated with such Holder in the same or a substantially similar manner as the Company prior to the Change in Control, or a parent or substidiary of such entity, provided that (A) such assumption or substitution is on a basis where the excess of the aggregate fair market value of the Stock subject to the Award immediately after the assumption or substitution over the aggregate exercise price of such Stock is equal to the excess of the aggregate fair market value of all Stock subject to the Award immediately before such assumption or substitution over the aggregate exercise price of such Stock; and (B) the assumed rights under such existing Award or the substituted rights under such new Award, as the case may be, will have the same terms and conditions as the rights under the existing Award assumed or substituted for, as the case may be;
  - (4) provide that the number and class or series of Stock covered by an Award (whether vested or unvested) theretofore granted shall be adjusted so that such Award when

exercised shall thereafter cover the number and class or series of Stock or other securities or property (including, without limitation, cash) to which the Holder would have been entitled pursuant to the terms of the agreement or plan relating to such Change in Control if, immediately prior to such Change in Control, the Holder had been the holder of record of the number of shares of Stock then covered by such Award; or

(5) make such adjustments to Awards then outstanding as the Committee deems appropriate to reflect such Change in Control (provided, however, that the Committee may determine in its sole and absolute discretion that no such adjustment is necessary to reflect such Change in Control).

Any adjustment effected by the Committee under Section 4.5 shall be designed to provide the Holder with the intrinsic value of his or her Award, as determined prior to the Change in Control, or, if applicable, equalize the Fair Market Value of the Award before and after the Change in Control.

In effecting one or more of the alternatives set out in paragraphs (3), (4) or (5) immediately above, and except as otherwise may be provided in an Award Agreement, the Committee, in its sole and absolute discretion and without the consent or approval of any Holder, may accelerate the time at which some or all Awards then outstanding may be exercised.

- (d) In the event of changes in the outstanding Stock by reason of recapitalizations, reorganizations, mergers, consolidations, combinations, exchanges or other relevant changes in capitalization occurring after the date of the grant of any Award and not otherwise provided for by this Section 4.5, any outstanding Award and any Award Agreement evidencing such Award shall be subject to adjustment by the Committee in its sole and absolute discretion as to the number and price of Stock or other consideration subject to such Award. In the event of any such change in the outstanding Stock, the aggregate number of shares of Stock available under the Plan may be appropriately adjusted by the Committee, whose determination shall be conclusive.
- (e) After a merger of one or more corporations into the Company or after a consolidation of the Company and one or more corporations in which the Company shall be the surviving corporation, each Holder shall be entitled to have his or her Restricted Stock appropriately adjusted based on the manner in which the shares of Stock were adjusted under the terms of the agreement of merger or consolidation.
- (f) The issuance by the Company of stock of any class or series, or securities convertible into, or exchangeable for, stock of any class or series, for cash or property, or for labor or services either upon direct sale or upon the exercise of rights or warrants to subscribe for them, or upon conversion or exchange of stock or obligations of the Company convertible into, or exchangeable for, stock or other securities, shall not affect, and no adjustment by reason of such issuance shall be made with respect to, the number, class or series, or price of shares of Stock then subject to outstanding Awards.
- 4.6 **Election Under Section 83(b) of the Code.** No Holder shall exercise the election permitted under section 83(b) of the Code with respect to any Award without the prior written approval of the General Counsel or the Chief Financial Officer of the Company. Any Holder who makes an election under section 83(b) of the Code with respect to any Award without the prior written approval of the General Counsel or the Chief Financial Officer of the Company may, in the discretion of the Committee, forfeit any or all Awards granted to him or her under the Plan.
- 4.7 **Forfeiture for Cause.** Notwithstanding any other provision of the Plan or an Award Agreement, if the Committee finds by a majority vote that a Holder, before or after his or her Separation from Service (a) committed a felony or a crime involving moral turpitude or committed any other act or omission involving fraud, embezzlement or any other act of dishonesty in the course of his or her employment by the Company or an Affiliate which conduct damaged the Company or an Affiliate; (b) substantially and repeatedly failed to perform duties of the office held by the Holder as reasonably directed by the Company or an Affiliate; (c) committed gross negligence or willful misconduct with respect to the Company or an Affiliate; (d) committed a material breach of any employment agreement between the Holder and the Company or an Affiliate that is not cured within ten (10) days after receipt of written notice thereof from the Company or the Affiliate, as applicable; (e) failed, within ten (10) days after receipt by the Holder of written notice thereof from the Company or an Affiliate, to correct, cease or otherwise alter any failure to comply with instructions or other action or omission which the Board reasonably believes does or may materially or adversely affect the Company's or an Affiliate's business or operations; (f) committed misconduct which is of

such a serious or substantial nature that a reasonable likelihood exists that such misconduct will materially injure the reputation of the Company or an Affiliate; (g) harassed or discriminated against the Company's or an Affiliate's employees, customers or vendors in violation of the Company's policies with respect to such matters; (h) misappropriated funds or assets of the Company or an Affiliate for personal use or willfully violated the Company policies or standards of business conduct as determined in good faith by the Board; (i) failed, due to some action or inaction on the part of the Holder, to have immigration status that permits the Holder to maintain full-time employment with the Company or an Affiliate in the United States in compliance with all applicable immigration law; (j) disclosed trade secrets of the Company or an Affiliate, then as of the date the Committee makes its finding, any Awards awarded to the Holder that have not been exercised by the Holder (including all Awards that have not yet vested) will be forfeited to the Company. The findings and decision of the Committee or the Board, if applicable, with respect to such matter, including those regarding the acts of the Holder and the damage done to the Company, will be final for all purposes. No decision of the Committee, however, will affect the finality of the discharge of the individual by the Company or an Affiliate.

- 4.8 **Forfeiture Events.** The Committee may specify in an Award Agreement that the Holder's rights, payments, and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture, or recoupment upon the occurrence of certain specified events, in addition to any otherwise applicable vesting or performance conditions of an Award. Such events may include, but shall not be limited to, Separation from Service for cause, Separation from Service for any other reason, violation of material policies of the Company and its Affiliates, breach of noncompetition, confidentiality, or other restrictive covenants that may apply to the Holder, or other conduct by the Holder that is detrimental to the business or reputation of the Company and its Affiliates.
- 4.9 **Award Agreements.** Each Award shall be embodied in a written Award Agreement that shall be subject to the terms and conditions of the Plan. The Award Agreement shall be signed by an executive officer of the Company, other than the Holder, on behalf of the Company, and may be signed by the Holder to the extent required by the Committee. The Award Agreement may specify the effect of a change in control of the Company on the Award. The Award Agreement may contain any other provisions that the Committee in its discretion shall deem advisable which are not inconsistent with the terms and provisions of the Plan.
- 4.10 **Amendments of Award Agreements.** The terms of any outstanding Award under the Plan may be amended from time to time by the Committee in its discretion in any manner that it deems appropriate and that is consistent with the terms of the Plan or necessary to implement the requirements of the Plan. However, no such amendment shall adversely affect in a material manner any right of a Holder without his or her written consent. Except as specified in Section 4.5(b), the Committee may not directly or indirectly lower the exercise price of a previously granted Option or the grant price of a previously granted SAR.
- 4.11 **Rights as Stockholder**. A Holder shall not have any rights as a stockholder with respect to Stock covered by an Option, a SAR, an RSU, a Performance Unit, or an Other Stock-Based Award payable in Stock until the date, if any, such Stock is issued by the Company; and, except as otherwise provided in Section 4.5, no adjustment for dividends, or otherwise, shall be made if the record date therefor is prior to the date of issuance of such Stock.
  - 4.12 **Issuance of Shares of Stock.** Shares of Stock, when issued, may be represented by a certificate or by book or electronic entry.
- 4.13 **Restrictions on Stock Received.** The Committee may impose such conditions and/or restrictions on any shares of Stock issued pursuant to an Award as it may deem advisable or desirable. These restrictions may include, but shall not be limited to, a requirement that the Holder hold the shares of Stock for a specified period of time.
- 4.14 **Compliance With Section 409A.** Awards shall be designed, granted and administered in such a manner that they are either exempt from the application of, or comply with, the requirements of Section 409A. The Plan and each Award Agreement under the Plan that is intended to comply the requirements of Section 409A shall be construed and interpreted in accordance with such intent. If the Committee determines that an Award, Award Agreement, payment, distribution, deferral election, transaction, or any other action or arrangement contemplated by the provisions of the Plan would, if undertaken, cause a Holder to become subject to additional taxes under Section 409A, then unless the Committee specifically provides otherwise, such Award, Award Agreement, payment, distribution, deferral election, transaction or other action or arrangement shall not be given effect to the extent it causes such result and the related provisions of the Plan and/or Award Agreement will be deemed modified, or, if necessary, suspended in order to comply with the requirements of Section 409A to the extent determined appropriate by the Committee, in each case without the consent of or notice to the Holder. The exercisability of an Option or a SAR shall not be extended to the extent that such extension would subject the Holder to additional taxes under Section 409A.

Although the Company will use its best efforts to avoid the imposition of taxation, interest and penalties under Section 409A, the tax treatment of the benefits provided under the Plan is not warranted or guaranteed. Neither the Company, its Affiliates nor their respective directors, officers, employees or advisers shall be held liable for any taxes, interest, penalties or other monetary amounts owed by a Holder (or any other individual claiming a benefit through Holder) as a result of the Plan.

- 4.15 **Date of Grant.** The date on which an option or SAR is granted shall be the date the Company completes the corporate action constituting an offer of stock for sale to a Holder under the terms and conditions of the Option or SAR; provided that such corporate action shall not be considered complete until the date on which the *maximum* number of shares that can be purchased under the Option and the minimum Option price are fixed or determinable. If the corporate action contemplates an immediate offer of stock for sale to a class of individuals, then the date of the granting of an Option is the time or date of that corporate action, if the offer is to be made immediately. If the corporate action contemplates a particular date on which the offer is to be made, then the date of grant is the contemplated date of the offer.
- 4.16 **Source of Shares Deliverable Under Awards.** Any shares of Stock delivered pursuant to an Award may consist, in whole or in part, of authorized and unissued shares of Stock or of treasury shares of Stock.

### ARTICLE V

## **OPTIONS**

- 5.1 **Authority to Grant Options.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Options under the Plan to eligible persons in such number and upon such terms as the Committee shall determine; provided that ISOs may be granted only to eligible Employees of the Company or of any Parent Corporation or Subsidiary Corporation (as permitted by section 422 of the Code and the regulations thereunder).
  - 5.2 **Type of Options Available.** Options granted under the Plan may be NQSOs or ISOs.
- 5.3 **Option Agreement.** Each Option grant under the Plan shall be evidenced by an Award Agreement that shall specify (a) whether the Option is intended to be an ISO or an NQSO, (b) the Option Price, (c) the duration of the Option, (d) the number of shares of Stock to which the Option pertains, (e) the exercise restrictions, if any, applicable to the Option, and (f) such other provisions as the Committee shall determine that are not inconsistent with the terms and provisions of the Plan. Notwithstanding the designation of an Option as an ISO in the applicable Award Agreement for such Option, to the extent the limitations of Section 5.11 of the Plan are exceeded with respect to the Option, the portion of the Option in excess of the limitation shall be treated as a NQSO. An Option granted under the Plan may not be granted with any Dividend Equivalents rights.
- 5.4 **Option Price.** The price at which shares of Stock may be purchased under an Option (the "*Option Price*") shall not be less than one hundred percent (100%) of the Fair Market Value of the shares of Stock on the date the Option is granted; provided, however, if the Option is an ISO granted to a Ten Percent Stockholder, the Option Price must not be less than one hundred ten percent (110%) of the Fair Market Value of the shares of Stock on the date the ISO is granted. Subject to the limitations set forth in the preceding sentences of this Section 5.4, the Committee shall determine the Option Price for each grant of an Option under the Plan.
- 5.5 **Duration of Option.** An Option shall not be exercisable after the earlier of (a) the general term of the Option specified in the applicable Award Agreement (which shall not exceed ten years, or, in the case of a Ten Percent Stockholder, no ISO shall be exercisable later than the fifth (5th) anniversary of the date of its grant), or (b) the period of time specified in the applicable Award Agreement that follows the Holder's Separation from Service.
- 5.6 **Amount Exercisable.** Each Option may be exercised at the time, in the manner and subject to the conditions the Committee specifies in the Award Agreement in its sole discretion.

## 5.7 Exercise of Option.

(a) *General Method of Exercise*. Subject to the terms and provisions of the Plan and the applicable Award Agreement, Options may be exercised in whole or in part from time to time by the delivery of written notice in the manner designated by the Committee. Except in the case of exercise by a third party broker or through a net exercise as provided below, in order for the notice to be effective the notice must be accompanied by payment of the

Option Price by any combination of the following in accordance with the applicable Award Agreement: (v) cash, certified check, bank draft or postal or express money order for an amount equal to the Option Price under the Option; (w) Mature Shares with a Fair Market Value on the date of exercise equal to the Option Price under the Option; (x) an election to make a cashless exercise through a registered broker-dealer; (y) an election to affect a net exercise directing the Company to reduce the number of shares of Stock that will be delivered pursuant to the exercise of the Option; or (z) except as specified below, any other form of payment which is acceptable to the Committee. If Mature Shares are used for payment by the Holder, the aggregate Fair Market Value of the shares of Stock tendered must be equal to or less than the aggregate Option Price of the shares of Stock being purchased upon exercise of the Option, and any difference must be paid by cash, certified check, bank draft or postal or express money order payable to the order of the Company.

If, at the time of receipt by the Company or its delegate of such written notice, (i) the Company has unrestricted surplus in an amount not less than the Option Price of such shares of Stock; (ii) all accrued cumulative preferential dividends and other current preferential dividends on all outstanding shares of preferred stock of the Company have been fully paid; (iii) the acquisition by the Company of its own shares of Stock for the purpose of enabling such Holder to exercise such Option is otherwise permitted by applicable law, does not require any vote or consent of any stockholder of the Company and does not violate the terms of any agreement to which the Company is a party or by which it is bound, and (iv) there shall have been adopted, and there shall be in full force and effect, a resolution of the Board authorizing the acquisition by the Company of its own shares of stock for such purpose, then such Holder may deliver to the Company, in payment of the Option Price of the shares of Stock with respect to which such Option is exercised, (x) certificates registered in the name of such Holder that represent a number of shares of stock legally and beneficially owned by such Holder and having a Fair Market Value on the date of receipt by the Company or its delegate of such written notice that is not greater than the Option Price of the shares of Stock with respect to which such Option is to be exercised, such certificates to be accompanied by stock powers duly endorsed in blank by the record holder of the shares of Stock represented by such certificates, with the signature of such record holder guaranteed by a national banking association, and (y) if the Option Price of the shares of Stock with respect to which such Option is to be exercised exceeds such Fair Market Value, a cashier's check drawn on a national banking association and payable to the order of the Company, in an amount, in United States dollars, equal to the amount of such excess. Notwithstanding the provisions of the immediately preceding sentence, the Committee, in its sole discretion, may refuse to accept shares of Stock in payment of the Option Price of the shares of Stock with respect to which such Option is to be exercised and, in that event, any certificates representing shares of Stock that were received by the Company or its delegate with such written notice shall be returned to such Holder, together with notice by the Company or its delegate to such Holder of the refusal of the Committee to accept such shares of Stock. If, at the expiration of seven (7) business days after the delivery to such Holder of such written notice from the Company or its delegate, such Holder shall not have delivered to the Company or its delegate a cashier's check drawn on a national banking association and payable to the order of the Company in an amount, in United States dollars, equal to the Option Price of the shares of Stock with respect to which such Option is to be exercised, such written notice from the Holder to the Company or its delegate shall be ineffective to exercise such Option.

- (b) *Issuance of Shares*. Subject to Section 4.3 and Section 5.7(c), as promptly as practicable after receipt of written notification and payment, in the form required by Section 5.7(a), of an amount of money necessary to satisfy any withholding tax liability that *may* result from the exercise of such Option, the Company shall deliver to the Holder certificates for the number of shares with respect to which the Option has been exercised, issued in the Holder's name. Delivery of the shares shall be deemed effected for all purposes when a stock transfer agent of the Company shall have deposited the certificates in the United States mail, addressed to the Holder, at the address specified by the Holder or shall have transferred to the account designated by the Holder to which the shares of Stock represented by book or electronic entry are to be delivered.
- (c) Cashless Exercise and Net Exercise. The Committee may permit a Holder to elect to pay the Option Price and any applicable tax withholding resulting from such exercise by (i) authorizing a third-party broker to sell all or a portion of the shares of Stock acquired upon exercise of the Option and remit to the Company a sufficient portion of the sale proceeds to pay the Option Price and any applicable federal, state, local and foreign tax withholding resulting from such exercise; or (ii) directing the Company to reduce the number of shares of Stock that will be delivered pursuant to the exercise of the Option.
- (d) *Limitations on Exercise Alternatives*. Except in the event the Option Price is paid pursuant to Section 5.7(c), the Committee shall not permit a Holder to pay such Holder's Option Price upon the exercise of an Option by using shares of Stock other than Mature Shares. An Option may not be exercised for a fraction of a share of Stock.

- 5.8 **Transferability—Incentive Stock Options.** Notwithstanding anything in the Plan or an Award Agreement to the contrary, no ISO granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution, and all ISOs granted to an Employee under this Article V shall be exercisable during his or her lifetime only by such Employee.
- 5.9 **Notification of Disqualifying Disposition.** If any Employee shall make any disposition of shares of Stock issued pursuant to the exercise of an ISO under the circumstances described in section 421(b) of the Code (relating to certain disqualifying dispositions), such Employee shall notify the Company of such disposition within ten (10) days thereof.
- 5.10 **No Rights as Stockholder.** A Holder of an Option shall not have any rights as a stockholder with respect to Stock covered by an Option until the date a stock certificate for such Stock is issued by the Company; and, except as otherwise provided in Section 4.5, no adjustment for dividends, or otherwise, shall be made if the record date therefor is prior to the date of issuance of such certificate.
- 5.11 **\$100,000 Limitation on ISOs.** To the extent that the aggregate Fair Market Value of shares of Stock with respect to which ISOs first become exercisable by a Holder in any calendar year exceeds \$100,000, taking into account both shares of Stock subject to ISOs under the Plan and Stock subject to ISOs under all other plans of the Company, such Options shall be treated as NQSOs. For this purpose, the "Fair Market Value" of the shares of Stock subject to Options shall be determined as of the date the Options were awarded. In reducing the number of Options treated as ISOs to meet the \$100,000 limit, the most recently granted Options shall be reduced first. To the extent a reduction of simultaneously granted Options is necessary to meet the \$100,000 limit, the Committee may, in the manner and to the extent permitted by law, designate which shares of Stock are to be treated as shares acquired pursuant to the exercise of an ISO.
- 5.12 **Separation from Service.** Each Award Agreement shall set forth the extent to which the Holder of an Option shall have the right to exercise the Option following the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee, need not be uniform among all Options issued pursuant to the Award Agreement or the Plan, and may reflect distinctions based on the reasons for termination.

### ARTICLE VI

## STOCK APPRECIATION RIGHTS

- 6.1 **Authority to Grant SAR Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant SARs under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. Subject to the terms and conditions of the Plan, the Committee shall have complete discretion in determining the number of SARs granted to each Holder and, consistent with the provisions of the Plan, in determining the terms and conditions pertaining to such SARs.
- 6.2 **Type of Stock Appreciation Rights Available.** The Committee may grant Freestanding SARs, Tandem SARs, or any combination of these forms of SARs.
- 6.3 **General Terms.** Subject to the terms and conditions of the Plan, a SAR granted under the Plan shall confer on the recipient a right to receive, upon exercise thereof, an amount equal to the excess of (a) the Fair Market Value of one share of the Stock on the date of exercise, over (b) the grant price of the SAR, which shall not be less than one hundred percent (100%) of the Fair Market Value of one share of the Stock on the date of grant of the SAR. The grant price of Tandem SARs shall be equal to the Option Price of the related Option. A SAR granted under the Plan may not be granted with any Dividend Equivalents rights.
- 6.4 **SAR Agreement.** Each Award of SARs granted under the Plan shall be evidenced by an Award Agreement that shall specify (a) the grant price of the SAR, (b) the term of the SAR, (c) the vesting and termination provisions of the SAR, and (d) such other provisions as the Committee shall determine that are not inconsistent with the terms and provisions of the Plan. The Committee may impose such additional conditions or restrictions on the exercise of any SAR as it may deem appropriate.
- 6.5 **Term of SAR.** The term of a SAR granted under the Plan shall be determined by the Committee, in its sole discretion; provided that no SAR shall be exercisable on or after the tenth (10<sup>th</sup>) anniversary date of its grant. Notwithstanding any other provision of this Plan to the contrary, with respect to a Tandem SAR granted in connection with an ISO: (a) the Tandem SAR will expire no later than the expiration of the underlying ISO; (b) the value of the payout with respect to the Tandem SAR

may be for no more than one hundred percent (100%) of the excess of the Fair Market Value of the shares of Stock subject to the underlying ISO at the time the Tandem SAR is exercised over the Option Price of the underlying ISO; and (c) the Tandem SAR may be exercised only when the Fair Market Value of the shares of Stock subject to the ISO exceeds the Option Price of the ISO.

- 6.6 **Exercise of Freestanding SARs.** Subject to the terms and provisions of the Plan and the applicable Award Agreement, Freestanding SARs may be exercised in whole or in part from time to time by the delivery of written notice in the manner designated by the Committee stating (a) that the Holder wishes to exercise such SAR on the date such notice is so delivered, (b) the number of shares of Stock with respect to which the SAR is to be exercised, and (c) the address to which the payment due under such SAR should be delivered. In accordance with applicable law, a Freestanding SAR may be exercised upon whatever additional terms and conditions the Committee, in its sole discretion, imposes.
- 6.7 **Exercise of Tandem SARs.** Subject to the terms and provisions of the Plan and the applicable Award Agreement, Tandem SARs may be exercised for all or part of the shares of Stock subject to the related Option upon the surrender of the right to exercise the equivalent portion of the related Option and by the delivery of written notice in the manner designated by the Committee stating (a) that the Holder wishes to exercise such SAR on the date such notice is so delivered, (b) the number of shares of Stock with respect to which the SAR is to be exercised, and (c) the address to which the payment due under such SAR should be delivered. A Tandem SAR may be exercised only with respect to the shares of Stock for which its related Option is then exercisable. In accordance with applicable law, a Tandem SAR may be exercised upon whatever additional terms and conditions the Committee, in its sole discretion, imposes.
- 6.8 **Payment of SAR Amount.** Upon the exercise of a SAR, a Holder shall be entitled to receive payment from the Company in an amount determined by multiplying the excess of the Fair Market Value of a share of Stock on the date of exercise over the grant price of the SAR by the number of shares of Stock with respect to which the SAR is exercised. At the discretion of the Committee, the payment upon SAR exercise may be in cash, in Stock of equivalent value, in some combination thereof or in any other manner approved by the Committee in its sole discretion. The Committee's determination regarding the form of SAR payout shall be set forth in the Award Agreement pertaining to the grant of the SAR.
- 6.9 **Separation from Service.** Each Award Agreement shall set forth the extent to which the Holder of a SAR shall have the right to exercise the SAR following the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee, may be included in the Award Agreement entered into with the Holder, need not be uniform among all SARs issued pursuant to the Plan, and may reflect distinctions based on the reasons for termination or severance.
- 6.10 **Nontransferability of SARs.** Except as otherwise provided in a Holder's Award Agreement, no SAR granted under the Plan may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, except as otherwise provided in a Holder's Award Agreement, all SARs granted to a Holder under the Plan shall be exercisable during his or her lifetime only by the Holder, and after that time, by the Holder's heirs or estate. Any attempted assignment of a SAR in violation of this Section 6.10 shall be null and void.
  - 6.11 **No Rights as Stockholder.** A grantee of a SAR award, as such, shall have no rights as a stockholder.
- 6.12 **Restrictions on Stock Received.** The Committee may impose such conditions and/or restrictions on any shares of Stock received upon exercise of a SAR granted pursuant to the Plan as it may deem advisable or desirable. These restrictions may include, but shall not be limited to, a requirement that the Holder hold the shares of Stock received upon exercise of a SAR for a specified period of time.

## **ARTICLE VII**

### RESTRICTED STOCK AWARDS

7.1 **Restricted Stock Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may make Awards of Restricted Stock under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any Restricted Stock Award shall be determined by the Committee in its sole discretion. If the Committee imposes vesting, forfeiture or transferability restrictions on a Holder's rights with respect to Restricted Stock, the Committee may issue such instructions to the Company's share transfer agent in connection therewith as it deems appropriate. The Committee may also cause the certificate for shares of Stock issued pursuant to a Restricted Stock Award to be imprinted with any legend which counsel for the

Company considers advisable with respect to the restrictions or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law.

- 7.2 **Restricted Stock Award Agreement.** Each Restricted Stock Award shall be evidenced by an Award Agreement that contains any vesting, forfeiture and transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
- 7.3 **Holder's Rights as Stockholder.** Subject to the terms and conditions of the Plan, each recipient of a Restricted Stock Award shall have all the rights of a stockholder with respect to the shares of Restricted Stock included in the Restricted Stock Award during the Period of Restriction established for the Restricted Stock Award. Dividends paid with respect to Restricted Stock in cash or property other than shares of Stock or rights to acquire shares of Stock shall be paid to the recipient of the Restricted Stock Award currently. Dividends paid in shares of Stock or rights to acquire shares of Stock shall be added to and become a part of the Restricted Stock. During the Period of Restriction, certificates representing the Restricted Stock shall be registered in the Holder's name and bear a restrictive legend to the effect that ownership of such Restricted Stock, and the enjoyment of all rights appurtenant thereto, are subject to the restrictions, terms, and conditions provided in the Plan and the applicable Award Agreement. Such certificates shall be deposited by the recipient with the Secretary of the Company or such other officer or agent of the Company as may be designated by the Committee, together with all stock powers or other instruments of assignment, each endorsed in blank, which will permit transfer to the Company of all or any portion of the Restricted Stock which shall be forfeited in accordance with the Plan and the applicable Award Agreement.

## ARTICLE VIII

## RESTRICTED STOCK UNIT AWARDS

- 8.1 **Authority to Grant RSU Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant RSU Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any RSU Award shall be determined by the Committee in its sole discretion. The Committee shall maintain a bookkeeping ledger account, which reflects the number of RSUs credited under the Plan for the benefit of a Holder.
- 8.2 **RSU** Award. An RSU Award shall be similar in nature to a Restricted Stock Award except that no shares of Stock are actually transferred to the Holder until a later date specified in the applicable Award Agreement. Each RSU shall have a value equal to the Fair Market Value of a share of Stock.
- 8.3 **RSU Award Agreement.** Each RSU Award shall be evidenced by an Award Agreement that contains any Substantial Risk of Forfeiture, transferability and forfeiture restrictions, form and time of payment provisions and other provisions not inconsistent with the Plan as the Committee may specify.
- 8.4 **Dividend Equivalents.** An Award Agreement for an RSU Award may specify that the Holder shall be entitled to the payment of Dividend Equivalents under the Award.
- 8.5 **Form of Payment Under RSU Award.** Payment under an RSU Award shall be made in cash, shares of Stock or any combination thereof, as specified in the applicable Award Agreement.
- 8.6 **Time of Payment Under RSU Award.** A Holder's payment under an RSU Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the Fiscal Year in which the RSU Award payment is no longer subject to a Substantial Risk of Forfeiture, or (b) as otherwise designated by the Committee.
- 8.7 **Holder's Rights as Stockholder.** Each recipient of an RSU Award shall have no rights of a stockholder with respect to the Holder's RSUs. A Holder shall have no voting rights with respect to any RSU Awards.

### ARTICLE IX

## PERFORMANCE STOCK AWARDS AND PERFORMANCE UNIT AWARDS

- 9.1 Authority to Grant Performance Stock Awards and Performance Unit Awards. Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Performance Stock Awards and Performance Unit Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine. The amount of, the vesting, forfeiture and the transferability restrictions applicable to any Performance Stock Award or Performance Unit Award shall be based upon the attainment of such Performance Goals as the Committee may determine; provided, however, that (a) the performance period for any Performance Stock Award or Performance Unit Award shall not be less than one (1) year, and (b) the Performance Goals must be established in writing by the Committee not later than ninety (90) days after the beginning of the performance period (but in no event after the outcome is substantially certain). If the Committee imposes vesting or transferability restrictions on a Holder's rights with respect to Performance Stock Award or Performance Unit Awards, the Committee may issue such instructions to the Company's share transfer agent in connection therewith as it deems appropriate. The Committee may also cause the certificate for shares of Stock issued pursuant to a Performance Stock Award or Performance Unit Award to be imprinted with any legend which counsel for the Company considers advisable with respect to the restrictions or, should the shares of Stock be represented by book or electronic entry rather than a certificate, the Company may take such steps to restrict transfer of the shares of Stock as counsel for the Company considers necessary or advisable to comply with applicable law
- 9.2 **Performance Goals.** The Performance Goals upon which the payment or vesting of an Award may be based on one or more of the following business criteria that apply to the Holder, one or more business units or subsidiaries of the Company, or the Company as a whole: earnings, earnings before interest and taxes, earnings before interest, taxes, depreciation, and amortization, earnings before any one of, or combination of two or more of, interest, taxes, depreciation, amortization and/or any other financial adjustment to earnings set forth in the Company's audited financial statements that is allowed under generally accepted accounting principles, adjusted earnings before interest, taxes, depreciation and amortization, net earnings, earnings per share, earnings per share growth, economic value added, economic value, operating profits, net operating profit, net profits, profit return, gross margin, profit margins, profit before tax, operating margin, cash return on capitalization, operating expense, operating expense as a percentage of revenue, revenue, increase in revenue, revenue ratios (including per employee or per customer), net revenue, billings, net income, stock price, market share, return on equity, return on assets, return on capital, return on capital compared to cost of capital, return on capital employed, return on invested capital, debt to capital ratio, total stockholder return, stockholder return, stockholder value, growth in stockholder value relative to a pre-determined index, financial return ratio, operating income, cash flow, net cash flow, cash flow from operations, free cash flow, cash value added performance, cost reductions, cost ratios (per employee or per customer), proceeds from dispositions, project completion time and budget goals, net cash flow before financing activities, customer growth, total market value, or people value added. The Committee may select one criterion or multiple criteria for measuring performance. Goals may be based on subjective performance criteria, provided the subjective criteria are bona fide and relate to the performance of the Holder or of the group that includes the Holder. Goals may also be based on performance relative to a peer group of companies. Unless otherwise stated, such a Performance Goal need not be based upon an increase or positive result under a particular business criterion and could include, for example, maintaining the status quo or limiting economic losses (measured, in each case, by reference to specific business criteria). Prior to the payment of any compensation based on the achievement of Performance Goals, the Committee must certify in writing that applicable Performance Goals and any of the material terms thereof were, in fact, satisfied. Subject to the foregoing provisions, the terms, conditions and limitations applicable to any Performance Stock or Performance Unit Awards made pursuant to the Plan shall be determined by the Committee.
- 9.3 **Written Agreement.** Each Performance Stock Award or Performance Unit Award shall be evidenced by an Award Agreement that contains any vesting, transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
- 9.4 **Form of Payment Under Performance Unit Award.** Payment under a Performance Unit Award shall be made in cash, shares of Stock or any combination thereof, as specified in the applicable Award Agreement.
- 9.5 **Time of Payment Under Performance Unit Award.** A Holder's payment under a Performance Unit Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Performance Unit Award payment is no longer subject to a Substantial Risk of Forfeiture, or (b) as otherwise designated by the Committee.

- 9.6 **Holder's Rights as Stockholder With Respect to a Performance Stock Award.** Subject to the terms and conditions of the Plan, each Holder of a Performance Stock Award shall have all the rights of a stockholder with respect to the shares of Stock issued to the Holder pursuant to the Award during any period in which such issued shares of Stock are subject to forfeiture and restrictions on transfer, including without limitation, the right to vote such shares of Stock.
- 9.7 **Dividend Equivalents.** An Award Agreement for a Performance Unit Award may specify that the Holder shall be entitled to the payment of Dividend Equivalents under the Award.

### ARTICLE X

### DIRECTOR AWARDS

All Awards to Directors shall be determined by the Board.

## ARTICLE XI

## ANNUAL CASH INCENTIVE AWARDS

- 11.1 **Authority to Grant Annual Cash Incentive Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Annual Cash Incentive Awards under the Plan upon such terms as the Committee shall determine. Subject to the following provisions in this Article XI, the amount of any Annual Cash Incentive Awards shall be based on the attainment of such Performance Goals as the Committee may determine and the term, conditions and limitations applicable to any Annual Cash Incentive Awards made pursuant to the Plan shall be determined by the Committee.
- 11.2 **Written Agreement.** Each Annual Cash Incentive Award shall be evidenced by an Award Agreement that contains any vesting, transferability restrictions and other provisions not inconsistent with the Plan as the Committee may specify.
  - 11.3 Form of Payment Under Annual Cash Incentive Award. Payment under an Annual Cash Incentive Award shall be made in cash.
- 11.4 **Time of Payment Under Annual Cash Incentive Award.** A Holder's payment under an Annual Cash Incentive Award shall be made at such time as is specified in the applicable Award Agreement. The Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Annual Cash Incentive Award payment is no longer subject to a Substantial Risk of Forfeiture, or (b) as otherwise designated by the Committee.

## ARTICLE XII

## OTHER STOCK-BASED AWARDS

- 12.1 **Authority to Grant Other Stock-Based Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant other types of equity-based or equity-related Awards not otherwise described by the terms and provisions of the Plan (including the grant or offer for sale of unrestricted shares of Stock) under the Plan to eligible persons in such number and upon such terms as the Committee shall determine. Such Awards may involve the transfer of actual shares of Stock to Holders, or payment in cash or otherwise of amounts based on the value of shares of Stock and may include, without limitation, Awards designed to comply with or take advantage of the applicable local laws of jurisdictions other than the United States.
- 12.2 **Value of Other Stock-Based Award.** Each Other Stock-Based Award shall be expressed in terms of shares of Stock or units based on shares of Stock, as determined by the Committee.
- 12.3 **Payment of Other Stock-Based Award.** Payment, if any, with respect to an Other Stock-Based Award shall be made in accordance with the terms of the Award, in cash, shares of Stock or any combination thereof, as the Committee determines.

- 12.4 **Separation from Service.** The Committee shall determine the extent to which a Holder's rights with respect to Other Stock-Based Awards shall be affected by the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee and need not be uniform among all Other Stock-Based Awards issued pursuant to the Plan.
- 12.5 **Time of Payment of Other Stock-Based Award.** A Holder's payment under an Other Stock-Based Award shall be made at such time as is specified in the applicable Award Agreement. If a payment under the Award Agreement is subject to Section 409A, the Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Other Stock-Based Award payment is no longer subject to a Substantial Risk of Forfeiture, or (b) as otherwise designated by the Committee.

### ARTICLE XIII

### **CASH-BASED AWARDS**

- 13.1 **Authority to Grant Cash-Based Awards.** Subject to the terms and provisions of the Plan, the Committee, at any time, and from time to time, may grant Cash-Based Awards under the Plan to eligible persons in such amounts and upon such terms as the Committee shall determine.
  - 13.2 Value of Cash-Based Award. Each Cash-Based Award shall specify a payment amount or payment range as determined by the Committee.
- 13.3 **Payment of Cash-Based Award.** Payment, if any, with respect to a Cash-Based Award shall be made in accordance with the terms of the Award, in cash.
- 13.4 **Time of Payment of Cash-Based Award.** Payment under a Cash-Based Award shall be made at such time as is specified in the applicable Award Agreement. If a payment under the Award Agreement is subject to Section 409A, the Award Agreement shall specify that the payment will be made (a) by a date that is no later than the date that is two and one-half (2 1/2) months after the end of the calendar year in which the Cash-Based Award payment is no longer subject to a Substantial Risk of Forfeiture, or (b) as otherwise designated by the Committee.
- 13.5 **Separation from Service.** The Committee shall determine the extent to which a Holder's rights with respect to Cash-Based Awards shall be affected by the Holder's Separation from Service. Such provisions shall be determined in the sole discretion of the Committee and need not be uniform among all Cash-Based Awards issued pursuant to the Plan.

### ARTICLE XIV

## SUBSTITUTION AWARDS

Awards may be granted under the Plan from time to time in substitution for stock options and other awards held by employees of other entities who are about to become Employees, or whose employer is about to become an Affiliate as the result of a merger or consolidation of the Company with another corporation, or the acquisition by the Company of substantially all the assets of another corporation, or the acquisition by the Company of at least fifty percent (50%) of the issued and outstanding stock of another corporation as the result of which such other corporation will become a subsidiary of the Company. The terms and conditions of the substitute Awards so granted may vary from the terms and conditions set forth in the Plan to such extent as the Board at the time of grant may deem appropriate to conform, in whole or in part, to the provisions of the awards in substitution for which they are granted. If shares of Stock are issued under the Plan with respect to an Award granted under this Article such shares of Stock will not count against the aggregate number of shares of Stock with respect to which Awards may be granted under the Plan.

## ARTICLE XV

## ADMINISTRATION

15.1 **Awards.** The Plan shall be administered by the Committee or, in the absence of the Committee or in the case of awards issued to Directors, the Plan shall be administered by the Board. The members of the Committee (that is not itself the Board) shall serve at the discretion of the Board. The Committee shall have full and exclusive power and authority to administer the Plan and to take all actions that the Plan expressly contemplates or are necessary or appropriate in connection with the administration of the Plan with respect to Awards granted under the Plan.

15.2 **Authority of the Committee.** The Committee shall have full and exclusive power to interpret and apply the terms and provisions of the Plan and Awards made under the Plan, and to adopt such rules, regulations and guidelines for implementing the Plan as the Committee may deem necessary or proper, all of which powers shall be exercised in the best interests of the Company and in keeping with the objectives of the Plan. A majority of the members of the Committee shall constitute a quorum for the transaction of business relating to the Plan or Awards made under the Plan, and the vote of a majority of those members present at any meeting shall decide any question brought before that meeting. Any decision or determination reduced to writing and signed by a majority of the members shall be as effective as if it had been made by a majority vote at a meeting properly called and held. All questions of interpretation and application of the Plan, or as to Awards granted under the Plan, shall be subject to the determination, which shall be final and binding, of a majority of the whole Committee. No member of the Committee shall be liable for any act or omission of any other member of the Committee or for any act or omission on his or her own part, including but not limited to the exercise of any power or discretion given to him or her under the Plan, except those resulting from his or her own willful misconduct. In carrying out its authority under the Plan, the Committee shall have full and final authority and discretion, including but not limited to the following rights, powers and authorities to (a) determine the persons to whom and the time or times at which Awards will be made; (b) determine the number and exercise price of shares of Stock covered in each Award subject to the terms and provisions of the Plan; (c) determine the terms, provisions and conditions of each Award, which need not be identical and need not match the default terms set forth in the Plan; (d) accelerate the time at which any outstanding Award will vest; (e) prescribe, amend and rescind rules and regulations relating to administration of the Plan; and (f) make all other determinations and take all other actions deemed necessary, appropriate or advisable for the proper administration of the Plan.

The Committee may make an Award to an individual who the Company expects to become an Employee of the Company or any of its Affiliates within six (6) months after the date of grant of the Award, with the Award being subject to and conditioned on the individual actually becoming an Employee within that time period and subject to other terms and conditions as the Committee may establish.

The Committee may correct any defect or supply any omission or reconcile any inconsistency in the Plan or in any Award to a Holder in the manner and to the extent the Committee deems necessary or desirable to further the Plan's objectives. Further, the Committee shall make all other determinations that may be necessary or advisable for the administration of the Plan. As permitted by law and the terms and provisions of the Plan, the Committee may delegate to one or more of its members or to one or more officers of the Company, and/or its Affiliates or to one or more agents or advisors such administrative duties or powers as it may deem advisable, and the Committee or any person to whom it has delegated duties or powers as aforesaid may employ one or more persons to render advice with respect to any responsibility the Committee or such person may have under the Plan. The Committee may, by resolution, authorize one or more officers of the Company to do one or more of the following on the same basis as can the Committee: (a) designate Employees to be recipients of Awards; (b) designate Third Party Service Providers to be recipients of Awards; and (c) determine the size of any such Awards; provided, however, (i) the Committee shall not delegate such responsibilities to any such officer for Awards granted to an Employee that is considered an Insider; (ii) the resolution providing such authorization sets forth the total number of Awards such officer(s) may grant; and (iii) the officer(s) shall report periodically to the Committee regarding the nature and scope of the Awards granted pursuant to the authority delegated. The Committee may employ attorneys, consultants, accountants, agents, and other persons, any of whom may be an Employee, and the Committee, the Committee may employ attorneys, consultants, accountants, opinions, or valuations of any such persons.

15.3 **Decisions Binding.** All determinations and decisions made by the Committee or the Board, as the case may be, pursuant to the provisions of the Plan and all related orders and resolutions of the Committee or the Board, as the case may be, shall be final, conclusive and binding on all persons, including the Company, its Affiliates, its stockholders, Holders and the estates and beneficiaries of Holders.

15.4 **No Liability.** Under no circumstances shall the Company, its Affiliates, the Board or the Committee incur liability for any indirect, incidental, consequential or special damages (including lost profits) of any form incurred by any person, whether or not foreseeable and regardless of the form of the act in which such a claim may be brought, with respect to the Plan or the Company's, its Affiliates', the Committee's or the Board's roles in connection with the Plan.

### ARTICLE XVI

### AMENDMENT OR TERMINATION OF PLAN

- 16.1 **Amendment, Modification, Suspension, and Termination.** Subject to Section 16.2, the Board may, at any time and from time to time, alter, amend, modify, suspend, or terminate the Plan and the Committee may, at any time and from time to time, alter, amend, modify, suspend, or terminate any Award Agreement in whole or in part; provided, however, that, without the prior approval of the Company's stockholders and except as provided in Section 4.5, the Committee shall not directly or indirectly lower the Option Price of a previously granted Option or the grant price of a previously granted SAR, cancel a previously granted Option or previously granted SAR for a payment of cash or other property if the aggregate fair market value of such Award is less than the aggregate Option Price of such Award in the case of an Option or the aggregate grant price of such Award in the case of a SAR, and no amendment of the Plan shall be made without stockholder approval if stockholder approval is required by applicable law or stock exchange rules.
- 16.2 **Awards Previously Granted.** Notwithstanding any other provision of the Plan to the contrary, no termination, amendment, suspension, or modification of the Plan or an Award Agreement shall adversely affect in any material way any Award previously granted under the Plan, without the written consent of the Holder holding such Award.

## ARTICLE XVII

### **MISCELLANEOUS**

- 17.1 **Unfunded Plan/No Establishment of a Trust Fund.** Holders shall have no right, title, or interest whatsoever in or to any investments that the Company or any of its Affiliates may make to aid in meeting obligations under the Plan. Nothing contained in the Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any Holder, beneficiary, legal representative, or any other person. To the extent that any person acquires a right to receive payments from the Company under the Plan, such right shall be no greater than the right of an unsecured general creditor of the Company. All payments to be made hereunder shall be paid from the general funds of the Company and no special or separate fund shall be established and no segregation of assets shall be made to assure payment of such amounts, except as expressly set forth in the Plan. No property shall be set aside nor shall a trust fund of any kind be established to secure the rights of any Holder under the Plan. The Plan is not intended to be subject to the Employee Retirement Income Security Act of 1974, as amended.
- 17.2 **No Employment Obligation.** The granting of any Award shall not constitute an employment or service contract, express or implied, and shall not impose upon the Company or any Affiliate any obligation to employ or continue to employ, or to utilize or continue to utilize the services of, any Holder. The right of the Company or any Affiliate to terminate the employment of, or the provision of services by, any person shall not be diminished or affected by reason of the fact that an Award has been granted to him, and nothing in the Plan or an Award Agreement shall interfere with or limit in any way the right of the Company or its Affiliates to terminate any Holder's employment or service relationship at any time or for any reason not prohibited by law.
- 17.3 **Tax Withholding.** The Company or any Affiliate shall be entitled to deduct from other compensation payable to each Holder any sums required by federal, state, local or foreign tax law to be withheld with respect to the vesting or exercise of an Award or lapse of restrictions on an Award. In the alternative, the Company may require the Holder (or other person validly exercising the Award) to pay such sums for taxes directly to the Company or any Affiliate in cash or by check within one day after the date of vesting, exercise or lapse of restrictions. In the discretion of the Committee, and with the consent of the Holder, the Company may reduce the number of shares of Stock issued to the Holder upon such Holder's exercise of an Award or the vesting of an Award to satisfy the tax withholding obligations of the Company or an Affiliate.

The Committee may, in its discretion, permit a Holder to satisfy any tax-withholding obligation arising upon the vesting of or payment under an Award by delivering to the Holder a reduced number of shares of Stock in the manner specified herein. If permitted by the Committee and acceptable to the Holder, at the time of vesting of shares under the Award, the Company shall (a) calculate the amount of the Company's or an Affiliate's tax withholding obligation on the assumption that all such shares of Stock vested under the Award are made available for delivery; (b) reduce the number of such shares of Stock made available for delivery so that the Fair Market Value of the shares of Stock withheld on the vesting date approximates the Company's or an Affiliate's tax withholding obligation; and (c) in lieu of the withheld shares of Stock, remit cash to the United States Treasury and/or other applicable governmental authorities, on behalf of the Holder, in the amount of the tax withholding obligation. The

Company shall withhold only whole shares of Stock to satisfy its tax-withholding obligation. Where the Fair Market Value of the withheld shares of Stock does not equal the amount of the tax withholding obligation, the Company shall withhold shares of Stock with a Fair Market Value slightly less than the amount of the tax withholding obligation and the Holder must satisfy the remaining minimum withholding obligation in some other manner permitted under this Section 17.3. The withheld shares of Stock not made available for delivery by the Company shall be retained as treasury shares or will be cancelled and the Holder's right, title and interest in such shares of Stock shall terminate.

The Company shall have no obligation upon vesting or exercise of any Award or lapse of restrictions on an Award until the Company or an Affiliate has received payment sufficient to cover the tax withholding obligation with respect to that vesting, exercise or lapse of restrictions. Neither the Company nor any Affiliate shall be obligated to advise a Holder of the existence of the tax or the amount which it will be required to withhold.

- 17.4 **Indemnification of the Committee.** The Company shall indemnify each past, present and future member of the Committee against, and each member of the Committee shall be entitled without further action on his or her part to indemnity from the Company for, all expenses (including attorney's fees, the amount of judgments and the amount of approved settlements made with a view to the curtailment of costs of litigation, other than amounts paid to the Company itself) reasonably incurred by such member in connection with or arising out of any action, suit or proceeding in which such member may be involved by reason of such member being or having been a member of the Committee, whether or not he or she continues to be a member of the Committee at the time of incurring the expenses, including, without limitation, matters as to which such member shall be finally adjudged in any action, suit or proceeding to have been negligent in the performance of such member's duty as a member of the Committee. However, this indemnity shall not include any expenses incurred by any member of the Committee in respect of matters as to which such member shall be finally adjudged in any action, suit or proceeding to have been guilty of willful misconduct in the performance of his or her duty as a member of the Committee. In addition, no right of indemnification under the Plan shall be available to or enforceable by any member of the Committee unless, within sixty (60) days after institution of any action, suit or proceeding, such member shall have offered the Company, in writing, the opportunity to handle and defend same at its own expense. This right of indemnification shall inure to the benefit of the heirs, executors or administrators of each member of the Committee and shall be in addition to all other rights to which a member of the Committee may be entitled as a matter of law, contract or otherwise.
- 17.5 **Gender and Number.** If the context requires, words of one gender when used in the Plan shall include the other and words used in the singular or plural shall include the other.
- 17.6 **Severability.** In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- 17.7 **Headings.** Headings of Articles and Sections are included for convenience of reference only and do not constitute part of the Plan and shall not be used in construing the terms and provisions of the Plan.
- 17.8 **Other Compensation Plans.** The adoption of the Plan shall not affect any other option, incentive or other compensation or benefit plans in effect for the Company or any Affiliate, nor shall the Plan preclude the Company from establishing any other forms of incentive compensation arrangements for Employees, Directors or Third Party Service Providers.
- 17.9 **Retirement and Welfare Plans.** Neither Awards made under the Plan nor shares of Stock or cash paid pursuant to such Awards, may be included as "compensation" for purposes of computing the benefits payable to any person under the Company's or any Affiliate's retirement plans (both qualified and non-qualified) or welfare benefit plans unless such other plan expressly provides that such compensation shall be taken into account in computing a participant's benefit.
- 17.10 **Other Awards.** The grant of an Award shall not confer upon the Holder the right to receive any future or other Awards under the Plan, whether or not Awards may be granted to similarly situated Holders, or the right to receive future Awards upon the same terms or conditions as previously granted.
- 17.11 **Successors.** All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business and/or assets of the Company.

- 17.12 **Law Limitations/Governmental Approvals.** The granting of Awards and the issuance of shares of Stock under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
- 17.13 **Delivery of Title.** The Company shall have no obligation to issue or deliver evidence of title for shares of Stock issued under the Plan prior to (a) obtaining any approvals from governmental agencies that the Company determines are necessary or advisable, and (b) completion of any registration or other qualification of the Stock under any applicable national or foreign law or ruling of any governmental body that the Company determines to be necessary or advisable.
- 17.14 **Inability to Obtain Authority.** The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any shares of Stock hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such shares of Stock as to which such requisite authority shall not have been obtained.
- 17.15 **Investment Representations.** The Committee may require any person receiving Stock pursuant to an Award under the Plan to represent and warrant in writing that the person is acquiring the shares of Stock for investment and without any present intention to sell or distribute such Stock.
- 17.16 **Persons Residing Outside of the United States.** Notwithstanding any provision of the Plan to the contrary, in order to comply with the laws in other countries in which the Company or any of its Affiliates operates or has employees, the Committee, in its sole discretion, shall have the power and authority to (a)determine which Affiliates shall be covered by the Plan; (b) determine which persons employed outside the United States are eligible to participate in the Plan; (c) amend or vary the terms and provisions of the Plan and the terms and conditions of any Award granted to persons who reside outside the United States; (d) establish subplans and modify exercise procedures and other terms and procedures to the extent such actions may be necessary or advisable—any subplans and modifications to Plan terms and procedures established under this Section 17.16 by the Committee shall be attached to the Plan document as Appendices; and (e) take any action, before or after an Award is made, that it deems advisable to obtain or comply with any necessary local government regulatory exemptions or approvals. Notwithstanding the above, the Committee may not take any actions hereunder, and no Awards shall be granted, that would violate the Exchange Act, the Code, any securities law or governing statute or any other applicable law.
- 17.17 **Settlement of Disputes.** Any controversy arising out of or relating to the Plan or an Award Agreement shall be resolved solely and exclusively by the state and federal courts in Houston, Texas.
- 17.18 **No Fractional Shares.** No fractional shares of Stock shall be issued or delivered pursuant to the Plan or any Award. The Committee shall determine whether cash, additional Awards, or other property shall be issued or paid in lieu of fractional shares of Stock or whether such fractional shares or any rights thereto shall be forfeited or otherwise eliminated.
- 17.19 **Governing Law.** The provisions of the Plan and the rights of all persons claiming thereunder shall be construed, administered and governed under the laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of the Plan to the substantive law of another jurisdiction. Unless otherwise provided in the Award Agreement, recipients of an Award under the Plan are deemed to submit to the sole and exclusive jurisdiction and venue of the federal or state courts of the State of Texas to resolve any and all issues that may arise out of or relate to the Plan or any related Award Agreement.

# FIRST AMENDMENT TO THE RIGNET, INC. 2019 OMNIBUS INCENTIVE PLAN

THIS FIRST AMENDMENT is made by RigNet, Inc. (the "Sponsor"),

#### WITNESSETH:

**WHEREAS**, the Sponsor adopted on May 8, 2019, and continues to sponsor and maintain the plan known as the "RigNet, Inc. 2019 Omnibus Incentive Plan" (the "*Plan*"); and

**WHEREAS**, the Board of Directors of the Sponsor (the "*Board*") retained the right in Section 16.1 of the Plan to amend the Plan from time to time; and

WHEREAS, the Board of Directors of the Sponsor approved resolutions on March 11, 2020 to amend the Plan as set forth below;

**NOW, THEREFORE**, the Sponsor agrees that, effective as set forth below, the Plan is amended as set forth below:

- 1. <u>Amendment to Sections 4.1(a), (b) and (c) of the Plan</u>. Sections 4.1(a), (b) and (c) of the Plan are hereby amended and restated in their entirety to read as follows:
  - (a) The aggregate number of shares of Stock with respect to which Awards may be granted under the Plan is 4,975,000 (the "Authorized Shares").
  - (b) The aggregate number of shares of Stock with respect to which ISOs may be granted under the Plan is equal to the Authorized Shares.
- (c) The maximum number of shares of Stock with respect to which ISOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which NQSOs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which SARs may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Stock Awards may be granted to an Employee during a Fiscal Year is equal to the Authorized Shares. The maximum number of shares of Stock with respect to which Performance Unit Awards payable in shares of Stock may be granted to an Employee during a Fiscal Year, determined as of the dates of grants of the Performance Unit Awards, is \$3,000,000. The maximum amount that may be paid under Annual Cash Incentive Award(s) granted to an Employee during a Fiscal Year is \$3,000,000.
- 2. <u>Effectiveness</u>. In accordance with Section 16.1 of the Plan, the effectiveness of this First Amendment to RigNet, Inc. 2019 Omnibus Incentive Plan (this "*Amendment*") is subject to the approval of the Sponsor's stockholders at the Sponsor's 2020 annual general meeting of stockholders. For the avoidance of doubt, if stockholder approval is not obtained, this Amendment shall be void and of no force and effect.
- 3. **Effect on the Plan**. This Amendment shall not constitute a waiver, amendment or modification of any provision of the Plan not expressly referred to herein. Except as expressly amended or modified herein, the provisions of the Plan are and shall remain in full force and effect and are hereby ratified and confirmed.

### NONQUALIFIED STOCK OPTION AWARD AGREEMENT

This NONQUALIFIED STOCK OPTION AWARD AGREEMENT (this "Agreement") is made by and between RigNet, Inc. a Delaware corporation (the "Company"), and Errol Olivier (the "Holder") effective as of January 8, 2020 (the "Grant Date"). The Company hereby grants to the Holder an option to purchase 10,299 shares of Stock of the Company at a price per share of \$6.20 (the "Option Price") as specified herein (the "Option"):

Vesting Schedule

The Option that is granted hereby shall vest and become exercisable in accordance with the following schedule, provided that the Holder's employment with the Company and its Affiliates has not terminated prior to the applicable vesting date:

1/3 of the Option Shares shall vest and become exercisable on each of the first, second and third anniversary of the Grant Date.

- 1. **Definitions.** For purposes of this Agreement, the following terms shall have the meanings indicated below:
  - a. "Affiliate" means any corporation, partnership, limited liability company or association, trust or other entity or organization which, directly or indirectly, controls, is controlled by, or is under common control with, the Company. For purposes of the preceding sentence, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any entity or organization, shall mean the possession, directly or indirectly, of the power (a) to vote more than fifty percent (50%) of the securities having ordinary voting power for the election of directors or comparable individuals of the controlled entity or organization, or (b) to direct or cause the direction of the management and policies of the controlled entity or organization, whether through the ownership of voting securities or by contract or otherwise.
  - b. "Cause" is defined as any of the following: (i) the Holder's plea of guilty or nolo contendere, or conviction of a felony or a misdemeanor involving moral turpitude; (ii) any act by the Holder of fraud or dishonesty with respect to any aspect of the Company's business including, but not limited to, falsification of Company records; (iii) the Holder's failure to perform his duties (other than by reason of Disability); (iv) the Holder's engagement in misconduct that is materially injurious to the Company (monetarily or otherwise); (v) the Holder's breach of any confidentiality, noncompetition or non-solicitation obligations to the Company, including but not limited to engagement in Detrimental Activity; (vi) the Holder's commencement of employment with an unrelated employer; (vii) material violation by the Holder of any of the Company's written policies, including but not limited to any harassment and/or non-discrimination policies; or (viii) the Holder's gross negligence in the performance of his or her duties.
  - c. "Change in Control" means (i) a change in ownership occurring as the result of a person or group acquiring Stock of the Company, which, when combined with the Stock held by such person or group, constitutes more than seventy-five percent (75%) of the total fair market value or total voting power of the Company; provided the person or group was not considered as owning more than seventy-five percent (75%) of the value or voting power prior to the acquisition; (ii) a change in effective control of the Company occurring as the result of the replacement of a majority of the members of the Board by directors whose appointment or election is not endorsed by a majority of the members of the Board before the date of the appointment or election; or (iii) a change in the ownership of a substantial portion of the assets of the Company occurring as the result of a person or group acquiring assets from the Company that have a total gross fair market value equal to or more than seventy-five percent (75%) of the total gross fair market value of all the assets of the Company immediately prior to such acquisition. The determination of whether a Change of Control has occurred will be made in accordance with Code Section 409A and the regulations thereunder.
  - d. "Code" means the United States Internal Revenue Code of 1986, as amended from time to time.

- e. "Confidential Information" means material of a secret or confidential nature relating to the business, products, or services of the Company or any Affiliate acquired by the Holder during employment with the Company or any Affiliate. "Confidential Information" excludes any information readily available to members of the general public.
- f. "Detrimental Activity" shall include, unless otherwise modified by the Company in connection with a Change in Control: (i) rendering services for any person or organization, or engaging directly or indirectly in any business, which is or becomes competitive with the Company or any Affiliate; (ii) disclosing to anyone outside the Company or any Affiliate, other than the Company's or any Affiliate's business, without prior written authorization from the Company or any Affiliate, any Confidential Information; (iii) soliciting, interfering, inducing, or attempting to cause any employee of the Company or any Affiliate to leave his or her employment, whether done on the Holder's own account or on account of any person, organization, or business which is or becomes competitive with the Company or any Affiliate; or (iv) directly or indirectly soliciting the trade or business of any customer of the Company or any Affiliate.
- g. "*Disability*" or "*Disabled*" means a determination by the Company's long-term disability carrier that a Holder is disabled in accordance with the Company's long-term disability insurance plan (or, in the case of a non-U.S. Holder, in accordance with a comparable disability plan), provided the definition of disability applied under such plan complies with the requirements of Treas. Reg. Section 1.409A-3(i)(4), or, in the case of a Holder who is not covered under such plan, a determination made by the Social Security Administration (or, in the case of a non-U.S. Holder, a determination made by a comparable governmental entity) that the Holder is totally disabled.
- h. "Good Reason" means the occurrence of any of the following without the Holder's prior written consent: (i) a material adverse change in the Holder's position, authority, duties or responsibilities, excluding a change in reporting relationships; (ii) a material reduction in the Holder's base salary; (iii) a material diminution of the Holder's employee benefits (including but not limited to medical, dental, life insurance and long-term disability plans); (iv) the relocation of the Holder's principal place of employment by more than 50 miles from such location as of the Grant Date; or (v) should Steven Pickett no longer serve as the Company's Chief Executive Officer for any reason. Notwithstanding the foregoing, a "Good Reason" shall not exist unless the Holder notifies the Company of the existence of the condition described in this Section 1(c) within ninety (90) days of the initial existence of the condition and the Company does not remedy the condition within thirty (30) days following receipt of such notice.
- i. "Mature Shares" means shares of Stock that the Holder has held for at least six (6) months.
- j. "*Retirement*" shall mean the Holder's voluntary election to retire from employment with the Company or any Affiliate at any time after you have reached both the age of 60 and 5 years of service.
- k. "Stock" means the common stock of the Company, \$0.001 par value per share (or such other par value as may be designated by act of the Company's stockholders).
- 2. **Grant of Option.** The Option is not intended to be an Incentive Stock Option within the meaning of Section 422 of the Internal Revenue Code (the "Code"). The Option shall expire on January 8, 2027 (the "*Expiration Date*").

#### Vesting.

a. *Forfeiture*. If the Holder ceases to be employed by the Company or an Affiliate for any reason before the applicable vesting date, other than in accordance with subsections (b) and (c) below, the unvested portion of the Option shall be forfeited to the Company on the date the Holder ceases to be employed by the Company or an Affiliate. If the Holder breaches, before the applicable vesting date, any non-competition, confidentiality, restrictive covenant or other similar agreement with the Company to which the Holder is subject, the unvested portion of the Option shall be forfeited to the Company on the date the Holder breaches such agreement or covenant. The Holder may exercise the vested portion of the Option, but only within such period of time ending on the earlier of: (i) the date three (3) months following the termination of the Holder's employment with the Company, or (ii) the Expiration Date.

- b. **Death, Retirement or Disability.** If the Holder's employment terminates due to death, Retirement or Disability, the unvested portion of the Option shall automatically become 100% vested on the Holder's date of termination. The Holder or, in the case of the Holder's death, the Holder's executor, administrator, heir or legatee, as the case may be, may exercise the vested portion of the Option, but only within such period of time ending on the earlier of: (i) the date twelve (12) months following the termination of the Holder's employment with the Company, or (ii) the Expiration Date.
- c. *Change in Control*. If a Change in Control occurs and the Holder's employment is terminated by the Company or an Affiliate without Cause or by the Holder for Good Reason, and the Holder's date of termination occurs within twelve (12) months following the Change in Control, the unvested portion of the Option shall automatically become 100% vested on the Holder's date of termination. The Holder may exercise the vested portion of the Option, but only within such period of time ending on the earlier of: (i) the date three (3) months following the termination of the Holder's employment with the Company, or (ii) the Expiration Date.
- 4. **Option Does Not Award Any Rights of a Shareholder.** The Holder shall not have the voting rights or any of the other rights, powers or privileges of a holder of Stock with respect to the Option that is awarded hereby. Only after the Option is exercised will the Holder have all of the rights of a shareholder with respect to each share of Stock issued in exchange for the Option.

### 5. **Manner of Exercise.**

- a. **Election to Exercise.** To exercise the Option, the Holder or in the case of exercise after the Holder's death or incapacity, the Holder's executor, administrator, heir or legatee, as the case may be shall deliver to the Company a fully completed and executed notice of exercise ("**Notice of Exercise**"), in such form as may be designated by the Company in its sole discretion, which shall set forth, inter alia:
  - i. the Holder's election to exercise the Option;
  - ii. the number of shares of Stock being purchased;
  - iii. any restrictions imposed on the shares; and
  - iv. any representations, warranties and agreements regarding the Holder's investment intent and access to information as may be required by the Company to comply with applicable securities laws.

If someone other than the Holder exercises the Option, then such person must submit documentation reasonably acceptable to the Company verifying that such person has the legal right to exercise the Option.

b. Payment of Option Price. The Option Price for the shares of Stock to be acquired on exercise of the Option shall be payable in full at the time of exercise as amended from time to time, plus an amount sufficient to satisfy any tax withholding obligations of the Company that arise in connection with such exercise (as determined by the Company) in any combination of the following (i) cash, certified check, bank draft or postal or express money order for an amount equal to the Option Price or (ii) except as specified below, any other form of payment which is acceptable to the Compensation Committee of the Board of Directors of the Company (the "Committee"). If Mature Share are used for payment by the Holder and allowed by the Committee, the aggregated fair market value of the shares of Stock tendered mut be equal to or less than the aggregate Option Price of the shares of Stock being purchased upon exercise of the Option, and any difference must be paid by cash, certified check, bank draft or postal or express money order payable to the order of the Company. The Committee may permit Holder to elect to pay the Option Price and any applicable tax withholding resulting from such exercise by (A) authorizing a third-party broker to sell all or a portion of the shares of Stock acquired upon exercise of the Option and remit to the Company a sufficient portion of the sale proceeds to pay the Option Price and any applicable federal, state, local and foreign tax withholding resulting from such exercise; or (ii) directing the Company to reduce the number of shares of Stock that will be delivered pursuant to the exercise of the Option.

- c. **Issuance of Shares.** Provided that the exercise notice and payment are in form and substance satisfactory to the Company, the Company shall issue the shares of Stock registered in the name of the Holder, the Holder's authorized assignee, or the Holder's legal representative which shall be evidenced by stock certificates representing the shares with the appropriate legends affixed thereto, appropriate entry on the books of the Company or of a duly authorized transfer agent, or other appropriate means as determined by the Company, and such shares shall be transferable by the Holder (except to the extent that any proposed transfer would, in the opinion of counsel satisfactory to the Company, constitute a violation of applicable securities law).
- 6. **Transferability.** No portion of the Option may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution.
- 7. **Capital Adjustments and Reorganizations.** The existence of the Option shall not affect in any way the right or power of the Company or any company the stock of which is awarded pursuant to this Agreement to make or authorize any adjustment, recapitalization, reorganization or other change in its capital structure or its business, engage in any merger or consolidation, issue any debt or equity securities, dissolve or liquidate, or sell, lease, exchange or otherwise dispose of all or any part of its assets or business, or engage in any other corporate act or proceeding.
- 8. Covenant Not To Compete; Solicit or Disclose Confidential Information.
  - a. The Holder acknowledges that he or she is in possession of and has access to Confidential Information and that he or she will continue to have such possession and access during employment by the Company. The Holder also acknowledges that the Company's business, products and services are highly specialized and that it is essential that they be protected, and, accordingly, the Holder agrees that as partial consideration for the Option granted herein that should the Holder engage in any Detrimental Activity at any time during his or her employment or during a period of one year following his or her termination, the Company shall be entitled to: (i) recover from the Holder any shares that have been issued under the Option; (ii) seek injunctive relief against the Holder pursuant to the provisions of subsection (b) below; (iii) recover all damages, court costs, and attorneys' fees incurred by the Company in enforcing the provisions of this Agreement; and (iv) set off any such sums to which the Company is entitled hereunder against any such sum which may be owed to the Holder by the Company.
  - b. Because of the difficulty of measuring economic losses to the Company as a result of a breach of the foregoing covenants, and because of the immediate and irreparable damage that could be caused to the Company for which it would have no other adequate remedy, the Holder agrees that the foregoing covenants may be enforced by the Company in the event of breach by him or her by injunction relief and restraining order, without the necessity of posting a bond, and that such enforcement shall not be the Company's exclusive remedy for a breach but instead shall be in addition to all other rights and remedies available to the Company.
  - c. The covenants and provisions of this Section 8 are severable and separate, and the unenforceability of any specific covenant or provision shall not affect the enforceability of any other covenant or provision. Moreover, in the event any arbitrator or court of competent jurisdiction shall determine that the scope or time set forth are unreasonable, then it is the intention of the parties that such restrictions be enforced to the fullest extent which the panel or court deems reasonable, and this Agreement shall thereby be reformed.
  - d. Each of the covenants in this Section 8 shall be construed as an agreement independent of any other provision in this Agreement, and the existence of any claim or cause of action of the Holder against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of such covenants or provisions.
- 9. **Tax Withholding.** To the extent that the receipt of the Option, any payment in cash or shares of Stock or the vesting of the Option results in income to the Holder for federal, state or local income, employment or other tax

purposes with respect to which the Company or any Affiliate has a withholding obligation, the Holder shall deliver to the Company at the time of such receipt, payment or vesting, as the case may be, such amount of money as the Company or any Affiliate may require to meet its obligation under applicable tax laws or regulations, and, if the Holder fails to do so, the Company is authorized to withhold from the shares granted hereby or from any cash or stock remuneration then or thereafter payable to the Holder in any capacity any tax required to be withheld by reason of such resulting income.

- 10. **No Fractional Shares.** All provisions of this Agreement concern whole shares. If the application of any provision hereunder would yield a fractional share, such fractional share shall be rounded down to the next whole share if it is less than 0.5 and rounded up to the next whole share if it is 0.5 or more.
- 11. Nontransferability. This Agreement is not transferable by the Holder otherwise than by will or by the laws of descent and distribution.
- 12. **Employment Relationship.** For purposes of this Agreement, the Holder shall be considered to be in the employment of the Company and its Affiliates as long as the Holder has an employment relationship with the Company and its Affiliates. The Committee shall determine any questions as to whether and when there has been a termination of such employment relationship, and the cause of such termination, and the Committee's determination shall be final and binding on all persons.
- 13. **Not an Employment Agreement.** This Agreement is not an employment agreement, and no provision of this Agreement shall be construed or interpreted to create an employment relationship between the Holder and the Company or any Affiliate, to guarantee the right to remain employed by the Company or any Affiliate for any specified term or require the Company or any Affiliate to employ the Holder for any period of time.
- 14. **Legend.** The Holder consents to the placing on the certificate for the shares issued hereunder an appropriate legend restricting resale or other transfer of the shares except in accordance with all applicable securities laws and rules thereunder.
- 15. **Notices.** Any notice, instruction, authorization, request or demand required hereunder shall be in writing, and shall be deemed to have been duly given when delivered or mailed to the Company or the Holder, as applicable, by (a) personal delivery; (b) United States registered mail, return receipt requested, postage prepaid, addressed to the Company at the then current address of the Company's principal corporate office, or to the Holder at the Holder's residential address indicated in the Company's records; or (c) email to the Company at LegalDesk@Rig.Net or to the Holder at the Holder's email address indicated in the Company's records.
- 16. **Amendment and Waiver.** This Agreement may be amended, modified or superseded only by written instrument executed by the Company and the Holder. Only a written instrument executed and delivered by the party waiving compliance hereof shall make any waiver of the terms or conditions. Any waiver granted by the Company shall be effective only if executed and delivered by a duly authorized executive officer of the Company other than the Holder. The failure of any party at any time or times to require performance of any provisions hereof shall in no manner effect the right to enforce the same. No waiver by any party of any term or condition, or the breach of any term or condition contained in this Agreement, in one or more instances, shall be construed as a continuing waiver of any such condition or breach, a waiver of any other condition, or the breach of any other term or condition.
- 17. **Arbitration.** In the event of any difference of opinion concerning the meaning or effect of this Agreement, such difference shall be resolved by the Committee. Any controversy arising out of or relating to this Agreement shall be resolved by arbitration conducted in Houston, Texas pursuant to the arbitration rules of the American Arbitration Association. The arbitration shall be final and binding on the parties.
- 18. **Governing Law and Severability.** The validity, construction and performance of this Agreement shall be governed by the laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of this Agreement to the substantive law of another jurisdiction. The invalidity of any provision of this Agreement shall not affect any other provision of this Agreement, which shall remain in full force and effect.

- 19. **Successors and Assigns.** Subject to the limitations which this Agreement imposes upon the transferability of the Option granted hereby, this Agreement shall bind, be enforceable by and inure to the benefit of the Company and its successors and assigns, and to the Holder, the Holder's permitted assigns, executors, administrators, agents, legal and personal representatives.
- 20. **Counterparts.** This Agreement may be executed in counterparts, each of which shall be deemed an original for all purposes but all of which taken together shall constitute one and the same instrument.
- 21. **Recoupment.** If the Holder is subject to the Company's clawback policy (the "*Policy*"), the Holder agrees that the Option is subject to the terms of the Policy, as may be amended from time to time.
- 22. **Compliance with Section 409A.** It is the Company's intent that this Agreement be exempt from the application of, or otherwise comply with, the requirements of Section 409A. Specifically, any taxable benefits or payments provided under this Agreement are intended to qualify for the statutory stock option exception to Section 409A to the maximum extent possible and, to the extent they do not so qualify, are intended to qualify for the nonstatutory stock option exception to Section 409A to the maximum extent possible. Although the Company will use its best efforts to avoid the imposition of taxation, interest and penalties under Section 409A, the tax treatment of the benefits is not warranted or guaranteed. Neither the Company, its Affiliates nor their respective directors, officers, employees or advisers shall be held liable for any taxes, interest, penalties or other monetary amounts owed by the Holder (or any other individual claiming a benefit through the Holder).

[SIGNATURE PAGE FOLLOWS]

**IN WITNESS WHEREOF**, the Holder and the Company agree and acknowledge that this Option is granted under and governed by the terms and conditions of this Agreement.

Holder RigNet, Inc.

By: /s/ STEVEN E. PICKETT By: /s/ ERROL OLIVIER

Name: Errol Olivier Name: Steven E. Pickett

Title: Chief Executive Officer and President

### RESTRICTED STOCK UNIT AWARD AGREEMENT

This RESTRICTED STOCK UNIT AWARD AGREEMENT (this "*Agreement*") is made by and between RigNet, Inc. a Delaware corporation (the "*Company*"), and Errol Olivier (the "*Holder*") effective as of January 8, 2020 (the "*Grant Date*"). The Company hereby grants to the Holder the 18,787 Restricted Stock Units specified herein (the "*RSUs*"):

Vesting Schedule

The RSUs that are granted hereby shall be subject to the Forfeiture Restrictions during the Period of Restriction. The Forfeiture Restrictions shall lapse as to the RSUs that are awarded hereby in accordance with the following schedule, provided that the Holder's employment with the Company and its Affiliates has not terminated prior to the applicable lapse date:

1/3 of the shares shall vest and the forfeiture restrictions shall lapse on each of the first, second and third anniversary of the Grant Date.

- 1. **Definitions.** For purposes of this Agreement, the following terms shall have the meanings indicated below:
  - a. "Affiliate" means any corporation, partnership, limited liability company or association, trust or other entity or organization which, directly or indirectly, controls, is controlled by, or is under common control with, the Company. For purposes of the preceding sentence, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any entity or organization, shall mean the possession, directly or indirectly, of the power (a) to vote more than fifty percent (50%) of the securities having ordinary voting power for the election of directors or comparable individuals of the controlled entity or organization, or (b) to direct or cause the direction of the management and policies of the controlled entity or organization, whether through the ownership of voting securities or by contract or otherwise.
  - b. "Cause" is defined as any of the following: (i) the Holder's plea of guilty or nolo contendere, or conviction of a felony or a misdemeanor involving moral turpitude; (ii) any act by the Holder of fraud or dishonesty with respect to any aspect of the Company's business including, but not limited to, falsification of Company records; (iii) the Holder's failure to perform his or her duties (other than by reason of Disability); (iv) the Holder's engagement in misconduct that is materially injurious to the Company (monetarily or otherwise); (v) the Holder's breach of any confidentiality, noncompetition or non-solicitation obligations to the Company, including but not limited to engagement in Detrimental Activity; (vi) the Holder's commencement of employment with an unrelated employer; (vii) material violation by the Holder of any of the Company's written policies, including but not limited to any harassment and/or non-discrimination policies; or (viii) the Holder's gross negligence in the performance of his or her duties.
  - c. "Change in Control" means (i) a change in ownership occurring as the result of a person or group acquiring Stock of the Company, which, when combined with the Stock held by such person or group, constitutes more than seventy-five percent (75%) of the total fair market value or total voting power of the Company; provided the person or group was not considered as owning more than seventy-five percent (75%) of the value or voting power prior to the acquisition; (ii) a change in effective control of the Company occurring as the result of the replacement of a majority of the members of the Board by directors whose appointment or election is not endorsed by a majority of the members of the Board before the date of the appointment or election; or (iii) a change in the ownership of a substantial portion of the assets of the Company occurring as the result of a person or group acquiring assets from the Company that have a total gross fair market value equal to or more than seventy-five percent (75%) of the total gross fair market value of all the assets of the Company immediately prior to such acquisition. The determination of whether a Change of Control has occurred will be made in accordance with Code Section 409A and the regulations thereunder.
  - d. "Code" means the United States Internal Revenue Code of 1986, as amended from time to time.

- e. "Confidential Information" means material of a secret or confidential nature relating to the business, products, or services of the Company or any Affiliate acquired by the Holder during employment with the Company or any Affiliate. "Confidential Information" excludes any information readily available to members of the general public.
- f. "Detrimental Activity" shall include, unless otherwise modified by the Company in connection with a Change in Control: (i) rendering services for any person or organization, or engaging directly or indirectly in any business, which is or becomes competitive with the Company or any Affiliate; (ii) disclosing to anyone outside the Company or any Affiliate, other than the Company's or any Affiliate's business, without prior written authorization from the Company or any Affiliate, any Confidential Information; (iii) soliciting, interfering, inducing, or attempting to cause any employee of the Company or any Affiliate to leave his or her employment, whether done on the Holder's own account or on account of any person, organization, or business which is or becomes competitive with the Company or any Affiliate; or (iv) directly or indirectly soliciting the trade or business of any customer of the Company or any Affiliate.
- g. "*Disability*" or "*Disabled*" means a determination by the Company's long-term disability carrier that a Holder is disabled in accordance with the Company's long-term disability insurance plan (or, in the case of a non-U.S. Holder, in accordance with a comparable disability plan), provided the definition of disability applied under such plan complies with the requirements of Treas. Reg. Section 1.409A-3(i)(4), or, in the case of a Holder who is not covered under such plan, a determination made by the Social Security Administration (or, in the case of a non-U.S. Holder, a determination made by a comparable governmental entity) that the Holder is totally disabled.
- h. "*Forfeiture Restrictions*" shall mean the prohibitions and restrictions set forth herein with respect to the sale or other disposition of the RSUs issued to the Holder hereunder and the obligation to forfeit and surrender such RSUs to the Company.
- i. "Good Reason" means the occurrence of any of the following without the Holder's prior written consent: (i) a material adverse change in the Holder's position, authority, duties or responsibilities, excluding a change in reporting relationships; (ii) a material reduction in the Holder's base salary; (iii) a material diminution of the Holder's employee benefits (including but not limited to medical, dental, life insurance and long-term disability plans); (iv) the relocation of the Holder's principal place of employment by more than 50 miles from such location as of the Grant Date; ; or (v) should Steven Pickett no longer serve as the Company's Chief Executive Officer for any reason. Notwithstanding the foregoing, a "Good Reason" shall not exist unless the Holder notifies the Company of the existence of the condition described in this Section 1(i) within ninety (90) days of the initial existence of the condition and the Company does not remedy the condition within thirty (30) days following receipt of such notice.
- j. "Period of Restriction" shall mean the period during which RSUs are subject to Forfeiture Restrictions.
- k. "*Retirement*" shall mean the Holder's voluntary election to retire from employment with the Company or any Affiliate at any time after you have reached both the age of 60 and 5 years of service.
- l. "*Stock*" means the common stock of the Company, \$0.001 par value per share (or such other par value as may be designated by act of the Company's stockholders).

Capitalized terms not otherwise defined in this Agreement shall have the meanings given to such terms in the Plan.

2. **Transfer Restrictions.** The RSUs granted hereby may not be sold, assigned, pledged, exchanged, hypothecated or otherwise transferred, encumbered or disposed of. Any such attempted sale, assignment, pledge, exchange, hypothecation, transfer, encumbrance or disposition in violation of this Agreement shall be void and the Company shall not be bound thereby. Further, any shares of the Stock granted hereby upon vesting of the RSUs (the "*Shares*") may not be sold or otherwise disposed of in any manner that would constitute a violation of any applicable securities laws. The Holder also agrees that the Company may (a) refuse to cause the transfer of the Shares to be registered on the applicable stock transfer records of the Company if such proposed transfer would, in the opinion of counsel satisfactory to the Company, constitute a violation of any applicable securities law; and (b) give related instructions to the transfer agent, if any, to stop registration of the transfer of the Shares.

## Vesting.

- a. *Forfeiture*. If the Holder ceases to be employed by the Company or an Affiliate for any reason before the applicable lapse date, other than in accordance with subsections (b) and (c) below, the Forfeiture Restrictions then applicable to the RSUs shall not lapse and all the RSUs then subject to the Forfeiture Restrictions shall be forfeited to the Company on the date the Holder ceases to be employed by the Company or an Affiliate. If the Holder breaches, before the applicable lapse date, any non-competition, confidentiality, restrictive covenant or other similar agreement with the Company to which the Holder is subject, the Forfeiture Restrictions then applicable to the RSUs shall not lapse and all the RSUs then subject to the Forfeiture Restrictions shall be forfeited to the Company on the date the Holder breaches such agreement or covenant.
- b. **Death, Retirement or Disability.** If the Holder's employment terminates due to death, Retirement or Disability, all unvested RSUs shall automatically become 100% vested on the Holder's date of termination.
- c. *Change in Control.* If a Change in Control occurs and the Holder's employment is terminated by the Company or an Affiliate without Cause or by the Holder for Good Reason, and the Holder's date of termination occurs within twelve (12) months following the Change in Control, all unvested RSUs shall automatically become 100% vested on the Holder's date of termination.
- 4. **RSUs Do Not Award Any Rights of a Shareholder.** The Holder shall not have the voting rights or any of the other rights, powers or privileges of a holder of Stock with respect to the RSUs that are awarded hereby. Only after a share is issued in exchange for an RSU will the Holder have all of the rights of a shareholder with respect to such share of Stock issued in exchange for an RSU.
- 5. **Delivery of Shares.** Upon the lapse of the Forfeiture Restrictions with respect to the RSUs granted hereby, the Company shall cause to be delivered to the Holder the Shares evidenced by stock certificates representing the Shares with the appropriate legends affixed thereto, appropriate entry on the books of the Company or of a duly authorized transfer agent, or other appropriate means as determined by the Company, and such Shares shall be transferable by the Holder.
- 6. **Capital Adjustments and Reorganizations.** The existence of the RSUs shall not affect in any way the right or power of the Company or any company the stock of which is awarded pursuant to this Agreement to make or authorize any adjustment, recapitalization, reorganization or other change in its capital structure or its business, engage in any merger or consolidation, issue any debt or equity securities, dissolve or liquidate, or sell, lease, exchange or otherwise dispose of all or any part of its assets or business, or engage in any other corporate act or proceeding.

# 7. Covenant Not To Compete; Solicit or Disclose Confidential Information.

- a. The Holder acknowledges that he or she is in possession of and has access to Confidential Information and that he or she will continue to have such possession and access during employment by the Company. The Holder also acknowledges that the Company's business, products and services are highly specialized and that it is essential that they be protected, and, accordingly, the Holder agrees that as partial consideration for the RSUs granted herein that should the Holder engage in any Detrimental Activity at any time during his or her employment or during a period of one year following his or her termination, the Company shall be entitled to: (i) recover from the Holder the value of any portion of the RSUs that has been paid; (ii) seek injunctive relief against the Holder pursuant to the provisions of subsection (b) below; (iii) recover all damages, court costs, and attorneys' fees incurred by the Company in enforcing the provisions of this Agreement; and (iv) set off any such sums to which the Company is entitled hereunder against any such sum which may be owed to the Holder by the Company.
- b. Because of the difficulty of measuring economic losses to the Company as a result of a breach of the foregoing covenants, and because of the immediate and irreparable damage that could be caused to the Company for which it would have no other adequate remedy, the Holder agrees that the foregoing covenants

may be enforced by the Company in the event of breach by him or her by injunction relief and restraining order, without the necessity of posting a bond, and that such enforcement shall not be the Company's exclusive remedy for a breach but instead shall be in addition to all other rights and remedies available to the Company.

- c. The covenants and provisions of this Section 7 are severable and separate, and the unenforceability of any specific covenant or provision shall not affect the enforceability of any other covenant or provision. Moreover, in the event any arbitrator or court of competent jurisdiction shall determine that the scope or time set forth are unreasonable, then it is the intention of the parties that such restrictions be enforced to the fullest extent which the panel or court deems reasonable, and this Agreement shall thereby be reformed.
- d. Each of the covenants in this Section 7 shall be construed as an agreement independent of any other provision in this Agreement, and the existence of any claim or cause of action of the Holder against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of such covenants or provisions.
- 8. **Tax Withholding.** To the extent that the receipt of the RSUs or the lapse of any Forfeiture Restrictions results in income to the Holder for federal, state or local income, employment or other tax purposes with respect to which the Company or any Affiliate has a withholding obligation, the Holder shall deliver to the Company at the time of such receipt or lapse, as the case may be, such amount of money as the Company or any Affiliate may require to meet its obligation under applicable tax laws or regulations, and, if the Holder fails to do so, the Company is authorized to withhold from the Shares granted hereby or from any cash or stock remuneration then or thereafter payable to the Holder in any capacity any tax required to be withheld by reason of such resulting income.
- 9. **No Fractional Shares.** All provisions of this Agreement concern whole shares. If the application of any provision hereunder would yield a fractional share, such fractional share shall be rounded down to the next whole share if it is less than 0.5 and rounded up to the next whole share if it is 0.5 or more.
- 10. Nontransferability. This Agreement is not transferable by the Holder otherwise than by will or by the laws of descent and distribution.
- 11. **Employment Relationship.** For purposes of this Agreement, the Holder shall be considered to be in the employment of the Company and its Affiliates as long as the Holder has an employment relationship with the Company and its Affiliates. The Compensation Committee of the Company's Board of Directors (the "*Committee*") shall determine any questions as to whether and when there has been a termination of such employment relationship, and the cause of such termination, and the Committee's determination shall be final and binding on all persons.
- 12. **Not an Employment Agreement.** This Agreement is not an employment agreement, and no provision of this Agreement shall be construed or interpreted to create an employment relationship between the Holder and the Company or any Affiliate, to guarantee the right to remain employed by the Company or any Affiliate for any specified term or require the Company or any Affiliate to employ the Holder for any period of time.
- 13. **Legend.** The Holder consents to the placing on the certificate for the Shares an appropriate legend restricting resale or other transfer of the Shares except in accordance with all applicable securities laws and rules thereunder.
- 14. **Notices.** Any notice, instruction, authorization, request or demand required hereunder shall be in writing, and shall be deemed to have been duly given when delivered or mailed to the Company or the Holder, as applicable, by (a) personal delivery; (b) United States registered mail, return receipt requested, postage prepaid, addressed to the Company at the then current address of the Company's principal corporate office, or to the Holder at the Holder's residential address indicated in the Company's records; or (c) email to the Company at legaldesk@rig.net or to the Holder at the Holder's email address indicated in the Company's records.
- 15. **Amendment and Waiver.** Except as otherwise provided herein, this Agreement may be amended, modified or superseded only by written instrument executed by the Company and the Holder. Only a written instrument executed and delivered by the party waiving compliance hereof shall make any waiver of the terms or conditions. Any waiver granted by the Company shall be effective only if executed and delivered by a duly authorized

executive officer of the Company other than the Holder. The failure of any party at any time or times to require performance of any provisions hereof shall in no manner effect the right to enforce the same. No waiver by any party of any term or condition, or the breach of any term or condition contained in this Agreement, in one or more instances, shall be construed as a continuing waiver of any such condition or breach, a waiver of any other condition, or the breach of any other term or condition.

- 16. **Arbitration.** In the event of any difference of opinion concerning the meaning or effect of the Plan or this Agreement, such difference shall be resolved by the Committee. Any controversy arising out of or relating to this Agreement shall be resolved by arbitration conducted in Houston, Texas pursuant to the arbitration rules of the American Arbitration Association. The arbitration shall be final and binding on the parties.
- 17. **Governing Law and Severability.** The validity, construction and performance of this Agreement shall be governed by the laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of this Agreement to the substantive law of another jurisdiction. The invalidity of any provision of this Agreement shall not affect any other provision of this Agreement, which shall remain in full force and effect.
- 18. **Successors and Assigns.** Subject to the limitations which this Agreement imposes upon the transferability of the RSUs granted hereby, this Agreement shall bind, be enforceable by and inure to the benefit of the Company and its successors and assigns, and to the Holder, the Holder's permitted assigns, executors, administrators, agents, legal and personal representatives.
- 19. **Counterparts.** This Agreement may be executed in counterparts, each of which shall be deemed an original for all purposes but all of which taken together shall constitute one and the same instrument.
- 20. **Recoupment.** If the Holder is subject to the Company's clawback policy (the "*Policy*"), the Holder agrees that the RSUs are subject to the terms of the Policy, as may be amended from time to time.
- 21. **Compliance with Section 409A.** It is the Company's intent that this Agreement be exempt from the application of, or otherwise comply with, the requirements of Section 409A. Specifically, any taxable benefits or payments provided under this Agreement are intended to qualify for the short term deferral exception to Section 409A to the maximum extent possible. Although the Company will use its best efforts to avoid the imposition of taxation, interest and penalties under Section 409A, the tax treatment of the benefits provided is not warranted or guaranteed. Neither the Company, its Affiliates nor their respective directors, officers, employees or advisers shall be held liable for any taxes, interest, penalties or other monetary amounts owed by the Holder (or any other individual claiming a benefit through the Holder).

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the Holder and the Company agree and acknowledge that this grant of Restricted Stock Units is granted under and governed by the terms and conditions of this Agreement.

Holder RigNet, Inc.

Name: Errol Olivier

By: /s/ ERROL OLIVIER By: /s/ STEVEN E. PICKETT

Title: Chief Executive Officer and President

Name: Steven E. Pickett

# CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of Viasat, Inc. of our report dated May 28, 2020 relating to the financial statements, financial statement schedule and the effectiveness of internal control over financial reporting, which appears in Viasat, Inc.'s Annual Report on Form 10-K for the year ended March 31, 2020.

/s/ PricewaterhouseCoopers LLP

San Diego, California

April 30, 2021